# Vacuum Solutions for Metric–affine Gravity with Spectral Analysis of the Massless Dirac Operator

submitted by

# Elvis Baraković

for the degree of Doctor of Philosophy of the

University of Tuzla Faculty of Natural Sciences and Mathematics Department of Mathematics



July 2016

# Contents

Su	Summary			
1	Introduction			
	1.1	General relativity and the alternatives	2	
	1.2	Gravitational and massless neutrino fields	6	
	1.3	Quadratic metric-affine gravity	7	
	1.4	Notation and background	11	
		1.4.1 Irreducible pieces of torsion and curvature	13	
		1.4.2 On the Levi-Civita tensor	18	
		1.4.3 Spinor formalism	20	
	1.5	Introducing the massless Dirac operator	21	
	1.6	Main results of the thesis	23	
	1.7	Structure of the thesis	25	
<b>2</b>	Nev	v Solutions for Quadratic Metric-Affine Gravity	27	
-	2.1	Classical pp-waves	$\frac{-}{27}$	
		2.1.1 Pauli matrices for pp-waves	$\frac{-}{30}$	
	2.2 Generalising pp-waves			
		2.2.1 Generalised pp-waves with purely tensor torsion	31	
		2.2.2 Generalised pp-waves with purely axial torsion	34	
	2.3	New solutions for quadratic metric-affine gravity	38	
		2.3.1 Explicit representation of the field equations	40	
		2.3.2 Variation with respect to metric	42	
		2.3.3 Variation with respect to connection	44	
		2.3.4 PP-type solutions of the field equations	46	
	2.4	The massless Dirac operator in theories of gravity	48	
		2.4.1 Metric-affine vs Einstein-Weyl pp-wave type solutions .	51	
3	$\mathbf{Spe}$	ctral Analysis of the Massless Dirac Operator on a 3-		

## i

 $\mathbf{54}$ 

dimensional Manifold

	3.1	Some properties of an elliptic self-adjoint	
		first-order differential operator	55
	3.2	The massless Dirac operator	57
	3.3	The spectrum of the massless Dirac operator	62
	3.4	The perturbation theory for the massless	
		Dirac operator	65
		3.4.1 Pseudoinverse operator construction	69
		3.4.2 Explicit formulae for the asymptotic coefficients	71
	3.5	Spectral asymmetry of the massless Dirac operator on a 3-torus	72
		3.5.1 Numerical analysis of the spectrum	75
		3.5.2 Analytical formulae for perturbed eigenvalues	78
		3.5.3 Explicit examples of spectral asymmetry	83
$\mathbf{A}$	Eins	stein and Yang-Mills Equations	88
	A.1	The Einstein field equations	88
	A.2	The Yang-Mills equations	89
в	Biaı	nchi Identity for Curvature	92
	B.1	Explicit formula for $R^{(5)}$ piece of curvature	92
	B.2	The derivation of the Bianchi identity	93
$\mathbf{C}$	Exp	licit Variations of Some Quadratic Forms on Curvature	98
U	C.1	Variation of $\int Ric_{\mu\nu}Ric^{\mu\nu}$	98
	$C_{2}$	Variation of $\int Bic^{(2)} Bic$	99
	$C_{2}$	Variation of $\int Dic^{(2)} Dic^{(2)} \mu\nu$	00
	0.5	Variation of $\int Ric_{\mu\nu}Ric^{\nu\mu}$	99
D	Det	ailed calculations of the asymptotic coefficients 10	01
	D.1	The calculations of the $\lambda_{\pm}^{(1)}$ coefficients $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	01
	D.2	The calculations of the $\lambda_{\pm}^{(2)}$ coefficients $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	02
Bi	bliog	craphy 10	07
	C		

# Summary

In this thesis we deal with quadratic metric-affine gravity and the massless Dirac operator. We review known solutions for quadratic metric-affine gravity and we present new non-Riemannian solutions for this theory. The new solution is presented in the form of generalised pp-waves with purely axial torsion. We also propose a physical interpretation of these new solutions by comparing them to solutions of the Einstein-Weyl theory.

Another aim of this thesis is the spectral analysis of the massless Dirac operator on a closed 3-dimensional manifold. The massless Dirac operator describes a massless neutrino. We review two examples where the spectrum of this operator can be evaluated explicitly and it turns out that in these two particular examples the spectrum is symmetric about zero, although there is no mathematical or physical reason for it to be symmetric in the general case. Applying perturbation theory methods to the massless Dirac operator, we successfully observe spectral asymmetry on the 3-torus and derive explicit asymptotic formulae for perturbations of the eigenvalues  $\pm 1$ .

# Chapter 1 Introduction

The first developments in the theory of gravity go back to ancient Greece. In those times, the motion of the body in a free fall was considered purely philosophically and much time had passed before any mathematical model was set. Aristotle presented a philosophical argument that heavier bodies must fall faster than lighter ones and for centuries no one doubted this argument.

Doubts that this theory of Aristotle was true arose during the Renaissance and were presented by the great Italian scientist Galileo Galilei. Galileo first began an analysis and experimental verification of the laws of movement and presented these results in his two works *Dialogo* [37] in 1632 and *Discorsi* [38] in 1638. From those observations Galileo concluded that two bodies under the influence of the gravitational field of the Earth move in the same way and independently of their respective masses. This was in stark contrast to Aristotle's arguments.

The greatest contribution after Galileo in the research of the theory of gravity was given by the great English scientist Sir Isaac Newton. Newton based his observations on Kepler's laws of planetary motion and in 1687 he presented his observations in his famous work *Philosophiae Naturalis Principia Mathematica* [66]. The basic premise of his work was that the geometry of the three-dimensional space is Euclidean in nature and time is considered as a separate parameter which is required for the description of movement. Newton's theory of gravity claimed that the mass of the body is the source of gravity and that any two bodies attract each other with the force that is proportional to the product of their masses and inversely proportional to the square of their distance. This theory of gravity was in accordance with the Galilean principle of equivalence, but the principle itself had no effect on the construction of Newton's theory.

However, as one of the fundamental laws of physics, Newton's theory of gravity had defects in the form of time invariance and depended only on spatial distance. According to Newton's theory of gravity, the strength of the gravitational force between two objects is proportional to the inertial masses of the objects. The inertial mass has a dual role: it is a measure of the resistance of the object to the change of its speed, but on the other hand it also plays the role of the gravitational charge. In the same way as an electric charge determines the strength of the electric force between two charged objects, in Newton's theory the inertial mass determines the strength of the gravitational force. The second problem was that the gravitational force was described as the force of attraction between two objects. Consequently, if one was to move one of these two objects, the other object would 'realise' that fact due to the changes in the gravitational force regardless of their distance. The third problem was the discrepancy between the predictions of Newton's theory.

# 1.1 General relativity and the alternatives

Newton's theory of gravity was further improved upon by Albert Einstein with his *theory of general relativity*, see [27]. The entire concept of Newton's understanding of space was changed by Einstein and space and time were no longer considered separately. The theory of general relativity is a theory of gravity that Einstein developed between 1907 and 1915 using Riemannian geometry. Einstein's colleague from his student days, the mathematician Marcel Grossmann, together with Hermann Minkowski, helped Einstein in the formulation of this theory. The new concept of *spacetime* of general relativity is fully described by the metric and this metric does not define only the distance but also the parallel transport, i.e. the Levi-Civita connection. As a central part of his theory of gravity, Einstein proposed the equation that relates geometry and matter

$$\underbrace{Ric_{\mu\nu} - \frac{1}{2}\mathcal{R}g_{\mu\nu}}_{\text{geometry}} = \underbrace{\frac{8\pi G}{c^4}T_{\mu\nu}}_{\text{matter}},$$

where  $T_{\mu\nu}$  is the stress energy tensor which depends on matter, G is the gravitational constant, c is the speed of light, Ric is the Ricci curvature (1.18) and  $\mathcal{R}$  is the scalar curvature (1.19). This equation presents a direct connection between the geometry of spacetime and matter. The vacuum Einstein equation has the form

$$Ric_{\mu\nu} - \frac{1}{2}\mathcal{R}g_{\mu\nu} = 0.$$

General relativity could be interpreted as follows: spacetime tells matter how to move and matter tells the spacetime how to curve, see [104]. Matter 'curves' spacetime, so in addition to the changes of spatial coordinates, something also happens to the time coordinate in the spacetime continuum. General relativity predicts that the time near some massive objects is more 'bent', i.e. time is slower near objects with greater mass than near objects with lesser mass. General relativity has been confirmed by much experimental data, such as the precession perihelion of Mercury, the deflection of light as it passes close to the Sun, the gravitational red shift, the time delay of the radar signal, the operation of the GPS device, etc. General relativity predicted the existence of gravitational waves, which was recently experimentally confirmed by the Laser Interferometer Gravitational-Wave Observatory (LIGO) in 2016. General relativity predicted the existence of black holes as objects in the spacetime continuum whose gravity is so strong that any form of matter can not escape them, even light which is moving as fast as it possible in nature. Modern cosmological observations support the natural existence of such objects in space. A comprehensive review of spacetimes in general relativity was done by Griffiths and Podolský [48].

Einstein himself expected much more from his theory of gravity. He was not fully satisfied because it failed to merge the gravitational field and the electromagnetic field into a single model, see [28]. In order to relate gravity and electromagnetism, Einstein thought about an alternative theory of gravity known as *teleparallelism*, see [97]. The teleparallel spacetime is seen as the connected real 4-dimensional manifold equipped with a Lorentzian metric whose curvature vanishes, but whose torsion does not vanish. A metric tensor is defined by the components of a dynamic tetrad field. For more results in the field of this alternative theory of gravity see e.g. [23, 34, 64, 69, 86, 95, 96, 102].

Gravity is the only physical interaction for which there is no consistent quantum formulation. Many attempts of the quantisation of gravity have thus far been unsuccessful. There are many alternative theories such as *metric-affine gravity, gauge symmetry, supergravity, Kaluza-Klein theory* or *string theory* that may lead us to a unified quantum theory of the fundamental interactions. Previous successes of these ideas motivate us to study them in more detail and hopefully it will bring us closer to the formulation of quantum gravity. General relativity has solved many disadvantages of earlier theories, but we still have many open questions which led to the development of alternative theories of gravity. In this thesis, we are dedicated to the study of an alternative theory of gravity known as *metric-affine gravity*.

Metric-affine gravity (MAG) is an alternative theory of gravity which is a extension of general relativity. In metric affine-gravity we leave the Riemannian spacetime of Einstein's general relativity and add *torsion* (1.12) which leads to the Riemann-Cartan spacetime and possible nonmetricity (1.16). In metric-affine gravity, spacetime is viewed as a connected real 4-dimensional manifold equipped with a Lorentzian metric and an affine connection. Unlike general relativity, in metric-affine gravity the metric and the connection are dynamic variables. The spacetime of metric-affine gravity reduces to the spacetime of the general relativity under the assumption that the torsion is zero and that the connection metric compatible. Metric-affine gravity is directly derived from the gauge theory of gravity where the linear connection plays the role of a gauge fields. The unknowns of metric-affine gravity are the 10 components of the metric tensor and the 64 connection coefficients. An advanced and comprehensive overview of this theory can be found in [13, 50, 51, 72, 73, 74, 75, 76, 77, 99, 100, 101].

In this thesis we are primarily concerned with the study of quadratic metric-affine gravity (QMAG), see Section 1.3. The mathematical model of this theory is that the Riemannian manifold is characterised by the principle of the action, which is a functional defined as an integral over the 4-dimensional manifold whose integrand is a purely quadratic form on curvature. The metric is still of Lorentzian signature and the unknown quantities are the components of the metric and the connection. Using variational calculus, we determine the stationary functions, i.e. the functions in which the change of the action is zero, which determines the metric and the connection. Mathematically, we consider the action defined by (1.1) where the Lagrangian is a quadratic form on curvature. Varying the action (1.1) with respect to the metric and the connection independently, we produce the system of Euler-Lagrange equation (1.2), (1.3). Our goal is to analyse the Riemannian and the non-Riemannian solutions of the system (1.2), (1.3), see Definition 1.3.3.

A large body of work is devoted to the analysis of Riemannian and non-Riemannian solutions of the Euler-Lagrange system (1.2), (1.3) in the case of quadratic Lagrangians, see e.g. [11, 30, 39, 50, 68, 70, 77, 82, 83, 87, 88, 89, 94, 99, 100, 101]. An important result in this field was given by Vassiliev [101], who solved the problem of existence and uniqueness of Riemannian solutions of the system (1.2), (1.3) in the most general case of the quadratic forms with 16  $R^2$  terms. Vassiliev showed that there are only three types of Riemannian solutions of the system (1.2), (1.3), see Section 4 and Section 5 of [101] and also presented the construction of "torsion waves" as one non-Riemannian solution of the system (1.2), (1.3). The same solution previously was obtained independently by Singh and Griffiths [89] and King and Vassiliev [53] for the Yang-Mills case (1.7). Torsion waves presented in [101] can be considered as non-Riemannian analogues of a pp-space, see Definition 2.1.1. Pasic and Vassiliev [77] presented a generalisation of classical pp-waves to spacetimes with torsion and constructed new class of non-Riemannian solutions in the most general case of quadratic form. A physical interpretation for that class of non-Riemannian solutions was given by Pasic and Barakovic [74], where it was suggested that a generalised pp-wave of parallel Ricci curvature represents a metric-affine model for the massless neutrino.

Motivated by the result of Singh [87] who presented a solution of the system (1.2), (1.3) with purely axial torsion (1.26) for the Yang-Mills action (1.7), we previously constructed a new non-Riemannian solution in [75], by generalising classical pp-waves to metric compatible spacetimes with purely axial torsion and proving that they are new vacuum solutions of the system (1.2), (1.3) for the Yang-Mills action (1.7). In this thesis, we aim to prove that these spacetimes are also a new solution of the system (1.2), (1.3) for the more general quadratic form with 11  $R^2$  terms (1.4). In the future it is our aim to see whether these spacetimes are also solutions of the system (1.2), (1.3)for the most general quadratic form with 16  $R^2$  terms (1.6). Also, in view of the construction of a new non-Riemannian solution for quadratic metricaffine gravity, the paper of Singh [88] is of particular interest to us, where a new solution of the field equations (1.2), (1.3) for the Yang-Mills action (1.7)with purely trace torsion (1.25) is presented. It would be interesting to see whether it is possible to generalise classical pp-waves to metric compatible spacetimes with purely trace torsion, as was similarly done in [75] and to see whether they are solutions of the system (1.2), (1.3). It is important to stress the fact that Sing [87], [88] did not use the most general quadratic form with 16  $R^2$  terms and the presented solutions can not be obtained using the double duality ansatz, see [62].

The analysis of pp-waves has a long and fruitful history. Pp-waves in the Yang-Mills type of quadratic metric-affine gravity with nontrivial torsion and with nonzero nonmetricity which do not belong to the triplet ansatz class [50] were analised by Obukhov [68], the motivation for which came from his earlier work [67]. The quadratic form considered by Obukhov is the most general with 16  $R^2$  terms. According to Obukhov, although Einstein's general relativity is supported by experiments on a macroscopic scale, the gravitational interaction on the microscopic level are not well understood. Gravity gauge models represent an alternative description of gravity in the micro cosmos. The study of the exact solutions of metric-affine gravity is essential for the understanding and development of physical aspects such as quantisation, physics of hadrons, studies of the early universe, etc. A construction and comparison of wave solutions in different models can explain the physical contents and relations between the microscopic and macroscopic gravitational theory.

An interesting analysis of plane-fronted gravitational and electromagnetic

waves in metric-affine gravity with a cosmological constant in the triplet ansatz class was done by Garcia et al. [39]. As the authors emphasise, for restricted irreducible pieces of torsion and nonmetricity, there are similarities between the Einstein-Maxwell system and the vacuum metric-affine gravity field equations. In the same paper, the authors give a review of pp-waves and electromagnetic waves in the Einstein-Maxwell theory and present ppwaves and electromagnetic waves as solutions of metric-affine gravity with the cosmological constant in the triplet ansatz class. These waves carry curvature, nonmetricity, torsion and an electromagnetic field.

Gravitational waves as exact solutions of quadratic metric-affine gravity are also analised by Baykal [11]. Using the null coframe formalism, the author introduces a new family of impulsive gravitational wave solutions in four dimensions for a specific Lagrangian.

## **1.2** Gravitational and massless neutrino fields

The non-Riemannian solutions of metric-affine gravity considered in this thesis and in [72, 73, 74, 75, 76, 77] are presented in the form of generalised pp-waves of parallel Ricci curvature. As stated in [74], classical pp-waves with parallel Ricci curvature are the Riemannian representative within the class of solutions of metric-affine gravity, called generalised pp-waves of parallel Ricci curvature. Torsion and torsion generated curvature of generalised pp-waves can be considered as waves traveling at the speed of light and classical pp-waves with parallel Ricci curvature are only a "gravitational imprint" created by a wave of some massless matter field. Therefore, it is of interest to compare these with solutions of the classical Einstein-Weyl theory, which describes the interaction between gravitational fields and massless neutrino fields.

The massless Dirac operator describes a massless neutrino living in a compact space. In this thesis we perform the spectral analysis of the massless Dirac operator on a 3-dimensional manifold. An advanced review of the theory of the Dirac operator in general can be found in e.g. [60]. An interesting analysis of the Dirac equation on 4-dimensional manifolds without boundary was done in [33], where the authors gave a non geometrical representation of the massless Dirac equation. The eigenvalues of the massless Dirac operator represent the energy levels of the massless particle and we are interested in studying the spectrum of that operator, i.e. the set of all eigenvalues of the operator. To our knowledge, the first explicit calculation of the spectrum of the massless Dirac operator for the flat torus was done by Friedrich [35]. In the same paper, the dependence of the spectrum on the choice of the spin structure was also shown. The spectrum of the massless Dirac operator on the sphere  $S^n$  can also be explicitly calculated, see e.g. Trautman [93] and Bär [10]. It turns out that in these two cases the spectrum is symmetric about zero. However, the analysis of Atiyah et al. [3, 4, 5, 6] shows that for the oriented Riemann 3-manifold there is no physical reason for the spectrum to be symmetric. It would mean that in these two cases there is no difference between the properties of the massless neutrino and the massless antineutrino. Therefore, the objective of our study is to break the spectral symmetry of the massless Dirac operator on the unit 3-torus and the unit 3-sphere.

A very significant result in this field was given by Vassiliev et al. [24] who managed to break the spectral symmetry of the massless Dirac operator on the 3-torus considering the eigenvalue  $\lambda = 0$ . In [24] the authors use the perturbations of the Euclidean metric and derive an asymptotic formula for the eigenvalue zero, i.e. the eigenvalue of the massless Dirac operator with the smallest modulus. The authors show that it is possible to choose a perturbation of the Euclidean metric to shift the eigenvalue zero and to obtain spectral asymmetry. In the same paper, the authors analyse the eta invariant  $\eta_H(0)$ , see e.g. [3, 4, 5, 6], of a first-order self-adjoint elliptic  $m \times m$  matrix classical pseudo-differential operator H as a measure of spectral asymmetry of the operator. In the case of a finite number of eigenvalues the eta invariant is an integer and it is the number of positive eigenvalues minus the number of negative eigenvalues. For the perturbed massless Dirac operator the limit formula for the eta invariant is derived.

Using a similar approach as in [24], in this thesis we aim to derive the asymptotic formulae for the next two eigenvalues considering unit 3-torus in the so-called axisymmetric case, see Section 3.5, i.e. for the eigenvalues  $\lambda = \pm 1$ .

In the future, it will be interesting to derive the asymptotic formulae for the eigenvalues of the massless Dirac operator acting on 3-sphere and see if and when it is possible to obtain the spectral asymmetry.

## **1.3** Quadratic metric-affine gravity

We consider spacetime to be a connected real 4-manifold M equipped with a Lorentzian metric g and an affine connection  $\Gamma$ . The 10 independent components of the symmetric metric tensor  $g_{\mu\nu}$  and the 64 connection coefficients  $\Gamma^{\lambda}{}_{\mu\nu}$  are the unknowns of our theory. We define the action as

$$S := \int q(R), \tag{1.1}$$

where q is O(1,3) invariant quadratic form on curvature R (1.17). Note that the coefficients of this quadratic form depend only on the metric. Independent variation of the action (1.1) with respect to the metric g and the connection  $\Gamma$  produces the system of Euler-Lagrange equations and symbolically we write

$$\partial S/\partial g = 0, \tag{1.2}$$

$$\partial S/\partial \Gamma = 0. \tag{1.3}$$

The system (1.2), (1.3) is system of 10 + 64 partial differential equations with 10 + 64 real unknown components of metric and connection tensor. We use a purely quadratic Lagrangian because we are hoping to describe phenomena whose characteristic wavelength is sufficiently small and curvature sufficiently large, see [101]. The beginning of the development of this theory we find in the work of Hermann Weyl [103]. He stipulated that the most natural gravitational action should be quadratic in curvature and involve all possible invariant quadratic combinations of curvature. The quadratic curvature Lagrangians were also considered by many authors, see [9, 26, 52, 57, 58, 78, 90, 99, 100, 101].

Since the curvature has the 11 irreducible pieces, see Section 1.4.1, the quadratic form q(R) can be represented as

$$q^{(11)}(R) := \sum_{i=1}^{11} c_i(R^{(i)}, R^{(i)})_{YM}, \qquad (1.4)$$

for some real constants  $c_i$ , where  $(\cdot, \cdot)_{YM}$  is Yang-Mills inner product

$$(R,Q)_{\rm YM} := R^{\kappa}{}_{\lambda\mu\nu}Q^{\lambda}{}_{\kappa}{}^{\mu\nu}.$$
(1.5)

This representation of the quadratic form, with 11  $R^2$  terms was considered by Vassiliev [99]. However, as Vassiliev also stated in [101], there are in fact sixteen ways of squaring the irreducible pieces to a scalar because some of the irreducible pieces are isomorphic. Respecting this argument, the quadratic form q(R) can be represented as

$$q^{(16)}(R) := b_1 \mathcal{R}^2 + b_1^* \mathcal{R}_*^2 + \sum_{l,m=1}^3 b_{6lm}(\mathcal{A}^{(l)}, \mathcal{A}^{(m)}) + \sum_{l,m=1}^2 b_{9lm}(\mathcal{S}^{(l)}, \mathcal{S}^{(m)}) + \sum_{l,m=1}^2 b_{9lm}^*(\mathcal{S}^{(l)}_*, \mathcal{S}^{(m)}_*) + b_{10}(R^{(10)}, R^{(10)})_{\rm YM} + b_{30}(R^{(30)}, R^{(30)})_{\rm YM},$$
(1.6)

with some real constants  $b_1$ ,  $b_1^*$ ,  $b_{6lm} = b_{6ml}$ ,  $b_{9lm} = b_{9ml}$ ,  $b_{9lm}^* = b_{9ml}^*$ ,  $b_{10}$ ,  $b_{30}$ . This quadratic form (1.6) has 16  $R^2$  terms and the scalars  $\mathcal{R}$ ,  $\mathcal{R}_*$ , and the tensors  $\mathcal{A}^{(l)}$ ,  $\mathcal{S}^{(l)}$ ,  $\mathcal{S}^{(l)}_*$ ,  $R^{(10)}$  and  $R^{(30)}$  are defined in Section 1.4.1. The inner products in (1.6) are defined with (1.5) and

$$(R,Q) := R_{\mu\nu}Q^{\mu\nu}.$$

**Remark 1.3.1.** The action (1.1) is conformally invariant, i.e. it does not change if we perform a Weyl rescaling of the metric  $g \to e^{2f}g$ ,  $f: M \to \mathbb{R}$ , without changing the connection  $\Gamma$ .

The development of this theory begins with the Yang-Mills theory. The Yang-Mills action for the affine connection is a special case of (1.1) with

$$q_{YM}(R) := R^{\kappa}{}_{\lambda\mu\nu}R^{\lambda\ \mu\nu}{}_{\kappa}.$$
(1.7)

For the quadratic form (1.7), the so-called Yang-Mills equation (1.3) was first analised by Yang [106]. He specialized the equation (1.3) for the Levi-Civita connection and he got the equation

$$\nabla_{\lambda} Ric_{\kappa\mu} - \nabla_{\kappa} Ric_{\lambda\mu} = 0, \qquad (1.8)$$

where by Ric we denote the Ricci curvature (1.18).

**Remark 1.3.2.** The Yang-Mills action can also be obtained from (1.4) by choosing that  $c_1 = \ldots = c_{11} = 1$ .

Depending on the type of the connection of spacetime, we consider two types of solutions of the system (1.2), (1.3), namely *Riemannian* and *non-Riemannian* solutions.

**Definition 1.3.3.** We call a spacetime  $\{M, g, \Gamma\}$  Riemannian if the connection is Levi-Civita, i.e.  $\Gamma^{\lambda}_{\mu\nu} = \{ {}^{\lambda}_{\mu\nu} \}$ , and non-Riemannian otherwise.

**Remark 1.3.4.** When we look for Riemannian solutions of the system (1.2), (1.3), we are still varying the action (1.1) independently with respect to the metric and with respect to the connection and only after we finish the variations we use the fact that connection is Levi-Civita.

The difference between the model with the quadratic form (1.4) and the model with the quadratic form (1.6) can be seen if one considers the specialisation of the equation (1.3) to the Levi-Civita connection. For the eleven parameter action equation (1.3) reduces to equation (1.8) whereas for the sixteen parameter action (1.6) equation (1.3) reduces to

$$\nabla Ric = 0. \tag{1.9}$$

The equations (1.8) and (1.9) are different and equation (1.9) is much more restrictive.

An important class of Riemannian solutions are so-called *Einstein spaces*.

**Definition 1.3.5.** An Einstein space is a Riemannian spacetime with  $Ric = \Lambda g$  where Ric is Ricci curvature (1.18) and  $\Lambda$  is some real constant.

Examining the equation (1.8), Yang concluded that Einstein spaces satisfy equation (1.8). It was shown later by many authors that Einstein spaces are satisfying the system (1.2), (1.3) for the quadratic form (1.7), see e.g. Yang [106] and Mielke [62]. We therefore refer to the special case (1.7) of the field equations (1.2), (1.3) as the Yang-Mielke theory of gravity. There are many works devoted to the study of the system (1.2), (1.3) in the special case (1.7)and one can get an idea of the historical development of the Yang-Mielke theory of gravity from [31, 32, 71, 79, 90, 92, 91, 105].

Vassiliev [101] solved the problem of existence and uniqueness of solutions of the system (1.2), (1.3) for the most general case of quadratic action with 16  $R^2$  terms. In the same paper, it was shown that Einstein spaces, pp-waves with parallel Ricci curvature (see Section 2.1) and Riemannian spacetimes which have zero scalar curvature and are locally a product of Einstein 2manifolds are the *only* Riemannian solutions of the system (1.2), (1.3) for the most general quadratic form (1.6). Hence, all new solutions of the system (1.2), (1.3) that we can find are non-Riemannian solutions.

We are particularly interested in the analysis of a class of spacetimes called *pseudoinstantons*, which were introduced by Vassiliev in [99].

**Definition 1.3.6.** We call a spacetime  $\{M, g, \Gamma\}$  a pseudoinstanton if the connection is metric-compatible and curvature is irreducible and simple.

The metric compatibility means that  $\nabla g = 0$ , where  $\nabla$  denotes the covariant derivative (1.15). Irreducibility of curvature means that only one of eleven irreducible pieces of curvature is nonzero and all others are identically equal to zero. Simplicity means that the given irreducible subspace that provides the non-zero piece of curvature is not isomorphic to any other irreducible subspace. Hence, there are only three types of pseudoinstantons:

- *scalar pseudoinstanton*, where only scalar curvature is not identically zero;
- *pseudoscalar pseudoinstanton*, where only pseudoscalar curvature is not identically zero;
- *Weyl pseudoinstanton*, where only Weyl curvature is not identically zero.

Pseudoinstantons are a very important class of spacetimes because they represent solutions of the field equations (1.2), (1.3) for the most general quadratic form (1.6), as was proved by Vassiliev [99]. For the construction of one non-Riemannian pseudoinstanton in Minkowski spacetime, see [101].

# 1.4 Notation and background

The notation in this thesis follows [53, 73, 74, 75, 76, 77, 99, 101]. We denote local coordinates by  $x^{\mu}$ ,  $\mu = 0, 1, 2, 3$ , and write  $\partial_{\mu} := \partial/\partial x^{\mu}$ . We define the covariant derivative of a vector field as

$$\nabla_{\mu}v^{\lambda} := \partial_{\mu}v^{\lambda} + \Gamma^{\lambda}_{\ \mu\nu}v^{\nu}, \quad \nabla_{\mu}v_{\lambda} := \partial_{\mu}v_{\lambda} - \Gamma^{\nu}_{\ \mu\lambda}v_{\nu}, \tag{1.10}$$

where  $\Gamma^{\lambda}_{\ \mu\nu}$  are the connection coefficients. The Christoffel symbol is

$$\{\Gamma\}^{\lambda}{}_{\mu\nu} = \begin{cases} \alpha \\ \beta\gamma \end{cases} := \frac{1}{2} g^{\lambda\kappa} (\partial_{\mu} g_{\nu\kappa} + \partial_{\nu} g_{\mu\kappa} - \partial_{\kappa} g_{\mu\nu}). \tag{1.11}$$

We denote by  $\{\nabla\}$  the covariant derivative with respect to the Levi-Civita connection, i.e.

$$\{\nabla\}_{\mu}v^{\lambda} := \partial_{\mu}v^{\lambda} + \{\Gamma\}^{\lambda}{}_{\mu\nu}v^{\nu}.$$

We define torsion as

$$T^{\lambda}_{\ \mu\nu} := \Gamma^{\lambda}_{\ \mu\nu} - \Gamma^{\lambda}_{\ \nu\mu} \tag{1.12}$$

and contortion as

$$K^{\lambda}_{\ \mu\nu} := \frac{1}{2} \left( T^{\lambda}_{\ \mu\nu} + T^{\ \lambda}_{\mu\ \nu} + T^{\ \lambda}_{\nu\ \mu} \right). \tag{1.13}$$

Torsion can be expressed by contortion as

$$T^{\lambda}_{\ \mu\nu} = K^{\lambda}_{\ \mu\nu} - K^{\lambda}_{\ \nu\mu}.$$

The Levi-Civita connection  $\{\Gamma\}$  and full connection  $\Gamma$  are related as

$$\Gamma^{\lambda}{}_{\mu\nu} = \{\Gamma\}^{\lambda}{}_{\mu\nu} + K^{\lambda}{}_{\mu\nu}.$$
(1.14)

We say that our connection  $\Gamma$  is *metric compatible* if  $\nabla g \equiv 0$ , i.e.

$$\nabla_{\lambda}g_{\mu\nu} = \partial_{\lambda}g_{\mu\mu} - \Gamma^{\kappa}{}_{\lambda\mu}g_{\kappa\nu} - \Gamma^{\kappa}{}_{\lambda\nu}g_{\mu\kappa} = 0.$$
(1.15)

We define *nonmetricity* Q by

$$Q_{\mu\alpha\beta} := \nabla_{\mu} g_{\alpha\beta}. \tag{1.16}$$

The curvature tensor is defined in terms of the affine connection as

$$R^{\kappa}_{\ \lambda\mu\nu} := \partial_{\mu}\Gamma^{\kappa}_{\ \nu\lambda} - \partial_{\nu}\Gamma^{\kappa}_{\ \mu\lambda} + \Gamma^{\kappa}_{\ \mu\eta}\Gamma^{\eta}_{\ \nu\lambda} - \Gamma^{\kappa}_{\ \nu\eta}\Gamma^{\eta}_{\ \mu\lambda}, \qquad (1.17)$$

Ricci curvature as

$$Ric_{\lambda\nu} := R^{\kappa}_{\ \lambda\kappa\nu},\tag{1.18}$$

scalar curvature as

$$\mathcal{R} := Ric^{\kappa}{}_{\kappa} \tag{1.19}$$

and trace-free Ricci curvature as

$$\mathcal{R}ic := Ric - \frac{1}{4}\mathcal{R}g.$$

We denote Weyl curvature by  $\mathcal{W}$ . Weyl curvature is the irreducible piece of curvature defined by the conditions

$$R_{\kappa\lambda\mu\nu} = R_{\mu\nu\kappa\lambda},$$
  

$$\varepsilon^{\kappa\lambda\mu\nu}R_{\kappa\lambda\mu\nu} = 0,$$
  

$$Ric = 0.$$
  
(1.20)

**Remark 1.4.1.** The torsion and curvature can be written also in *anholonomic* notation which differs to the holonomic notation used in this thesis. The torsion is defined as

$$T^{\alpha} := D \vartheta^{\alpha} = \frac{1}{2} T_{ij}{}^{\alpha} dx^{i} \wedge dx^{j}$$

and the curvature as

$$R^{\alpha\beta} := d\Gamma^{\alpha\beta} - \Gamma^{\alpha}{}_{\gamma} \wedge \Gamma^{\gamma\beta} = \frac{1}{2} R_{ij}{}^{\alpha\beta} dx^i \wedge dx^j,$$

where  $\vartheta^{\alpha}$  is orthonormal frame and  $\Gamma_{\alpha}{}^{\beta}$  is linear connection. For more about the anholonomic notation of curvature and torsion see Chapter 2 of [13].

We raise and lower tensor indices in the standard way, i.e.  $g_{\alpha\beta}v^{\beta} = v_{\alpha}$ ,  $g^{\alpha\beta}v_{\beta} = v^{\alpha}$ . We define the action of the Hodge star on a rank q antisymmetric tensor as

$$(*Q)_{\mu_{q+1}\dots\mu_{4}} := (q!)^{-1} \sqrt{|\det g|} \ Q^{\mu_{1}\dots\mu_{q}} \varepsilon_{\mu_{1}\dots\mu_{4}} , \qquad (1.21)$$

where  $\varepsilon$  is the totally antisymmetric quantity and  $\varepsilon_{0123} := +1$ . When we apply the Hodge star to curvature we have a choice between acting either on the first or the second pair of indices, so we introduce two different operators: the left Hodge star

$$({}^{*}R)_{\kappa\lambda\mu\nu} := \frac{1}{2}\sqrt{|\det g|} \varepsilon^{\kappa'\lambda'}{}_{\kappa\lambda} R_{\kappa'\lambda'\mu\nu}$$
(1.22)

and the right Hodge star

$$(R^*)_{\kappa\lambda\mu\nu} := \frac{1}{2}\sqrt{|\det g|} R_{\kappa\lambda\mu'\nu'} \varepsilon^{\mu'\nu'}{}_{\mu\nu}. \qquad (1.23)$$

Given a scalar function  $f: M \to R$  we write for brevity

$$\int f := \int f \sqrt{|\det g|} \mathrm{d}x^0 \mathrm{d}x^1 \mathrm{d}x^2 \mathrm{d}x^3, \quad \det g := \det(g_{\mu\nu}).$$

#### 1.4.1 Irreducible pieces of torsion and curvature

In this section we provide the details of the irreducible pieces of torsion and curvature, where we mostly follow the exposition from [72, 99, 101].

#### Irreducible pieces of torsion

Under the local Lorentz group, the torsion T with its 24 independent components, can be irreducibly decomposed into three pieces. According to [51, 99], the irreducible pieces of torsion are

$$T^{(1)} = T - T^{(2)} - T^{(3)}, (1.24)$$

$$T^{(2)}{}_{\lambda\mu\nu} = g_{\lambda\mu}v_{\nu} - g_{\lambda\nu}v_{\mu}, \qquad (1.25)$$

$$T^{(3)} = *w,$$
 (1.26)

where

$$v_{\nu} = \frac{1}{3} T^{\lambda}_{\ \lambda\nu}, \quad w_{\nu} = \frac{1}{6} \sqrt{|\det g|} T^{\kappa\lambda\mu} \varepsilon_{\kappa\lambda\mu\nu}. \tag{1.27}$$

The pieces  $T^{(1)}, T^{(2)}$  i  $T^{(3)}$  are called *tensor torsion* with its 16 independent components, *trace torsion* with its 4 independent components and *axial torsion* with its 4 independent components.

We define the action of the Hodge star on torsion as

$$(*T)_{\lambda\mu\nu} := \frac{1}{2}\sqrt{|\det g|}T_{\lambda\xi\eta}\varepsilon^{\xi\eta}{}_{\mu\nu}.$$
(1.28)

The Hodge star maps tensor torsion to tensor torsion, trace to axial, and axial to trace:

$$(*T)^{(1)} = *(T^{(1)}),$$
  

$$(*T)^{(2)}{}_{\lambda\mu\nu} = g_{\lambda\mu}w_{\nu} - g_{\lambda\nu}w_{\mu},$$
  

$$(*T)^{(3)} = -*v.$$
(1.29)

The decomposition described above assumes torsion to be real and metric to be Lorentzian.

**Remark 1.4.2.** The Hodge star which appears in the RHS's of the formulae (1.26) and (1.29) is the standard Hodge star (1.21) and differs from the Hodge star on torsions (1.28).

Using formulae (1.12), (1.13) and (1.24)-(1.26), we obtain the irreducible decomposition of contortion

$$K^{(1)} = K - K^{(2)} - K^{(3)},$$
  

$$K^{(2)}{}_{\lambda\mu\nu} = g_{\lambda\mu}v_{\nu} - g_{\nu\mu}v_{\lambda},$$
  

$$K^{(3)} = \frac{1}{2} * w,$$

where

$$w_{\nu} = \frac{1}{3} K^{\lambda}_{\ \lambda\nu}, \quad w_{\nu} = \frac{1}{3} \sqrt{|\det g|} K^{\kappa\lambda\mu} \varepsilon_{\kappa\lambda\mu\nu}.$$

The irreducible pieces of torsion and contortion are related as<sup>1</sup>

$$T^{(1)}{}_{\kappa\lambda\mu} = K^{(1)}{}_{\lambda\kappa\mu}, \ T^{(2)}{}_{\kappa\lambda\mu} = K^{(2)}{}_{\lambda\kappa\mu}, \ T^{(3)}{}_{\kappa\lambda\mu} = 2K^{(3)}{}_{\kappa\lambda\mu}.$$

#### Irreducible pieces of curvature - version 1

In this section we present one decomposition of the curvature tensor generated by a general affine connection, where we follow the exposition from [99]. We denote by **R** the 96-dimensional vector space of real rank 4 tensors  $R^{\kappa}_{\lambda\mu\nu}$ . A curvature generated by a general affine connection has only one antisymmetry which satisfy the condition

$$R^{\kappa}_{\ \lambda\mu\nu} = -R^{\kappa}_{\ \lambda\nu\mu}.\tag{1.30}$$

Let g be Lorentzian metric at the point  $x \in M$  and let O(1,3) be the corresponding full Lorentz group. The vector space **R** decomposes into a direct sum of 11 subspaces which are invariant and irreducible under the action of O(1,3), see [51]. According to Vassiliev [99], we have the orthogonal decomposition  $\mathbf{R} = \mathbf{R}^+ \oplus \mathbf{R}^-$  where

$$\mathbf{R}^{\pm} = \{ R \in \mathbf{R} | R_{\kappa\lambda\mu\nu} = \pm R_{\lambda\kappa\mu\nu} \}$$

and dim  $\mathbf{R}^+ = 60$  and dim  $\mathbf{R}^- = 36$ . The subspaces  $\mathbf{R}^+$  and  $\mathbf{R}^-$  decompose further into five and six irreducible subspaces, respectively. The vector space  $\mathbf{R}^-$  is the vector space of curvatures generated by metric compatible connections and it can be written as  $\mathbf{R}^- = \bigoplus_{a,b=\pm} \mathbf{R}^-_{ab}$  where

$$\mathbf{R}_{ab}^{-} = \{ R \in \mathbf{R}^{-} | R^{T} = aR, \ ^{*}R^{*} = bR \},$$

<sup>&</sup>lt;sup>1</sup>Note the order of indices.

where the map  $R \to R^T$  defined with  $(R^T)_{\kappa\lambda\mu\nu} := R_{\mu\nu\kappa\lambda}$  is endomorphism in  $\mathbf{R}^-$ . The map  $R \to R^T$  is called *transposition* and the so-called *double duality map*  $R \to {}^*R^*$  is defined with

$${}^{*}R^{*} := ({}^{*}R)^{*} = {}^{*}(R^{*}),$$

where  $R \to {}^{*}R$  and  $R \to R^{*}$  are left Hodge star (1.22) and right Hodge star (1.23) respectively.

**Remark 1.4.3.** The subspaces  $\mathbf{R}_{++}^-$ ,  $\mathbf{R}_{+-}^-$ ,  $\mathbf{R}_{-+}^-$  and  $\mathbf{R}_{--}^-$  are mutually orthogonal and their dimensions are dim  $\mathbf{R}_{++}^- = \dim \mathbf{R}_{-+}^- = 9$ , dim  $\mathbf{R}_{+-}^- = 12$  and dim  $\mathbf{R}_{--}^- = 6$ . The subspaces  $\mathbf{R}_{++}^-$ ,  $\mathbf{R}_{-+}^-$ ,  $\mathbf{R}_{--}^-$  are irreducible and the only subspace which decomposes further is  $\mathbf{R}_{+-}^-$  as

$$\mathbf{R}^{-}_{+-} = \mathbf{R}_{\text{scalar}} \oplus \mathbf{R}_{\text{Weyl}} \oplus \mathbf{R}_{\text{pseudoscalar}}.$$

We now denote the subspaces  $\mathbf{R}_{++}^-$ ,  $\mathbf{R}_{\text{scalar}}$ ,  $\mathbf{R}_{\text{Weyl}}$ ,  $\mathbf{R}_{\text{pseudoscalar}}$ ,  $\mathbf{R}_{-+}^-$ ,  $\mathbf{R}_{--}^-$  by  $\mathbf{R}^{(j)}$ ,  $j = 1, \ldots, 6$ , respectively.

The described decomposition assumes curvature to be real and metric to be Lorentzian. The explicit formulae for the six pieces of the curvature generated by metric compatible connections are

$$R^{(1)} = \frac{1}{2} (g_{\kappa\mu} \overline{\mathcal{R}ic}_{\lambda\nu} - g_{\lambda\mu} \overline{\mathcal{R}ic}_{\kappa\nu} - g_{\kappa\nu} \overline{\mathcal{R}ic}_{\lambda\mu} + g_{\lambda\nu} \overline{\mathcal{R}ic}_{\kappa\mu}), \qquad (1.31)$$

$$R^{(2)} = \frac{1}{12} (g_{\kappa\mu} g_{\lambda\nu} - g_{\lambda\mu} g_{\kappa\nu}) \mathcal{R}, \qquad (1.32)$$

$$R^{(3)} = \bar{R} - R^{(1)} - R^{(2)} - R^{(4)}, \qquad (1.33)$$

$$R^{(4)} = -\frac{1}{24}\sqrt{|\det g|} \varepsilon_{\kappa\lambda\mu\nu}\vec{\mathcal{R}},\tag{1.34}$$

$$R^{(5)} = \hat{R} - R^{(6)}, \tag{1.35}$$

$$R^{(6)} = \frac{1}{2} (g_{\kappa\mu} \widehat{Ric}_{\lambda\nu} - g_{\lambda\mu} \widehat{Ric}_{\kappa\nu} - g_{\kappa\nu} \widehat{Ric}_{\lambda\mu} + g_{\lambda\nu} \widehat{Ric}_{\kappa\mu}, \qquad (1.36)$$

where

$$\bar{R}_{\kappa\lambda\mu\nu} = \frac{1}{4} (R_{\kappa\lambda\mu\nu} - R_{\lambda\kappa\mu\nu} + R_{\mu\nu\kappa\lambda} - R_{\nu\mu\kappa\lambda}),$$
$$\hat{R}_{\kappa\lambda\mu\nu} = \frac{1}{4} (R_{\kappa\lambda\mu\nu} - R_{\lambda\kappa\mu\nu} - R_{\mu\nu\kappa\lambda} + R_{\nu\mu\kappa\lambda}),$$
$$\overline{Ric}_{\lambda\nu} = \bar{R}^{\kappa}_{\ \lambda\kappa\nu}, \quad \mathcal{R} = \overline{Ric}^{\lambda}_{\ \lambda} = R^{\kappa\lambda}_{\ \kappa\lambda}, \quad \overline{\mathcal{R}ic}_{\lambda\nu} = \overline{Ric}_{\lambda\nu} - \frac{1}{4}g_{\lambda\nu}\mathcal{R},$$
$$\widehat{Ric} = \hat{R}^{\kappa}_{\ \lambda\kappa\nu}, \quad \breve{\mathcal{R}} = \sqrt{|\det g|}\varepsilon^{\kappa\lambda\mu\nu}\bar{R}_{\kappa\lambda\mu\nu} = \sqrt{|\det g|}\varepsilon^{\kappa\lambda\mu\nu}R_{\kappa\lambda\mu\nu}.$$

Note that in the Riemannian case curvature has only three irreducible pieces  $R^{(1)}, R^{(2)}$  and  $R^{(3)}$ .

#### Irreducible pieces of curvature-version 2

The vector space  $\mathbf{R}$  can be decomposed in a different way, following the exposition from [72, 101]. The subspaces of the vector space  $\mathbf{R}$  are:

- the two subspaces of dimension 1 noted by  $\mathbf{R}^{(1)}$  and  $\mathbf{R}^{(1)}_{*}$ ,
- the three subspaces of dimension 6 noted by  $\mathbf{R}^{(6,l)}$ , (l = 1, 2, 3),
- the four subspaces of dimension 9 noted by  $\mathbf{R}^{(9,l)}$  and  $\mathbf{R}^{(9,l)}_{*}$ , (l=1,2),
- the one subspace of dimension 10 noted as  $\mathbf{R}^{(10)}$ , and
- the one subspace of dimension 30 noted as  $\mathbf{R}^{(30)}$ .

Two subspaces are said to be isomorphic if there is a linear bijection between them which commutes with the action of O(1,3). There are three groups of isomorphic subspaces, namely

$$\{\mathbf{R}^{(6,l)}, l = 1, 2, 3\}, \{\mathbf{R}^{(9,l)}, l = 1, 2\}, \{\mathbf{R}^{(9,l)}_{*}, l = 1, 2\}.$$

We provide the explicit formulae of the irreducible subspaces of dimension less than 10:

(a) the subspace  $\mathbf{R}^{(1)}$  has the formula for the curvature

$$R_{\kappa\lambda\mu\nu} = a_1 (g_{\kappa\mu}g_{\lambda\nu} - g_{\kappa\nu}g_{\lambda\mu})\mathcal{R}, \qquad (1.37)$$

(b) the subspace  $\mathbf{R}^{(1)}_*$  has the formula for the curvature

$$(R^*)_{\kappa\lambda\mu\nu} = a_1^* (g_{\kappa\mu}g_{\lambda\nu} - g_{\kappa\nu}g_{\lambda\mu})\mathcal{R}_*, \qquad (1.38)$$

(c) the subspaces  $\mathbf{R}^{(6,l)}$ , (l = 1, 2, 3) have the formula for the curvature

$$R_{\kappa\lambda\mu\nu} = a_{6l1} (g_{\kappa\mu} \mathcal{A}^{(l)}{}_{\lambda\nu} - g_{\kappa\nu} \mathcal{A}^{(l)}{}_{\lambda\mu}) + a_{6l2} (g_{\lambda\mu} \mathcal{A}^{(l)}{}_{\kappa\nu} - g_{\lambda\nu} \mathcal{A}^{(l)}{}_{\kappa\mu}) + a_{6l3} g_{\kappa\lambda} \mathcal{A}^{(l)}{}_{\mu\nu}, \qquad (1.39)$$

(d) the subspaces  $\mathbf{R}^{(9,l)}$ , (l=1,2) have the formula for the curvature

$$R_{\kappa\lambda\mu\nu} = a_{9l1} (g_{\kappa\mu} \mathcal{S}^{(l)}{}_{\lambda\nu} - g_{\kappa\nu} \mathcal{S}^{(l)}{}_{\lambda\mu}) + a_{9l2} (g_{\lambda\mu} \mathcal{S}^{(l)}{}_{\kappa\nu} - g_{\lambda\nu} \mathcal{S}^{(l)}{}_{\kappa\mu}), \quad (1.40)$$

(e) the subspaces  $\mathbf{R}^{(9,l)}_{*}$ , (l=1,2) have the formula for the curvature

$$(R^*)_{\kappa\lambda\mu\nu} = a^*_{9l1}(g_{\kappa\mu}\mathcal{S}^{(l)}_{*\ \lambda\nu} - g_{\kappa\nu}\mathcal{S}^{(l)}_{*\ \lambda\mu}) + a^*_{9l2}(g_{\lambda\mu}\mathcal{S}^{(l)}_{*\ \kappa\nu} - g_{\lambda\nu}\mathcal{S}^{(l)}_{*\ \kappa\mu}), \ (1.41)$$

where  $a_1 = a_1^* = 1/12$  and

$$(a_{6lm}) = \begin{pmatrix} 5/12 & -1/12 & -1/6 \\ -1/12 & 5/12 & -1/6 \\ -1/12 & -1/12 & 1/3 \end{pmatrix}, \ (a_{9lm}) = (a_{9lm}^*) = \begin{pmatrix} 3/8 & -1/8 \\ -1/8 & 3/8 \end{pmatrix}.$$

The scalars  $\mathcal{R}, \mathcal{R}_*$  and the tensors  $\mathcal{A}^{(l)}, \mathcal{S}^{(l)}, \mathcal{S}^{(l)}_*$  appearing in formulae (1.37)-(1.41) are expressed via the full curvature tensor R according to following formulae:

$$\begin{split} \mathcal{R} &:= R^{\kappa\lambda}{}_{\kappa\lambda}, \\ Ric^{(1)}{}_{\mu\nu} &:= R^{\kappa}{}_{\mu\kappa\nu}, \\ \mathcal{R}ic^{(1)}{}_{i} &:= Ric^{(1)} - \frac{1}{4}\mathcal{R}g, \\ \mathcal{S}^{(l)}{}_{\mu\nu} &:= \frac{\mathcal{R}ic^{(l)}{}_{\mu\nu} + \mathcal{R}ic^{(l)}{}_{\nu\mu}}{2}, \\ \mathcal{A}^{(l)}{}_{\mu\nu} &:= \frac{\mathcal{R}ic^{(l)}{}_{\mu\nu} - \mathcal{R}ic^{(l)}{}_{\nu\mu}}{2}, \\ \mathcal{A}^{(3)}{}_{\mu\nu} &:= R^{\kappa}{}_{\kappa\mu\nu}, \end{split}$$

and

$$\begin{aligned} \mathcal{R}_{*} &:= (R^{*})^{\kappa\lambda}_{\ \kappa\lambda}, \\ Ric_{*}^{(1)}{}_{\mu\nu} &:= (R^{*})^{\kappa}{}_{\mu\kappa\nu}, \\ \mathcal{R}ic_{*}^{(1)}{}_{\mu\nu} &:= (R^{*})^{\kappa}{}_{\mu\kappa\nu}, \\ \mathcal{R}ic_{*}^{(1)} &:= Ric_{*}^{(1)} - \frac{1}{4}\mathcal{R}_{*}g, \\ \mathcal{S}_{*}^{(l)}{}_{\mu\nu} &:= \frac{\mathcal{R}ic_{*}^{(l)}{}_{\mu\nu} + \mathcal{R}ic_{*}^{(l)}{}_{\nu\mu}}{2}, \\ \mathcal{A}_{*}^{(3)}{}_{\mu\nu} &:= (R^{*})^{\kappa}{}_{\kappa\mu\nu}. \end{aligned}$$

**Remark 1.4.4.** Note that the tensor  $Ric^{(1)}_{\mu\nu}$  and the Ricci tensor (1.18) are equivalent.

**Remark 1.4.5.** The tensors  $\mathcal{A}_*^{(l)}$  does not appear in formulae (1.37)-(1.41). This is because the tensors  $\mathcal{A}^{(l)}$  and  $\mathcal{A}_*^{(l)}$  are not independent and the tensors  $\mathcal{A}^{(l)}$  are linear combinations of the Hodge duals of the tensors  $\mathcal{A}_*^{(l)}$ , and vice versa, see [101]. Now we can write

$$\mathcal{A}_{*}^{(1)} = \alpha_1 * \mathcal{A}^{(1)} + \alpha_2 * \mathcal{A}^{(2)}, \qquad (1.42)$$

$$\mathcal{A}_{*}^{(2)} = \beta_1 * \mathcal{A}^{(1)} + \beta_2 * \mathcal{A}^{(2)}, \qquad (1.43)$$

where  $\alpha_1, \alpha_2, \beta_1, \beta_2$  are an arbitrary scalars.

The subspace  $\mathbf{R}^{(10)}$  is the subspace of curvatures R such that all possible traces are equal to zero, i.e.

$$R^{\kappa}_{\ \lambda\kappa\nu} = (R^{*})^{\kappa}_{\ \lambda\kappa\nu} = 0, \quad R^{\ \kappa}_{\mu\ \kappa\nu} = (R^{*})^{\ \kappa}_{\mu\ \kappa\nu} = 0, \quad R^{\kappa}_{\ \kappa\mu\nu} = 0$$
(1.44)

and  $R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu}$ . The subspace  $\mathbf{R}^{(30)}$  is the subspace of curvatures R such that it satisfies the relations (1.44) and  $R_{\kappa\lambda\mu\nu} = R_{\lambda\kappa\mu\nu}$ . Using the described subspaces, the space  $\mathbf{R}$  has a following decomposition

$$\mathbf{R} = \mathbf{R}^{(1)} \oplus \mathbf{R}_{*}^{(1)} \oplus_{i=1}^{3} \mathbf{R}^{(6,l)} \oplus_{i=1}^{2} \mathbf{R}^{(9,l)} \oplus_{i=1}^{2} \mathbf{R}_{*}^{(9,l)} \oplus \mathbf{R}^{(10)} \oplus \mathbf{R}^{(30)}$$

and it means that arbitrary element  $R \in \mathbf{R}$  can be uniquely written as

$$R = R^{(1)} + R^{(1)}_* + \sum_{l=1}^3 R^{(6,l)} + \sum_{l=1}^2 R^{(9,l)} + \sum_{l=1}^2 R^{(9,l)}_* + R^{(10)} + R^{(30)},$$

where the R's in the RHS in the formulae (1.37)-(1.41).

**Remark 1.4.6.** We call the subspaces  $\mathbf{R}^{(1)}$ ,  $\mathbf{R}^{(1)}_*$ ,  $\mathbf{R}^{(10)}$ ,  $\mathbf{R}^{(30)}$  simple because they are not isomorphic to any other subspaces. Accordingly, we call the irreducible pieces  $R^{(1)}$ ,  $R^{(1)}_*$ ,  $R^{(10)}$ ,  $R^{(30)}$  simple.

#### 1.4.2 On the Levi-Civita tensor

A very often used tensor in linear algebra, tensor analysis and differential geometry is so-called *Levi-Civita tensor*.

**Definition 1.4.7.** The Levi-Civita tensor is defined as

$$\epsilon_{\kappa\lambda\mu\nu} := \sqrt{|\det g|} \varepsilon_{\kappa\lambda\mu\nu}, \qquad (1.45)$$

where  $\varepsilon_{\kappa\lambda\mu\nu}$  is totally antisymmetric quantity and  $\varepsilon_{0123} = +1$ .

**Remark 1.4.8.** In this thesis, we are mostly dealing with the Minkowski metric  $g_{\mu\nu} = \text{diag}(+1, -1, -1, -1)$  and the pp-metric (2.1), whose determinant is equal to 1, and in those two cases the Levi-Civita tensor (1.45) is totally antisymmetric quantity  $\varepsilon_{\kappa\lambda\mu\nu}$ .

Lemma 1.4.9. The covariant derivative of Levi-Civita tensor is zero, i.e.

$$\nabla_{\xi} \epsilon_{\kappa\lambda\mu\nu} = 0. \tag{1.46}$$

*Proof.* Since the tensor  $\epsilon_{\kappa\lambda\mu\nu}$  is by definition a totally antisymmetric tensor, it is enough to calculate calculate  $\nabla_{\xi}\epsilon_{0123}$ . Using the definition of the covariant derivative (1.10), we have that

$$\begin{aligned} \nabla_{\xi} \epsilon_{0123} &= \partial_{\xi} \epsilon_{0123} - \Gamma^{\eta}{}_{\xi 0} \epsilon_{\eta 123} - \Gamma^{\eta}{}_{\xi 1} \epsilon_{0\eta 23} - \Gamma^{\eta}{}_{\xi 2} \epsilon_{01\eta 3} - \Gamma^{\eta}{}_{\xi 3} \epsilon_{012\eta} \\ &= \partial_{\xi} \epsilon_{0123} - \Gamma^{0}{}_{\xi 0} \epsilon_{0123} - \Gamma^{1}{}_{\xi 1} \epsilon_{0123} - \Gamma^{2}{}_{\xi 2} \epsilon_{0123} - \Gamma^{3}{}_{\xi 3} \epsilon_{0123} \\ &= \partial_{\xi} \epsilon_{0123} - \sqrt{|\det g|} \Gamma^{\eta}{}_{\xi \eta}. \end{aligned}$$

The partial derivative is

$$\partial_{\xi} \epsilon_{0123} = \partial_{\xi} (\sqrt{|\det g|}) \varepsilon_{0123} + \sqrt{|\det g|} \partial_{\xi} (\varepsilon_{0123})$$
$$= \frac{1}{2} \frac{1}{\sqrt{|\det g|}} \partial_{\xi} |\det g| = \sqrt{|\det g|} \{\Gamma\}^{\eta}_{\xi\eta}$$

since  $\{\Gamma\}^{\eta}_{\xi\eta} = \frac{\partial_{\xi} |\det g|}{2|\det g|}$ . So, using formula (1.14), we get that  $\nabla_{\xi}\epsilon_{0123} = \sqrt{|\det g|} \{\Gamma\}^{\eta}_{\xi\eta} - \sqrt{|\det g|} \Gamma^{\eta}_{\xi\eta} = -\sqrt{|\det g|} K^{\eta}_{\xi\eta}$ . The contention tensor K is antisymmetric in first and third index as

The contortion tensor K is antisymmetric in first and third index and since we have contraction in these two indices, it is equal to zero, and hence we get the equation (1.46).

**Remark 1.4.10.** Direct calculations show that the totally antisymmetric quantity  $\varepsilon_{\kappa\lambda\mu\nu}$  satisfies the following identities:

$$\varepsilon_{\kappa\lambda\mu\nu}\varepsilon^{\kappa\lambda\mu\nu} = -4!,\tag{1.47}$$

$$\varepsilon_{\kappa\lambda\mu\nu}\varepsilon^{\eta\lambda\mu\nu} = -3! \ \delta_{\kappa}{}^{\eta},\tag{1.48}$$

$$\varepsilon_{\kappa\lambda\mu\nu}\varepsilon^{\eta\xi\mu\nu} = -2! \ (\delta_{\kappa}{}^{\eta}\delta_{\lambda}{}^{\xi} - \delta_{\kappa}{}^{\xi}\delta_{\lambda}{}^{\eta}), \tag{1.49}$$

$$\varepsilon_{\kappa\lambda\mu\nu}\varepsilon^{\eta\xi\zeta\nu} = -(\delta_{\kappa}^{\eta}\delta_{\lambda}^{\xi}\delta_{\mu}^{\zeta} + \delta_{\kappa}^{\xi}\delta_{\lambda}^{\zeta}\delta_{\mu}^{\eta} + \delta_{\kappa}^{\zeta}\delta_{\lambda}^{\eta}\delta_{\mu}^{\xi} - \delta_{\kappa}^{\eta}\delta_{\lambda}^{\zeta}\delta_{\mu}^{\xi} - \delta_{\kappa}^{\xi}\delta_{\lambda}^{\eta}\delta_{\mu}^{\zeta} - \delta_{\kappa}^{\zeta}\delta_{\lambda}^{\xi}\delta_{\mu}^{\eta}).$$
(1.50)

The Levi-Civita tensor is also appearing in the explicit formula (B.3) for the piece  $R^{(5)}$ . The relation between the Levi-Civita tensor (1.45) and the Weyl curvature is very interesting. It is known that the product of the Levi-Civita tensor and Weyl curvature with four contractions is equal to zero, see (1.20). It is of interest to see what happens in some other cases when contracting the Levi-Civita and Weyl tensors.

**Lemma 1.4.11.** The Levi-Civita tensor is related to the Weyl tensor in the following way:

$$\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\lambda\eta\xi} = 2\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\eta\lambda\xi},\tag{1.51}$$

$$\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\lambda\mu\xi} = 0. \tag{1.52}$$

*Proof.* Using the well known Bianchi identity for the Weyl curvature

$$\mathcal{W}_{\kappa\lambda\eta\xi} + \mathcal{W}_{\kappa\eta\xi\lambda} + \mathcal{W}_{\kappa\xi\lambda\eta} = 0,$$

we have that

$$\epsilon^{\kappa\lambda\mu
u}(\mathcal{W}_{\kappa\lambda\eta\xi}+\mathcal{W}_{\kappa\eta\xi\lambda}+\mathcal{W}_{\kappa\xi\lambda\eta})=0.$$

Renaming some indices and using the antisymmetry of  $\epsilon_{\kappa\lambda\mu\nu}$  and the known properties of the Weyl tensor, we get that

$$\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\lambda\eta\xi} = -\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\eta\xi\lambda} - \epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\xi\lambda\eta} = 2\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\eta\lambda\xi},$$

which gives us the equation (1.51).

Contracting the indices  $\mu$  and  $\eta$  in (1.51) and using the antisymmetry properties of the Weyl tensor we get  $\epsilon^{\kappa\lambda\mu\nu}\mathcal{W}_{\kappa\lambda\mu\xi} = 0$ , which is exactly equation (1.52).

#### 1.4.3 Spinor formalism

In this section we present the spinor formalism used in this thesis. We use the formalism introduced in [72, 74, 77].

Definition 1.4.12. The 'metric spinor' is defined as

$$\epsilon_{ab} = \epsilon_{\dot{a}\dot{b}} = \epsilon^{ab} = \epsilon^{\dot{a}\dot{b}} = \begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix}, \qquad (1.53)$$

with the first index enumerating rows and the second enumerating columns.

We raise and lower spinor indices according to the formulae

$$\xi^a = \epsilon^{ab} \xi_b, \qquad \xi_a = \epsilon_{ab} \xi^b, \qquad \eta^{\dot{a}} = \epsilon^{\dot{a}\dot{b}} \eta_{\dot{b}}, \qquad \eta_{\dot{a}} = \epsilon_{\dot{a}\dot{b}} \eta^{\dot{b}}. \tag{1.54}$$

In this choice of spinor, the 'contravariant' and 'covariant' metric spinors are raised and lowered versions of each other, i.e.  $\epsilon^{ab} = \epsilon^{ac} \epsilon_{cd} \epsilon^{bd}$  and  $\epsilon_{ab} = \epsilon_{ac} \epsilon^{cd} \epsilon_{bd}$ . Also, the spinor inner product is invariant under the operation of raising and lowering of indices, i.e.  $(\epsilon_{ac} \xi^c) (\epsilon^{ad} \eta_d) = \xi^a \eta_a$ . But, consecutive raising and lowering of a single spinor index leads to a change of sign, i.e.  $\epsilon_{ab} \epsilon^{bc} \xi_c = -\xi_a$ .

**Definition 1.4.13.** Let  $\mathfrak{v}$  be the real vector space of Hermitian  $2 \times 2$  matrices  $\sigma_{ab}$ . Pauli matrices  $\sigma^{\alpha}{}_{ab}$ ,  $\alpha = 0, 1, 2, 3$ , are a basis in  $\mathfrak{v}$  satisfying  $\sigma^{\alpha}{}_{ab}\sigma^{\beta c b} + \sigma^{\beta}{}_{ab}\sigma^{\alpha c b} = 2g^{\alpha\beta}\delta_{a}{}^{c}$  where

$$\sigma^{\alpha ab} := \epsilon^{ac} \sigma^{\alpha}_{\ cd} \epsilon^{bd}. \tag{1.55}$$

**Remark 1.4.14.** At every point of the manifold M Pauli matrices are defined uniquely up to a Lorentz transformation.

Definition 1.4.15. The second order Pauli matrices are

$$\sigma_{\alpha\beta ac} := \frac{1}{2} \left( \sigma_{\alpha a\dot{b}} \epsilon^{\dot{b}\dot{d}} \sigma_{\beta c\dot{d}} - \sigma_{\beta a\dot{b}} \epsilon^{\dot{b}\dot{d}} \sigma_{\alpha c\dot{d}} \right), \tag{1.56}$$

where  $\epsilon$  is the metric spinor (1.53).

These matrices are polarized, i.e.

$$*\sigma = \pm i \sigma, \tag{1.57}$$

depending on the orientation of basic Pauli matrices  $\sigma^{\alpha}_{\ ab}$ ,  $\alpha = 0, 1, 2, 3$ .

Definition 1.4.16. We define the covariant derivatives of spinor fields as

$$\nabla_{\mu}\xi^{a} = \partial_{\mu}\xi^{a} + \Gamma^{a}{}_{\mu b}\xi^{b}, \qquad \nabla_{\mu}\xi_{a} = \partial_{\mu}\xi_{a} - \Gamma^{b}{}_{\mu a}\xi_{b},$$
$$\nabla_{\mu}\eta^{\dot{a}} = \partial_{\mu}\eta^{\dot{a}} + \bar{\Gamma}^{\dot{a}}{}_{\mu \dot{b}}\eta^{\dot{b}}, \qquad \nabla_{\mu}\eta_{\dot{a}} = \partial_{\mu}\eta_{\dot{a}} - \bar{\Gamma}^{\dot{b}}{}_{\mu \dot{a}}\eta_{\dot{b}}$$

where  $\bar{\Gamma}^{\dot{a}}_{\ \mu\dot{b}} = \overline{\Gamma^{a}_{\ \mu b}}$ .

The explicit formula for the spinor connection coefficients  $\Gamma^a{}_{\mu b}$  can be derived from the following two conditions:

$$\nabla_{\mu}\epsilon^{ab} = 0, \quad \nabla_{\mu}j^{\alpha} = \sigma^{\alpha}{}_{ab}\nabla_{\mu}\zeta^{ab}, \tag{1.58}$$

where  $\zeta$  is an arbitrary rank 2 mixed spinor field and  $j^{\alpha} := \sigma^{\alpha}{}_{ab} \zeta^{ab}$  is the corresponding vector field. Conditions (1.58) give a system of linear algebraic equations for  $\operatorname{Re} \Gamma^{a}{}_{\mu b}$ ,  $\operatorname{Im} \Gamma^{a}{}_{\mu b}$  the unique solution of which is

$$\Gamma^{a}{}_{\mu b} = \frac{1}{4} \sigma_{\alpha}{}^{a\dot{c}} \left( \partial_{\mu} \sigma^{\alpha}{}_{b\dot{c}} + \Gamma^{\alpha}{}_{\mu\beta} \sigma^{\beta}{}_{b\dot{c}} \right).$$
(1.59)

## 1.5 Introducing the massless Dirac operator

The massless Dirac operator describes a single massless neutrino living in a compact universe and physically its eigenvalues are interpreted to be the energy levels of that massless particle. Using the massless Dirac operator, see Section 3.2, the massless Dirac equation can be obtained by varying the action (3.12) with respect to the spinor  $\xi$ . As we show in Section 2.4, the spinor field which determines the complexified curvature of the axial torsion waves introduced in Section 2.2.2 is an exact solution of the massless Dirac equation, see Lemma 2.4.2. We have the same situation in the case of purely tensor torsion waves introduced in Section 2.2.1. Hence, we have a connection between the generalised pp-waves with purely axial torsion and a massless neutrino field. **Remark 1.5.1.** In this thesis, when we consider the solutions of the massless Dirac equation we work in the standard gravitational setting of four dimensions as opposed to the three dimensional setting when we perform spectral analysis of the massless Dirac operator.

Let M be a 3-dimensional connected compact oriented manifold equipped with a Riemannian metric g. The massless Dirac operator, see Definition 3.2.1, is the matrix operator

$$W = -i\sigma^{\alpha} \left( \frac{\partial}{\partial x^{\alpha}} + \frac{1}{4}\sigma_{\beta} \left( \frac{\partial\sigma^{\beta}}{\partial x^{\alpha}} + \left\{ \begin{array}{c} \beta \\ \alpha\gamma \end{array} \right\} \sigma^{\gamma} \right) \right).$$

To our knowledge, the eigenvalues of the massless Dirac operator can be explicitly calculated when we consider the unit torus  $\mathbb{T}^3$  equipped with Euclidean metric and the unit sphere  $\mathbb{S}^3$  equipped with the metric induced by the natural embedding of  $\mathbb{S}^3$  in the Euclidean space  $\mathbb{R}^4$ . It turns out that in these cases the spectrum of the massless Dirac operator is symmetric. However, according to [3, 4, 5, 6], for a general oriented Riemannian 3-manifold (M, g) there is no physical reason for the spectrum of the massless Dirac operator to be symmetric. The spectral symmetry would mean that in these two examples, the massless neutrino and the massless antineutrino have the same properties. Hence, we are interested in a more detailed spectral analysis of the massless Dirac operator and creating spectral asymmetry.

Primarily, we consider the unit torus  $\mathbb{T}^3$  equipped with Euclidean metric. In that case, the spectrum of the massless Dirac operator is as follows: zero is an eigenvalue of multiplicity two and for each  $m \in \mathbb{Z}^3 \setminus \{0\}$  the eigenvalues are  $\pm ||m||$ . Our goal is to break this spectral symmetry using the perturbations of Euclidean metric, see Section 3.4 and to derive the asymptotic formulae of the eigenvalues  $\lambda = \pm 1$  in powers of the small perturbation parameter  $\epsilon$ . For the eigenvalue  $\lambda = 0$  of the massless Dirac operator the asymptotic formula was obtained in [24]. In the same paper the authors determined the conditions under which it is possible to obtain spectral asymmetry. Also, the authors give two explicit examples of perturbations of the Euclidean metric, for which the eigenvalues of the massless Dirac operator on half-densities (3.20) have been evaluated explicitly. One is an example of quadratic dependence and the second is an example of quartic dependence on parameter  $\epsilon$ .

Similarly to the approach in [24], in this thesis we derive the asymptotic formulae for the eigenvalues  $\pm 1$  in the axisymmetric case, see Remark 3.5.2. We show that it is not possible to break the spectral asymmetry on the linear term, see Remark 3.5.9 and we also determine the perturbations of the Euclidean metric for which it is possible to obtain spectral asymmetry, see Remark 3.5.11.

Our aim in the near future is to obtain spectral asymmetry in the case of the unit sphere  $\mathbb{S}^3$  equipped with the metric induced by the natural embedding of  $\mathbb{S}^3$  in the Euclidean space  $\mathbb{R}^4$ .

## **1.6** Main results of the thesis

In this section we briefly explain the main results of the thesis. In order to construct new non-Riemannian solutions for the quadratic metric-affine gravity, we consider the generalisations of classical pp-waves to metric compatible spacetimes whose connection is not Levi-Civita. The whole of Chapter 2 is devoted to generalising pp-waves by adding torsion. Our new generalisation of pp-waves is introduced in Section 2.2.2, where we present the spacetime whose torsion is purely axial and where we list its main properties. We deal with a particular choice of local coordinates where the pp-metric can be written as

$$ds^{2} = 2 dx^{0} dx^{3} - (dx^{1})^{2} - (dx^{2})^{2} + f(x^{1}, x^{2}, x^{3}) (dx^{3})^{2}$$

in some local coordinates  $(x^0, x^1, x^2, x^3)$ . Then we define generalised ppwaves with axial torsion as metric compatible spacetime with the pp-metric and torsion

$$T := *A,$$

where A is a real vector field defined by  $A = k(\varphi) l$ , where  $k : \mathbb{R} \to \mathbb{R}$  is an arbitrary real function and l is a parallel null light-like vector  $l^{\mu} = (1, 0, 0, 0)$ . Torsion defined in such a way is purely axial, see Lemma 2.2.9, and in our local coordinates it can be written as

$$T_{\kappa\mu\nu} = k(x^3)l^{\alpha}\varepsilon_{\alpha\kappa\mu\nu}.$$

The explicit formula for the curvature in our local coordinates is

$$R_{\alpha\beta\gamma\delta} = -\frac{1}{2}(l\wedge\partial)_{\alpha\beta}(l\wedge\partial)_{\gamma\delta}f + \sum_{i,j=1}^{2}r_{ij}(l\wedge m_{i})_{\alpha\beta}(l\wedge m_{j})_{\gamma\delta},$$

where  $r_{11} = r_{22} = \frac{1}{4}(k(x^3))^2$ ,  $r_{12} = -r_{21} = \pm \frac{1}{2}k'(x^3)$ ,  $m^{\mu} = (0, 1, \pm i, 0)$ ,  $m_1 = \operatorname{Re}(m)$  and  $m_2 = \operatorname{Im}(m)$ .

One of the main results of this thesis is the following

**Theorem 1.6.1.** Generalised pp-waves with purely axial torsion of parallel Ricci curvature are solutions of (1.2), (1.3) in the case (1.7).

In order to prove that generalised pp-waves solve the system of equations (1.2), (1.3) in the case (1.7), we first explicitly write down these equations in the form (2.47) and (2.48), which is already well known, see e.g. [53]. Then using (2.8), special local coordinates (2.1), (2.5), as well as the formulae for curvature (2.40), (2.41), torsion (2.35) and torsion generated curvature (2.43) we prove that equations (2.47) and (2.48) are satisfied.

Another main result is the following

# **Theorem 1.6.2.** Generalised pp-waves with purely axial torsion of parallel Ricci curvature are solutions of (1.2), (1.3) in the case (1.4).

In order to prove Theorem 1.6.2, we write down the field equations (1.2), (1.3) explicitly under the certain assumptions, see Section 2.3.1. Then we show that the field equations are satisfied by substituting the explicit formulae for the irreducible pieces of torsion and curvature of generalised pp-waves, see Section 2.3.4.

The generalised pp-waves of parallel Ricci curvature with purely axial torsion have their particular physical interpretation, as we show in Section 2.4. The spinor field  $\xi$ , which completely determines the complexified curvature of generalised pp-waves, satisfies the massless Dirac equation. The axial torsion is the irreducible piece of torsion which is usually used when one models the massless neutrino, see [20], or the electron, see [16], by means of Cosserat elasticity. In Section 2.4 we compare the generalised pp-waves with purely axial torsion, as new vacuum solutions for quadratic metric-affine gravity, to the solutions of the Einstein-Weyl theory. We conclude that generalised pp-waves with purely axial torsion of parallel Ricci curvature are very similar to pp-type solutions of the Einstein-Weyl model and we propose that generalised pp-waves with purely axial torsion of parallel Ricci curvature represent a metric-affine model for the massless neutrino.

Further, in Chapter 3, we are interested in a more mathematical approach to the analysis of the massless Dirac operator on a 3-dimensional manifold. As we stated in Section 1.5, physically interpreted, the massless Dirac operator describes a single massless neutrino. We consider the unit torus  $\mathbb{T}^3$ equipped with the Euclidean metric where the spectrum is calculated explicitly. It turns out that in this case the spectrum is symmetric about zero. Our concern about the spectral symmetry arises from the fact that for a general oriented Riemannian 3-manifold (M, g) there is no physical reason for the spectrum of the massless Dirac operator to be symmetric. Hence, our aim is to create spectral asymmetry. For creating spectral asymmetry we can choose one of the two different approaches: we can consider 3-manifolds with flat metric but nontrivial topology as in [81] or the trivial topology with a perturbed metric as in [24]. We have decided for the latter. Our third main result is the following

**Theorem 1.6.3.** Under certain perturbations of the Euclidean metric, we can obtain spectral asymmetry of the massless Dirac operator (in the axisymmetric case) on the unit 3-torus.

In order to prove Theorem 1.6.3, we consider the perturbed Euclidean metric

$$g_{\alpha\beta}(x;\epsilon) = \delta_{\alpha\beta} + \epsilon h_{\alpha\beta}(x) + \frac{\epsilon^2}{4}k_{\alpha\beta}(x) + O(\epsilon^3),$$

where  $\epsilon$  is a positive small parameter. Using the perturbation theory developed for the massless Dirac operator, see Section 3.4, we explicitly derive the asymptotic formulae for the eigenvalues  $\lambda = \pm 1$  of the massless Dirac operator in the axisymmetric case (3.57). In terms of perturbation matrices  $h_{\alpha\beta}$  and  $k_{\alpha\beta}$  the asymptotic formulae are given in Theorem 3.5.7. Analysing these asymptotic formulae, we see that under certain perturbations of the Euclidean metric, it is possible to obtain spectral asymmetry of the massless Dirac operator in the axisymmetric case, see Remark 3.5.9 and Remark 3.5.11.

## **1.7** Structure of the thesis

This thesis has the following structure:

- Chapter 2 deals with new solutions for quadratic metric-affine gravity and is divided into several sections. In Section 2.1 we present the classical pp-waves and we list the main properties of these spacetimes. In Section 2.2 we are dealing with generalisations of classical pp-waves, extending classical pp-waves to metric compatible spacetimes with torsion. We introduce generalised pp-waves with purely tensor torsion and generalised pp-waves with purely axial torsion and list the main properties of these spacetimes. In Section 2.3 we use generalised pp-waves with purely axial torsion described in Section 2.2 to present a class of new solutions for quadratic metric-affine gravity. In Section 2.4 we give the mathematical and physical significance of our new solutions. We compare our new solutions for quadratic metric-affine gravity to the pptype solutions of Einstein-Weyl theory which describes the interaction of gravitational and massless neutrino fields.
- Chapter 3 deals with the spectral analysis of the massless Dirac operator on a 3-manifold. In Section 3.1 we list some general properties of an elliptic self-adjoint first-order differential operator. In Section 3.2

we present the massless Dirac operator and we list its main properties. The Section 3.3 deals with the spectrum of the massless Dirac operator with special emphasis on studying the spectral function and the counting function of the massless Dirac operator. In Section 3.4 we develop the perturbation theory for the massless Dirac operator used through this thesis. Section 3.5 deals with spectral asymmetry of the massless Dirac operator which is the main objective of Chapter 3. In this section we present our new results for obtaining spectral asymmetry of the massless Dirac operator.

• Appendices provide the auxiliary mathematical facts and the calculations used to obtain the results in this thesis. Appendix A provides the detailed derivation of the Einstein and Yang-Mills equations. Appendix B gives the derivation of the Bianchi identity for the curvature which is used to obtain the explicit representations of the field equations in Section 2.3. In Appendix C we provide the explicit variations of certain quadratic forms on curvature with respect to the metric which are used to get the explicit representations of the field equations. Finally, Appendix D contains the detailed calculations of the coefficients in the asymptotic expansion of the eigenvalues ±1 of the massless Dirac operator.

# Chapter 2

# New Solutions for Quadratic Metric-Affine Gravity

### 2.1 Classical pp-waves

Pp-waves are a very important family of exact solutions of Einstein's field equations and are very well known spacetimes in general relativity. They are introduced long time ago by Brinkmann [14] in 1923, and after rediscovered by several authors. Pp-wave metrics are interpreted as the metrics representing the gravitational waves by Peres [80]. As stated Baykal [12], the pp-wave metrics can be regarded as a far-field description of an isolated astrophysical source radiating gravitational waves. Griffiths [47] and Kramer et al. [55] gave an explanation that the abbreviation 'pp' stands for 'plane-fronted gravitational waves with parallel rays'. It was recently discovered by Vassiliev [101] that pp-waves of parallel Ricci curvature are solutions of the system (1.2), (1.3). For more information on pp-waves and pp-wave type solutions of metric-affine gravity, see e.g. [1, 8, 13, 14, 39, 47, 55, 67, 68, 73, 77, 80, 100, 101]. In this chapter, classical pp-waves will be used for the construction of the new solutions for quadratic metric-affine gravity. We mostly follow the exposition of [74, 75].

**Definition 2.1.1.** A *pp-wave* is a Riemannian spacetime whose metric can be written locally in the form

$$ds^{2} = 2 dx^{0} dx^{3} - (dx^{1})^{2} - (dx^{2})^{2} + f(x^{1}, x^{2}, x^{3}) (dx^{3})^{2}$$
(2.1)

in some local coordinates  $(x^0, x^1, x^2, x^3)$ .

The advantage of Definition 2.1.1 is that it gives an explicit formula for the metric of a pp-wave but its disadvantage is that it relies to the specific coordinate system. We also have the coordinate-free definition of the pp-wave spacetime.

**Definition 2.1.2.** A *pp-wave* is a Riemannian spacetime which admits a nonvanishing parallel spinor field.

**Remark 2.1.3.** The term *parallel* means that covariant derivative of the spinor field is zero, see Definition 1.4.16.

It is known, see [2, 15], that Definition 2.1.1 and Definition 2.1.2 are equivalent. The nonvanishing parallel spinor field appearing in Definition 2.1.2 of pp-waves will be denoted throughout this chapter by

$$\chi = \chi^a, \tag{2.2}$$

and we assume this spinor field to be *fixed*. Let define the vector l as

$$l^{\alpha} := \sigma^{\alpha}{}_{ab} \,\chi^a \bar{\chi}^b \tag{2.3}$$

where  $\sigma^{\alpha}$  are Pauli matrices, see Definition 1.4.13 and Section 1.4.3. Clearly, l is a nonvanishing parallel real null vector field. Using the vector l, we define the real scalar function

$$\varphi: M \to \mathbb{R}, \qquad \varphi(x) := \int l \cdot dx.$$
 (2.4)

This function is called the *phase*.

**Definition 2.1.4.** The complex vector field v is transversal if  $l_{\alpha}v^{\alpha} = 0$ .

**Definition 2.1.5.** The complex vector field v is plane wave if  $v^{\alpha} \nabla_{\alpha} v^{\beta} = 0$  for arbitrary transversal vector field v.

**Remark 2.1.6.** Clearly, the vector l is a transversal and a plane wave.

The choice of local coordinates in which the pp-metric has the form (2.1) is not unique. Our choice of those coordinates is such in which

$$\chi^a = (1,0), \qquad l^\mu = (1,0,0,0), \qquad m^\mu = (0,1,\pm i,0).$$
 (2.5)

With the choice (2.5), the phase function (2.4) explicitly reads  $\varphi(x) = x^3 +$ const. Using the second order Pauli matrices (1.56) and the spinor field (2.2) we can define the complex 2-form

$$F_{\alpha\beta} := \sigma_{\alpha\beta ab} \,\chi^a \chi^b. \tag{2.6}$$

J

Also, it can be written as the wedge product

$$F = l \wedge m, \tag{2.7}$$

where m is a complex vector field satisfying

$$m_{\alpha}m^{\alpha} = l_{\alpha}m^{\alpha} = l_{\alpha}\overline{m}^{\alpha} = 0, \ m_{\alpha}\overline{m}^{\alpha} = -2.$$
(2.8)

If we denote by  $m_1 = \operatorname{Re}(m)$  and  $m_2 = \operatorname{Im}(m)$ , then from (2.8) we also have the following relations

$$l_{\alpha}l^{\alpha} = m_{1\alpha}m_{2}^{\ \alpha} = l_{\alpha}m_{1}^{\ \alpha} = l_{\alpha}m_{2}^{\ \alpha} = 0, \ m_{1\alpha}m_{1}^{\ \alpha} = m_{2\alpha}m_{2}^{\ \alpha} = -1.$$
(2.9)

**Remark 2.1.7.** Clearly, F is a nonvanishing parallel complex 2-form. Applying the Hodge star operator (1.21) in (2.7) we get that  $*F = \pm iF$ . Also det F = 0 for the choice (2.5).

For the metric (2.1), we get the explicit formula for the curvature of the classical pp-wave

$$R_{\alpha\beta\gamma\delta} = -\frac{1}{2} (l \wedge \partial)_{\alpha\beta} (l \wedge \partial)_{\gamma\delta} f, \qquad (2.10)$$

where  $(l \wedge \partial)_{\alpha\beta} := l_{\alpha}\partial_{\beta} - \partial_{\alpha}l_{\beta}$ . The curvature tensor R is linear in f. Formula (2.10) can be written in invariant form

$$R = -\frac{1}{2}(l \wedge \nabla) \otimes (l \wedge \nabla)f, \qquad (2.11)$$

where  $l \wedge \nabla := l \otimes \nabla - \nabla \otimes l$ . The curvature of a pp-wave has only two irreducible pieces, see Section 1.4.1, namely (symmetric) trace-free Ricci and Weyl. Ricci curvature (1.18) is proportional to  $l \otimes l$  whereas Weyl curvature is a linear combination of Re  $((l \wedge m) \otimes (l \wedge m))$  and Im  $((l \wedge m) \otimes (l \wedge m))$ . In our special local coordinates (2.1), (2.5), we can express Ricci and Weyl curvature as

$$Ric_{\mu\nu} = \frac{1}{2}(f_{11} + f_{22}) l_{\mu}l_{\nu}, \qquad (2.12)$$

$$\mathcal{W}_{\kappa\lambda\mu\nu} = \sum_{j,k=1}^{2} w_{jk}(l \wedge m_j) \otimes (l \wedge m_k), \qquad (2.13)$$

where  $f_{\alpha\beta} := \partial_{\alpha}\partial_{\beta}f$  and  $w_{jk}$  are real scalars given by

$$w_{11} = \frac{1}{4}(-f_{11} + f_{22}), \ w_{12} = \pm \frac{1}{2}f_{12}, \ w_{22} = -w_{11}, \ w_{21} = w_{12}.$$

The Cotton tensor, see [40], of classical pp-waves is given by

$$C_{\lambda\mu\nu} = \nabla_{\lambda} Ric_{\mu\nu} - \nabla_{\mu} Ric_{\lambda\nu}. \tag{2.14}$$

In the theory of conformal spaces the main geometrical objects to be analised are the Weyl and the Cotton tensors, see [40]. It is well known that for conformally flat spaces the Weyl tensor has to vanish and consequently the Cotton tensor has to vanish too. The Cotton tensor is only conformally invariant in three dimensions.

Note that if the Ricci curvature is parallel the Cotton tensor (2.14) vanishes.

#### 2.1.1 Pauli matrices for pp-waves

We choose the Pauli matrices for the pp-metric (2.1.1) as

$$\sigma^{0}{}_{ab} = \begin{pmatrix} 1 & 0 \\ 0 & -f \end{pmatrix}, \qquad \sigma^{1}{}_{ab} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix},$$
$$\sigma^{2}{}_{ab} = \begin{pmatrix} 0 & \mp i \\ \pm i & 0 \end{pmatrix}, \qquad \sigma^{3}{}_{ab} = \begin{pmatrix} 0 & 0 \\ 0 & 2 \end{pmatrix}. \qquad (2.15)$$

Our two choices of Pauli matrices differ by orientation. When dealing with a classical pp-wave the choice of orientation of Pauli matrices does not really matter, but for a generalised pp-wave it is convenient to choose orientation of Pauli matrices in agreement with the sign in (2.19) as this simplifies the resulting formulae.

**Remark 2.1.8.** In the case f = 0, formulae (2.15) do not turn into the standard Minkowski space Pauli matrices, since we write the metric in the form (2.1). This is a matter of convenience in calculations.

The second order Pauli matrices  $\sigma^{\alpha\beta}_{\ ab}$  (1.56) for the pp-metric (2.1) are also antisymmetric over the tensor indices, so we only give the independent non-zero components. The explicit formulae for the second order Pauli matrices for the pp-metric (2.1) are

$$\sigma^{01}{}_{ab} = \begin{pmatrix} 1 & 0 \\ 0 & f \end{pmatrix}, \quad \sigma^{02}{}_{ab} = \begin{pmatrix} \mp i & 0 \\ 0 & \pm if \end{pmatrix}, \quad \sigma^{03}{}_{ab} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix},$$
$$\sigma^{12}{}_{ab} = \begin{pmatrix} 0 & \mp i \\ \mp i & 0 \end{pmatrix}, \quad \sigma^{13}{}_{ab} = \begin{pmatrix} 0 & 0 \\ 0 & 2 \end{pmatrix}, \quad \sigma^{23}{}_{ab} = \begin{pmatrix} 0 & 0 \\ 0 & \pm 2i \end{pmatrix}. \quad (2.16)$$

## 2.2 Generalising pp-waves

As we previously stated, the classical pp-waves of parallel Ricci curvature are solutions of the system (1.2), (1.3). Since the classical pp-waves are Riemannian spacetimes with zero torsion, we use these spacetimes in order to construct new solutions of the quadratic metric-affine gravity with torsion. In this section we present the generalisation of the classical pp-wave to the spacetimes whose connection is not necessarily Levi-Civita, but in a very particular way: we will still use the pp-metric (2.1) and introduce explicitly given torsion. We present two types of such spacetimes: generalised pp-waves with purely tensor torsion and generalised pp-waves with purely axial torsion. The former were introduced and analised in [72, 74, 77], while the latter were first introduced and analised to a certain degree in [75].

#### 2.2.1 Generalised pp-waves with purely tensor torsion

This generalisation of the classical pp-waves is done by Pasic and Vassiliev [77]. In the same paper it was shown that the generalised pp-waves with purely tensor torsion of parallel Ricci curvature are solutions of the system (1.2), (1.3) in the general case with 16  $R^2$  quadratic form (1.6). The physical description of that new solutions is given by Pasic and Barakovic [74]. Here we present a review of these results.

**Definition 2.2.1.** A generalised pp-wave with purely tensor torsion is a metric compatible spacetime with pp-metric (2.1) and torsion

$$T := \frac{1}{2} \operatorname{Re}(A \otimes dA), \qquad (2.17)$$

where A is a vector field of the form

$$A = h(\varphi) m + k(\varphi) l, \qquad (2.18)$$

which is a plane wave solution of the polarized Maxwell equation

$$*dA = \pm i \, dA. \tag{2.19}$$

The vector fields l and m appearing in (2.18) are defined in Section 2.1,  $h, k : \mathbb{R} \to \mathbb{C}$  are arbitrary functions, and  $\varphi$  is the phase (2.4).

**Remark 2.2.2.** We denote by  $\{\nabla\}$  the covariant derivative with respect to the Levi-Civita connection which should be distinguished from the full covariant derivative  $\nabla$  incorporating torsion.

These spacetimes have an explicit formulae for the curvature and the torsion. This remarkable property is not a trivial fact. The curvature of a generalised pp-wave is

$$R = -\frac{1}{2}(l \wedge \{\nabla\}) \otimes (l \wedge \{\nabla\})f + \frac{1}{4}\operatorname{Re}\left((h^2)''(l \wedge m) \otimes (l \wedge m)\right). \quad (2.20)$$

**Remark 2.2.3.** The curvatures generated by the Levi-Civita connection and torsion simply add up (compare formulae (2.11) and (2.20)).

The torsion of a generalised pp-wave is

$$T = \operatorname{Re}\left(\left(a \ l+b \ m\right) \otimes \left(l \wedge m\right)\right), \tag{2.21}$$

where

$$a := \frac{1}{2}h'(\varphi) \ k(\varphi), \quad b := \frac{1}{2}h'(\varphi) \ h(\varphi). \tag{2.22}$$

Torsion can be written down even more explicitly in the following form

$$T = \sum_{j,k=1}^{2} t_{jk} m_j \otimes (l \wedge m_k) + \sum_{j=1}^{2} t_j l \otimes (l \wedge m_j), \qquad (2.23)$$

where

$$t_{11} = -t_{22} = \frac{1}{2} \operatorname{Re}(b), \ t_{12} = t_{21} = -\frac{1}{2} \operatorname{Im}(b), \ t_1 = \frac{1}{2} \operatorname{Re}(a), \ t_2 = -\frac{1}{2} \operatorname{Im}(a),$$

where a and b are the functions (2.22).

**Remark 2.2.4.** The torsion (2.17) of a generalised pp-wave is purely tensor. For the proof see Lemma 2 of [74].

The spinor filed (2.2) introduced in Section 2.1 satisfies  $\{\nabla\}\chi = 0$ . As was shown in [74], for the generalised pp-waves with purely tensor torsion we also have that  $\nabla\chi = 0$ . It means that the generalised pp-wave with purely tensor torsion and the underlying classical pp-wave admit the same nonvanishing parallel spinor field. Also, the generalised pp-wave and the underlying classical pp-wave admit the same nonvanishing parallel real null vector field l and the same nonvanishing parallel complex 2-form (2.6), (2.7).

We list below the main properties of generalised pp-waves with purely tensor torsion:

(a) The second term in the explicit formula for the curvature (2.20) of the generalised pp-wave with purely tensor torsion is purely Weyl.
(b) The generalised pp-waves with purely tensor torsion have the same nonzero irreducible pieces of curvature as classical pp-waves, namely symmetric trace-free Ricci and Weyl. Using special local coordinates (2.1), (2.5), these can be expressed explicitly as

$$Ric_{\mu\nu} = \frac{1}{2}(f_{11} + f_{22}) \, l_{\mu}l_{\nu}, \qquad (2.24)$$

$$\mathcal{W}_{\kappa\lambda\mu\nu} = \sum_{j,k=1}^{2} w_{jk} (l \wedge m_j) \otimes (l \wedge m_k), \qquad (2.25)$$

where  $w_{jk}$  are real scalars given by

$$w_{11} = \frac{1}{4} [-f_{11} + f_{22} + \operatorname{Re}((h^2)'')], \qquad w_{22} = -w_{11}$$
$$w_{12} = \pm \frac{1}{2} f_{12} - \frac{1}{4} \operatorname{Im}((h^2)''), \qquad w_{21} = w_{12}.$$

The formulae (2.24) and (2.25) are very similar to the formulae (2.12)and (2.13). The only difference is in the coefficients  $w_{ij}$ .

- (c) The Ricci curvature (2.24) of a generalised pp-wave with purely tensor torsion is completely determined by the pp-metric (2.1).
- (d) The curvature of a generalised pp-wave with purely tensor torsion (2.20)has all the usual symmetries of curvature in the Riemannian case, that is

$$R_{\kappa\lambda\mu\nu} = R_{\mu\nu\kappa\lambda}, \qquad (2.26)$$

$$\varepsilon^{\kappa\lambda\mu\nu}R_{\kappa\lambda\mu\nu} = 0, \qquad (2.27)$$

$$R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu}, \qquad (2.28)$$

$$R_{\kappa\lambda\mu\nu} = -R_{\kappa\lambda\nu\mu}.\tag{2.29}$$

Of course, (2.29) is true for any curvature whereas (2.28) is a consequence of metric compatibility. Also, (2.28) follows from (2.26) and (2.29).

- (e) The second term in the vector field A (2.18) does not affect curvature (2.20) and it only affects torsion (2.17).
- (f) The Ricci curvature of a generalised pp-wave with purely tensor torsion (2.24) is parallel if and only if

$$f_{11} + f_{22} = C, (2.30)$$

where C is an arbitrary real constant.

(g) The Ricci curvature of a generalised pp-wave with purely tensor torsion (2.24) is zero if and only if

$$f_{11} + f_{22} = 0 \tag{2.31}$$

and the Weyl curvature is zero if and only if

$$f_{11} - f_{22} = \operatorname{Re}\left((h^2)''\right), \qquad f_{12} = \pm \frac{1}{2} \operatorname{Im}\left((h^2)''\right).$$
 (2.32)

Here we use special local coordinates (2.1), (2.5).

(h) The curvature of a generalised pp-wave with purely tensor torsion (2.20) is zero if and only if we have both (2.31) and (2.32).

#### 2.2.2 Generalised pp-waves with purely axial torsion

In this section we present the generalisation of the classical pp-wave to the spacetimes whose torsion is purely axial. This generalisation of the classical pp-wave spacetimes was introduced by Pasic and Barakovic [75]. In the same paper it was shown that these spacetimes are solutions of the system (1.2), (1.3) for the Yang-Mills action (1.7). In this chapter we will show that generalised pp-waves with purely axial torsion are solutions of the system (1.2), (1.3) for the case quadratic form (1.4) with  $11 R^2$  terms.

**Definition 2.2.5.** A generalised pp-wave with purely axial torsion is a metric compatible spacetime with pp-metric and torsion

$$T := *A \tag{2.33}$$

where A is a real vector field defined by  $A = k(\varphi) l$ , where  $k : \mathbb{R} \to \mathbb{R}$  is an arbitrary real function of the phase (2.4) for the vector l (2.5).

**Remark 2.2.6.** In our special local coordinates (2.1), (2.5), torsion (2.33) can be written as

$$T_{\kappa\mu\nu} = k(x^3)l^{\alpha}\varepsilon_{\alpha\kappa\mu\nu}, \qquad (2.34)$$

where  $\varepsilon_{\kappa\lambda\mu\nu}$  is totally antisymmetric quantity and  $\varepsilon_{0123} = +1$ .

**Remark 2.2.7.** The real vector field A is a plane wave solution of the polarized Maxwell equation  $*dA = \pm i \, dA$ . This is not surprising as in our special local coordinates the vector field A is the gradient of a scalar function.

**Remark 2.2.8.** It has been suggested that one can interpret the axial component of torsion as the Hodge dual of the electromagnetic vector potential, see [53, 98].

Lemma 2.2.9. The torsion (2.33) of a generalised pp-wave is purely axial.

*Proof.* In our special local coordinates (2.1), (2.5), using formulae (1.27) and (2.34), we have that

$$w_{\nu} = \frac{1}{6}\sqrt{|\det g|}T^{\kappa\lambda\mu}\varepsilon_{\kappa\lambda\mu\nu} = \frac{1}{6}k(x^{3})l_{\alpha}\varepsilon^{\alpha\kappa\lambda\mu}\varepsilon_{\kappa\lambda\mu\nu}.$$

Hence, using formula (1.48), we get that

$$(T^{(3)})_{\alpha\beta\gamma} = (*w)_{\alpha\beta\gamma} = \frac{1}{6}k(x^3)l_{\alpha}\varepsilon^{\alpha\kappa\lambda\mu}\varepsilon_{\kappa\lambda\mu}{}^{\eta}\varepsilon_{\eta\alpha\beta\gamma}$$
$$= -\frac{1}{6}k(x^3)l_{\xi}(-3!)g^{\xi\eta}\varepsilon_{\eta\alpha\beta\gamma} = k(x^3)l^{\xi}\varepsilon_{\xi\alpha\beta\gamma} = T_{\alpha\beta\gamma}.$$

**Lemma 2.2.10.** Using our special local coordinates (2.1), (2.5), we can express torsion as

$$T = \mp \frac{\mathbf{i}}{2} k(x^3) \ l \wedge m \wedge \overline{m}, \tag{2.35}$$

where the  $\mp$  sign is chosen to correspond to the sign in (2.5).

*Proof.* We will prove that using our special local coordinates (2.1), (2.5) we have that

$$l_{\mu}\varepsilon^{\mu\alpha\beta\gamma} = \mp \frac{\mathrm{i}}{2} (l \wedge m \wedge \overline{m})^{\alpha\beta\gamma}.$$
(2.36)

By definition we have that

$$(l \wedge m \wedge \overline{m})^{\alpha\beta\gamma} = l^{\alpha}m^{\beta}\overline{m}^{\gamma} + l^{\beta}m^{\gamma}\overline{m}^{\alpha} + l^{\gamma}m^{\alpha}\overline{m}^{\beta} - l^{\alpha}m^{\gamma}\overline{m}^{\beta} - l^{\beta}m^{\alpha}\overline{m}^{\gamma} - l^{\gamma}m^{\beta}\overline{m}^{\alpha}.$$
(2.37)

Using special local coordinates (2.1), (2.5), we conclude that if any of the  $\alpha, \beta$  or  $\gamma$  is equal to 3, then the wedge product (2.37) is zero and the quantity  $l_{\mu}\varepsilon^{\mu\alpha\beta\gamma} = l_{3}\varepsilon^{3\alpha\beta\gamma}$  is also zero since  $\varepsilon$  is totally antisymmetric. Hence (2.36) holds.

Let then  $\alpha, \beta, \gamma \neq 3$ . By definition, tensors  $(l \wedge m \wedge \overline{m})^{\alpha\beta\gamma}$  and  $l_{\mu}\varepsilon^{\mu\alpha\beta\gamma}$ are totally antisymmetric, hence the only non-zero terms in both quantities appear when  $\alpha, \beta, \gamma$  are a permutation of 0, 1, 2. It is therefore enough to calculate the only independent non-zero term, i.e.

$$(l \wedge m \wedge \overline{m})^{012} = l^0 m^1 \overline{m}^2 - l^0 m^2 \overline{m}^1 = \pm 2i.$$

Now we have that  $\pm \frac{i}{2}(l \wedge m \wedge \overline{m})^{012} = 1$  and  $l_{\mu}\varepsilon^{\mu 012} = l_3\varepsilon_{3012} = 1$ . The other cases are shown analogously. Hence we have the formula (2.36). Combining formulae (2.34) and (2.36), we get formula (2.35).

Clearly, from the equation (2.35) we have that we can express contortion as

$$K = \mp \frac{\mathbf{i}}{4} k(x^3) \ l \wedge m \wedge \overline{m}. \tag{2.38}$$

Also, since we have that

$$l \wedge m \wedge \overline{m} = l \wedge (m_1 + im_2) \wedge (m_1 - im_2) = -2i (l \wedge m_1 \wedge m_2),$$

where  $m_1 = \operatorname{Re}(m)$  and  $m_2 = \operatorname{Im}(m)$ , we get an equivalent formula for the torsion

$$T = \mp k(x^3) \ l \wedge m_1 \wedge m_2 \tag{2.39}$$

in our special local coordinates (2.1), (2.5).

**Remark 2.2.11.** Our torsion completely corresponds to Singh's axial torsion from [87, 88]. Put  $m = -\frac{1}{2}k(x^3)$  in formula (16) of [87] or put  $n = 0, m = -\frac{1}{2}k(x^3)$  in formula (20) of [88].

**Lemma 2.2.12.** The connection of a generalised pp-wave with purely axial torsion is metric compatible.

*Proof.* Since

$$\Gamma^{\kappa}{}_{\mu\nu} = \left\{ \begin{matrix} \lambda \\ \mu\nu \end{matrix} \right\} + K^{\kappa}{}_{\mu\nu} = \left\{ \begin{matrix} \lambda \\ \mu\nu \end{matrix} \right\} + \frac{1}{2}T^{\kappa}{}_{\mu\nu},$$

we get that

$$\nabla_{\mu}g_{\alpha\beta} = \{\nabla\}_{\mu}g_{\alpha\beta} - K^{\eta}{}_{\mu\alpha}g_{\eta\beta} - K^{\eta}{}_{\mu\beta}g_{\alpha\eta}$$

and  $\{\nabla\}_{\mu}g_{\alpha\beta} = 0$  as classical pp-waves are metric compatible. However, since our torsion is purely axial, we get that

$$\nabla_{\mu}g_{\alpha\beta} = -K_{\beta\mu\alpha} - K_{\alpha\mu\beta} = K_{\alpha\mu\beta} - K_{\alpha\mu\beta} = 0,$$

i.e. we have metric-compatibility.

**Remark 2.2.13.** Note that the spinor field (2.2) is no longer parallel in generalised pp-waves with purely axial torsion, as was the case with generalised pp-waves with purely tensor torsion. Moreover, using local coordinates (2.1),

(2.5), the covariant derivative of this spinor field is  $\nabla \chi = \begin{pmatrix} 2i \\ 0 \end{pmatrix} \neq 0.$ 

The remarkable property of these spacetimes is that we have the explicit formula for the curvature which is

$$R = -\frac{1}{2}(l \wedge \{\nabla\}) \otimes (l \wedge \{\nabla\})f + \frac{1}{4}(k(x^3))^2 \operatorname{Re}\left((l \wedge m) \otimes (l \wedge \overline{m})\right)$$
  
$$\mp \frac{1}{2}k'(x^3) \operatorname{Im}\left((l \wedge m) \otimes (l \wedge \overline{m})\right).$$
(2.40)

This can be equivalently written down as

$$R_{\alpha\beta\gamma\delta} = -\frac{1}{2} (l \wedge \partial)_{\alpha\beta} (l \wedge \partial)_{\gamma\delta} f + \sum_{i,j=1}^{2} r_{ij} (l \wedge m_i)_{\alpha\beta} (l \wedge m_j)_{\gamma\delta}, \qquad (2.41)$$

where  $r_{11} = r_{22} = \frac{1}{4}(k(x^3))^2$ ,  $r_{12} = -r_{21} = \pm \frac{1}{2}k'(x^3)$ , and  $m_1 = \operatorname{Re}(m)$ ,  $m_2 = \operatorname{Im}(m)$ . It is a highly nontrivial fact that the torsion generated curvature, i.e.

$$R_T^{\ \kappa}{}_{\lambda\mu\nu} = \partial_{\mu}K^{\kappa}{}_{\nu\lambda} - \partial_{\nu}K^{\kappa}{}_{\mu\lambda} + K^{\kappa}{}_{\mu\eta}K^{\eta}{}_{\nu\lambda} - K^{\kappa}{}_{\nu\eta}K^{\eta}{}_{\mu\lambda}, \qquad (2.42)$$

which is equal to

$$R_T = \frac{1}{4} (k(x^3))^2 \operatorname{Re} \left( (l \wedge m) \otimes (l \wedge \overline{m}) \right)$$
  
$$\mp \frac{1}{2} k'(x^3) \operatorname{Im} \left( (l \wedge m) \otimes (l \wedge \overline{m}) \right)$$
(2.43)

and the Riemannian curvature (2.11) simply add up to produce formula (2.40).

**Remark 2.2.14.** Note that the above property of curvature of the pp-waves with purely axial torsion corresponds to the similar property that pp-waves with purely tensor torsion also poses.

We know that the Riemannian part of curvature has two irreducible pieces of curvature, namely Weyl and (symmetric) trace-free Ricci. It turns out that the torsion also generates Ricci curvature and it reads

$$Ric_{\mu\nu} = \frac{1}{2} \left( f_{11} + f_{22} - (k(x^3))^2 \right) l_{\mu} l_{\nu}$$
(2.44)

and

$$Ric_{*\mu\nu} = -k'(x^3)l_{\mu}l_{\nu}.$$
 (2.45)

Scalar curvature is then clearly zero by the properties of the vector l. The curvature of generalised pp-waves with purely axial torsion (2.40) has only three irreducible pieces, namely  $R^{(1)}$  (1.31),  $R^{(3)}$  (1.33) and  $R^{(5)}$  (1.35). The irreducible piece  $R^{(1)}$  partly comes from the classical pp-waves and partly from torsion generate curvature. The irreducible piece  $R^{(3)}$  entirely arises from the classical pp-wave and the irreducible piece  $R^{(5)}$  from the torsion

generated curvature (2.43). In our special local coordinates (2.1), (2.5), the explicit formula for the irreducible piece  $R^{(1)}$  is given by

$$R^{(1)}{}_{\kappa\lambda\mu\nu} = \frac{1}{4} \left( f_{11} + f_{22} \right) \left( g_{\kappa\mu} l_{\lambda} l_{\nu} - g_{\lambda\mu} l_{\kappa} l_{\nu} + g_{\lambda\nu} l_{\kappa} l_{\mu} - g_{\kappa\nu} l_{\lambda} l_{\mu} \right) + \frac{1}{4} k^2 \operatorname{Re} \left( (l \wedge m)_{\kappa\lambda} (l \wedge \overline{m})_{\mu\nu} \right)$$

and the explicit formula for the Weyl curvature  $R^{(3)}$  is given by the earlier formula (2.13). The explicit formula for the irreducible piece  $R^{(5)}$  is given by

$$R^{(5)}{}_{\kappa\lambda\mu\nu} = \pm \frac{1}{2} k'(x^3) (\varepsilon^{\eta}{}_{\kappa\mu\nu} l_{\lambda} l_{\eta} - \varepsilon^{\eta}{}_{\lambda\mu\nu} l_{\kappa} l_{\eta}).$$
(2.46)

For the detailed derivation of the formula (2.46) see Appendix B.1.

Examination of the formula (2.40) for the curvature of a generalised ppwave with purely axial torsion reveals the following properties:

- (a) The curvatures generated by the Levi-Civita connection (2.11) and torsion (2.43) simply add up to produce the formula (2.40).
- (b) The curvature of a generalised pp-wave has the symmetries

$$\varepsilon^{\kappa\lambda\mu\nu}R_{\kappa\lambda\mu\nu} = 0,$$
  
$$R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu},$$
  
$$R_{\kappa\lambda\mu\nu} = -R_{\kappa\lambda\nu\mu},$$

but we do not have the symmetry  $R_{\kappa\lambda\mu\nu} = R_{\mu\nu\lambda\kappa}$  as was the case with generalised pp-waves with purely tensor torsion.

- (c) For an arbitrary purely axial torsion, the Weyl curvature of the resulting torsion generated curvature (2.43) is zero.
- (d) Ricci is parallel if  $f_{11} + f_{22} = (k(x^3))^2 + C$ , in which case  $Ric = \Lambda \ l \otimes l$ , for some constant  $\Lambda$ .
- (e) The Ricci curvature (2.44) is zero if Poisson's equation  $f_{11} + f_{22} = (k(x^3))^2$  is satisfied.

## 2.3 New solutions for quadratic metric-affine gravity

In this section we prove that the generalised pp-waves with purely axial torsion presented in Section 2.2.2 are the solutions of the system (1.2), (1.3)

for the Yang-Mills action (1.7) and also in the case of the 11  $R^2$  term quadratic form (1.4).

First, we will prove that the generalised pp-waves with purely axial torsion are solutions of the system (1.2), (1.3) for the Yang-Mills action (1.7).

The main result is the following

**Theorem 2.3.1.** Generalised pp-waves with purely axial torsion of parallel  $\{Ric\}$  are solutions of (1.2), (1.3) in the special case (1.7).

**Remark 2.3.2.** Note that by  $\{Ric\}$  we denote the Ricci curvature generated by the Levi-Civita connection only. The condition  $\{\nabla\}\{Ric\} = 0$  implies that  $f_{11} + f_{22} = C$ . Note that the result also holds if Ric is assumed to be parallel.

**Remark 2.3.3.** In the special case (1.7), we call equation (1.3) the Yang-Mills equation for the affine connection, i.e.

$$\partial_{\nu}R^{\mu\nu} + [\Gamma_{\nu}, R^{\mu\nu}] = 0, \qquad (2.47)$$

where  $[\Gamma_{\nu}, R^{\mu\nu}]^{\kappa}{}_{\lambda} = \Gamma^{\kappa}{}_{\nu\eta}R^{\eta}{}_{\lambda}{}^{\mu\nu} - \Gamma^{\eta}{}_{\nu\lambda}R^{\kappa}{}_{\eta}{}^{\mu\nu}$ . We call the equation (1.2) in the special case (1.7) the complementary Yang-Mills equation, i.e.

$$H - \frac{1}{4}(\operatorname{tr} H)g = 0,$$
 (2.48)

where  $H = H_{\nu}^{\rho} := R^{\kappa}_{\lambda\mu\nu} R^{\lambda}{}_{\kappa}{}^{\mu\rho}$ . Equivalently, equation (2.48) can written down as

$$R^{\kappa}{}_{\lambda\nu}{}^{\alpha}R^{\lambda}{}_{\kappa}{}^{\nu\beta} - \frac{1}{4}g^{\alpha\beta}R^{\kappa}{}_{\lambda\mu\nu}R^{\lambda}{}_{\kappa}{}^{\mu\nu} = 0.$$

For the explicit derivations of the equations (2.47) and (2.48) see Appendix A.2.

**Proof** of Theorem 2.3.1. Since we know, see e.g. [74, 77, 100], that classical pp-waves of parallel Ricci curvature are solutions of (1.2), (1.3) in the special case (1.7), it is enough to show the result for the torsion generated part of curvature (2.43). In proving that generalised pp-waves solve the equations (2.47) and (2.48), we will use equations (2.8), special local coordinates (2.1), (2.5), as well as the formulae for curvature (2.40), (2.41), torsion (2.35) and torsion generated curvature (2.43). To show that equation (2.47) is satisfied, we only need to show that

$$\partial_{\nu}R_T^{\mu\nu} + [K_{\nu}, R_T^{\mu\nu}] = 0,$$

since the curvature is the sum of the Riemannian curvature (2.11) and the torsion generated curvature (2.43), the connection the sum of the Christoffel symbol and contortion and since classical pp-waves solve the Yang-Mills equation. Now, since  $F = l \wedge m$ , in special local coordinates (2.1), (2.5), the antisymmetric tensor F (2.6) only has two non-zero independent components, namely

$$F^{01} = 1, \quad F^{02} = \mp i.$$

Since the function  $k(\varphi)$  is the function of  $x^3$  in special local coordinates, using the formula for curvature (2.43) we directly get that  $\partial_{\nu}R_T^{\mu\nu} = 0$ . Using the explicit formula for torsion (2.35), the fact that it is purely axial, which implies that T = 2K, the explicit formula for torsion generated curvature (2.43) and special local coordinates (2.1), (2.5), we get that the only nonzero term of  $\Gamma^{\kappa}{}_{\nu\eta}R^{\eta}{}_{\lambda}{}^{\mu\nu}$  is for  $\kappa = 0, \lambda = 3, \mu = 0$ , i.e.  $-\frac{1}{2}k \cdot k'$ . However, the only nonzero term of  $\Gamma^{\eta}{}_{\nu\lambda}R^{\kappa}{}_{\eta}{}^{\mu\nu}$  is also when  $\kappa = 0, \lambda = 3, \mu = 0$ , i.e.  $-\frac{1}{2}k \cdot k'$ , so these two terms cancel out. Hence, the equation (2.47) is satisfied. Checking that all the terms in equation (2.48) are zero is a straightforward, using the formulae for curvature (2.40), (2.41), special local coordinates (2.1), (2.5) and equation (2.8).

Now, we will prove that the generalised pp-waves with purely axial torsion are solutions of the system (1.2), (1.3) for the quadratic form (1.4). The main result of this chapter is the following

**Theorem 2.3.4.** Generalised pp-waves with purely axial torsion of parallel Ricci curvature are solutions of (1.2), (1.3) in the case (1.4).

**Remark 2.3.5.** The condition of parallel Ricci curvature can be exchanged with the condition  $\{\nabla\}\{Ric\} = 0$ , see Remark 2.3.13.

In order to prove Theorem 2.3.4 we need to first explicitly write down our system of equations (1.2), (1.3).

#### 2.3.1 Explicit representation of the field equations

In this section we explicitly write down the system of equations (1.2), (1.3) for the quadratic form (1.4) under the following assumptions:

- (i) Our spacetime is metric compatible.
- (ii) Torsion is purely axial.
- (iii) Ricci curvature (1.18) is symmetric.
- (iv) Scalar curvature  $\mathcal{R}$  and pseudoscalar curvature  $\mathcal{R}_*$  are zero.

**Remark 2.3.6.** Note that pp-waves with purely axial torsion automatically satisfy the above assumptions (i) - (iv).

**Remark 2.3.7.** Note that the above assumptions (i) - (iv) are only applied *after* the variations have been performed.

**Remark 2.3.8.** In the general case, the curvature has the symmetry  $R_{\kappa\lambda\mu\nu} = -R_{\kappa\lambda\nu\mu}$  and the symmetry  $R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu}$  is the consequence of the metric compatibility. In the derivation of the explicit form of the system of equations (1.2), (1.3), we will not use the property that  $R_{\kappa\lambda\mu\nu} = R_{\mu\nu\kappa\lambda}$ , since the curvature of the generalised pp-waves with axial torsion does not possess this property.

**Remark 2.3.9.** Note that the symmetry of *Ric* implies the symmetry of *Ric*<sub>\*</sub>. Indeed, under the assumption that *Ric* is symmetric consequently the tensors  $A^{(l)}$  from Section 1.4.1 are all equal to zero. Equations (1.42), (1.43) imply that the tensors  $\mathcal{A}^{(i)}_*$  are also equal to zero, which implies that *Ric*<sub>\*</sub> is also symmetric.

The main result of this section is the following.

**Theorem 2.3.10.** Under the above assumptions (i) - (iv) the field equations (1.2), (1.3) in the special case (1.4), can be written down as

$$0 = 2d_1 W^{\kappa\beta\alpha\nu} Ric_{\kappa\nu} + d_2 \epsilon^{\eta\nu\alpha\beta} Ric_{\kappa\nu} Ric_{\ast}^{\ \kappa} \eta - d_3 \epsilon^{\kappa\lambda\xi\alpha} W_{\kappa\lambda\mu}{}^{\beta} Ric_{\ast}{}^{\mu}{}_{\xi}, \qquad (2.49)$$

$$0 = d_1 \{ \nabla_{\lambda} Ric_{\kappa\mu} - \nabla_{\kappa} Ric_{\lambda\mu} + T_{\mu\eta\lambda} Ric_{\kappa}{}^{\eta} + T_{\mu\kappa\eta} Ric_{\lambda}{}^{\eta} \}$$

$$- d_4 \{ (g_{\kappa\mu} \mathcal{W}^{\xi\zeta}{}_{\lambda\eta} - g_{\lambda\mu} \mathcal{W}^{\xi\zeta}{}_{\kappa\eta}) T^{\eta}{}_{\xi\zeta} + (g_{\kappa\mu} \epsilon^{\vartheta\zeta}{}_{\eta\lambda} - g_{\lambda\mu} \epsilon^{\vartheta\zeta}{}_{\eta\kappa}) T^{\eta}{}_{\xi\zeta} Ric_{\ast}{}^{\xi}{}_{\vartheta} \}$$

$$+ c_5 \{ \epsilon^{\eta\xi}{}_{\kappa\mu} \nabla_{\xi} Ric_{\ast\lambda\eta} - \epsilon^{\eta\xi}{}_{\lambda\mu} \nabla_{\xi} Ric_{\ast\kappa\eta} + \frac{1}{2} (\epsilon_{\kappa}{}^{\eta\xi\zeta} Ric_{\ast\lambda\eta} - \epsilon_{\lambda}{}^{\eta\xi\zeta} Ric_{\ast\kappa\eta}) T_{\mu\zeta\xi} \}$$

$$- c_3 \{ 2T^{\eta}{}_{\lambda\xi} \mathcal{W}^{\xi}{}_{\mu\kappa\eta} + 2T^{\eta}{}_{\xi\kappa} \mathcal{W}^{\xi}{}_{\mu\lambda\eta} + T_{\mu\xi\eta} \mathcal{W}_{\kappa\lambda}{}^{\eta\xi} + \epsilon^{\vartheta}{}_{\mu\eta\lambda} T^{\eta}{}_{\xi\kappa} Ric_{\ast}{}^{\xi}{}_{\vartheta}$$

$$- \epsilon^{\vartheta}{}_{\mu\kappa\eta} T^{\eta}{}_{\xi\lambda} Ric_{\ast}{}^{\xi}{}_{\vartheta} - \epsilon^{\vartheta\xi}{}_{\eta\kappa} \nabla_{\xi} Ric_{\ast\mu\vartheta} + \epsilon^{\vartheta\xi}{}_{\eta\lambda} T^{\eta}{}_{\xi\kappa} Ric_{\ast\mu\vartheta}$$

$$+ \epsilon^{\vartheta}{}_{\mu\lambda\kappa} \nabla_{\xi} Ric_{\ast}{}^{\xi}{}_{\vartheta} - \epsilon^{\vartheta\xi}{}_{\lambda\kappa} \nabla_{\xi} Ric_{\ast\mu\vartheta} \}, \qquad (2.50)$$

where  $c_1, c_3, c_5$  are the coefficients of the quadratic form (1.4) and  $d_1 = c_1 + c_3$ ,  $d_2 = c_1 - c_5$ ,  $d_3 = c_3 + c_5$ ,  $d_4 = \frac{1}{2}(c_1 - c_3)$ .

**Remark 2.3.11.** Since the action (1.1) is conformally invariant, the system (2.49), (2.50) is actually the system of 9 + 64 equations, i.e. the equation (2.49) has 9 independent equations, not 10.

Note that the equations (2.49), (2.50) are obtained by varying the action (1.1) for the quadratic form (2.52) independently with respect to the metric and with respect to the connection and without any a priori assumptions on the curvature or torsion. Only *after* we have finished these variations will we use the assumptions (i) - (iv) and explicit formulae for the pieces of curvature in order to obtain the equations (2.49), (2.50).

**Remark 2.3.12.** Assumption (i) implies that the piece of curvature  $R^{(6)}$  (1.36) is zero and assumption (iv) clearly implies that the  $R^{(2)}$  (1.32) and  $R^{(4)}$  (1.34) pieces of curvature are zero. Hence, under the above assumptions, the curvature (1.17) has only *three* nonzero irreducible pieces, namely  $R^{(1)}$ ,  $R^{(3)}$  and  $R^{(5)}$ . It can therefore be represented as

$$R_{\kappa\lambda\mu\nu} = \frac{1}{2} (g_{\kappa\mu}Ric_{\lambda\nu} - g_{\lambda\mu}Ric_{\kappa\nu} + g_{\lambda\nu}Ric_{\kappa\mu} - g_{\kappa\nu}Ric_{\lambda\mu}) + \mathcal{W}_{\kappa\lambda\mu\nu} + \frac{1}{2} (-\epsilon^{\eta}{}_{\lambda\mu\nu}Ric_{*\kappa\eta} + \epsilon^{\eta}{}_{\kappa\mu\nu}Ric_{*\lambda\eta}).$$
(2.51)

Now we will prove Theorem 2.3.10. Since the curvature of the generalised pp-waves with axial torsion (2.40) has only three irreducible pieces  $R^{(1)}$  (1.31),  $R^{(3)}$  (1.33) and  $R^{(5)}$  (1.35), we consider the quadratic form (1.4) written as

$$q(R) = c_1(R^{(1)}, R^{(1)}) + c_3(R^{(3)}, R^{(3)}) + c_5(R^{(5)}, R^{(5)}) + \dots$$
(2.52)

where by ... we denote the terms of the quadratic form that do not contribute  $\delta S$  when we vary using the above assumptions (i)-(iv).

#### 2.3.2 Variation with respect to metric

Using the formula (1.31) for the piece  $R^{(1)}$  of curvature and the formula for the Yang-Mills inner product (1.5), we get that

$$(R^{(1)}, R^{(1)})_{YM} = R^{(1)\kappa}{}_{\lambda\mu\nu}R^{(1)\lambda}{}_{\kappa}{}^{\mu\nu} = -2\overline{Ric}{}_{\lambda\nu}\overline{Ric}{}^{\lambda\nu} + \frac{1}{2}\mathcal{R}^2.$$

Since the variation of the scalar curvature with respect to the metric is zero, we get that

$$(R^{(1)}, R^{(1)})_{YM} = -2\overline{Ric}_{\lambda\nu}\overline{Ric}^{\lambda\nu}$$
$$= -\frac{1}{2}Ric_{\lambda\nu}Ric^{\lambda\nu} - \frac{1}{2}Ric^{(2)}_{\lambda\nu}Ric^{(2)\lambda\nu} + Ric_{\lambda\nu}Ric^{(2)\lambda\nu}.$$

Using the results from Appendix C we get that

$$\frac{\delta}{\delta g} \int (R^{(1)}, R^{(1)})_{YM} = \int (\delta g_{\alpha\beta}) \left( 2Ric^{\alpha}{}_{\nu}Ric^{\beta\nu} - 2R^{\kappa\beta\alpha\nu}Ric_{\kappa\nu} - Ric_{\lambda\nu}Ric^{\lambda\nu}g^{\alpha\beta} \right). \quad (2.53)$$

Substituting the explicit formula for the curvature

$$R^{\kappa\lambda\mu\nu} = R^{(1)\kappa\lambda\mu\nu} + R^{(3)\kappa\lambda\mu\nu} + R^{(5)\kappa\lambda\mu\nu}$$
  
=  $\frac{1}{2} \left( g^{\kappa\mu}Ric^{\lambda\nu} - g^{\lambda\mu}Ric^{\kappa\nu} - g^{\kappa\nu}Ric^{\lambda\mu} + g^{\lambda\nu}Ric^{\kappa\mu} \right)$   
+  $W^{\kappa\lambda\mu\nu} + \frac{1}{2} \left( \epsilon^{\eta\kappa\mu\nu}Ric_{*}{}^{\lambda}{}_{\eta} - \epsilon^{\eta\lambda\mu\nu}Ric_{*}{}^{\kappa}{}_{\eta} \right)$  (2.54)

into (2.53) we get

$$\frac{\delta}{\delta g} \int (R^{(1)}, R^{(1)})_{YM} = \int (\delta g_{\alpha\beta}) \left( -2W^{\kappa\beta\alpha\nu} Ric_{\kappa\nu} + \epsilon^{\eta\beta\alpha\nu} Ric_{*}{}^{\kappa}{}_{\eta}Ric_{\kappa\nu} \right).$$
(2.55)

Since the piece  $R^{(6)}$  of curvature is zero, according to formula (1.35), the piece  $R^{(5)}$  of curvature is

$$R_{\kappa\lambda\mu\nu}^{(5)} = \frac{1}{4} \left( R_{\kappa\lambda\mu\nu} - R_{\lambda\kappa\mu\nu} - R_{\mu\nu\kappa\lambda} + R_{\nu\mu\kappa\lambda} \right).$$

Hence

$$\frac{\delta}{\delta g} \int (R^{(5)}, R^{(5)})_{YM} = 
= \frac{1}{16} \frac{\delta}{\delta g} \int (R^{\kappa}_{\lambda\mu\nu} R^{\lambda}_{\kappa}{}^{\mu\nu} - R^{\kappa}_{\lambda\mu\nu} R^{\lambda\mu\nu} - R^{\kappa}_{\lambda\mu\nu} R^{\mu\nu\lambda}_{\kappa} + R^{\kappa}_{\lambda\mu\nu} R^{\nu\mu\lambda}_{\kappa}) 
- \frac{1}{16} \frac{\delta}{\delta g} \int (R^{\kappa}_{\lambda}{}^{\mu}_{\mu\nu} R^{\lambda}_{\kappa}{}^{\mu\nu} - R^{\kappa}_{\lambda}{}^{\mu}_{\mu\nu} R^{\lambda\mu\nu} - R^{\kappa}_{\lambda}{}^{\mu}_{\mu\nu} R^{\mu\nu\lambda}_{\kappa} + R^{\kappa}_{\lambda}{}^{\mu}_{\mu\nu} R^{\nu\mu\lambda}_{\kappa}) 
- \frac{1}{16} \frac{\delta}{\delta g} \int (R_{\mu\nu}{}^{\kappa}_{\lambda} R^{\lambda}_{\kappa}{}^{\mu\nu} - R_{\mu\nu}{}^{\kappa}_{\lambda} R^{\lambda\mu\nu} - R_{\mu\nu}{}^{\kappa}_{\lambda} R^{\mu\nu\lambda}_{\kappa} + R_{\mu\nu}{}^{\kappa}_{\lambda} R^{\nu\mu\lambda}_{\kappa}) 
+ \frac{1}{16} \frac{\delta}{\delta g} \int (R_{\nu\mu}{}^{\kappa}_{\lambda} R^{\lambda}_{\kappa}{}^{\mu\nu} - R_{\nu\mu}{}^{\kappa}_{\lambda} R^{\lambda\mu\nu} - R_{\nu\mu}{}^{\kappa}_{\lambda} R^{\mu\nu\lambda}_{\kappa} + R_{\nu\mu}{}^{\kappa}_{\lambda} R^{\nu\mu\lambda}_{\kappa}).$$

Using Proposition A.1.1 and calculating these 16 variations separately, lengthy but straightforward calculations gives

$$\frac{\delta}{\delta g} \int (R^{(5)}, R^{(5)})_{YM} = \int (\delta g_{\alpha\beta}) \left( R_{\kappa\lambda\mu}^{\ \alpha} \left( R^{\mu\beta\lambda\kappa} - R^{\lambda\kappa\mu\beta} \right) + \frac{1}{4} R_{\lambda\kappa\mu\nu} \left( R^{\kappa\lambda\mu\nu} - R^{\mu\nu\kappa\lambda} \right) g^{\alpha\beta} \right).$$

Using the explicit formula for curvature (2.54) and the results from Remark 1.4.10 and Lemma 1.4.11, we get that

$$\frac{\delta}{\delta g} \int (R^{(5)}, R^{(5)})_{YM} = \int (\delta g_{\alpha\beta}) \left( \epsilon^{\xi\beta\lambda\alpha} Ric_{\lambda\mu} Ric_{*}{}^{\mu}{}_{\xi} + \epsilon^{\kappa\mu\xi\beta} W_{\kappa\mu\lambda}{}^{\alpha} Ric_{*}{}^{\lambda}{}_{\xi} \right).$$
(2.56)

Since  $R^{(3)} = R - R^{(1)} - R^{(5)}$  we get that

$$\frac{\delta}{\delta g} \int (R^{(3)}, R^{(3)})_{YM} = \frac{\delta}{\delta g} \int (R, R)_{YM} - \frac{\delta}{\delta g} \int (R^{(5)}, R^{(5)})_{YM} - \frac{\delta}{\delta g} \int (R^{(1)}, R^{(1)})_{YM}.$$

The variation  $\frac{\delta}{\delta g} \int (R, R)_{YM}$  is given in Appendix A. Combining the formulae (2.54), (2.55), (2.56), (A.4) and using the results from Remark 1.4.10 and Lemma 1.4.11, we get that

$$\frac{\delta}{\delta g} \int (R^{(3)}, R^{(3)})_{YM} = \int (\delta g_{\alpha\beta}) (-2W^{\kappa\beta\alpha\nu} Ric_{\kappa\nu} + \epsilon^{\eta\kappa\alpha\nu} Ric_{*}{}^{\beta}{}_{\eta}Ric_{\kappa\nu} + \epsilon^{\kappa\mu\eta\alpha} W_{\kappa\mu\lambda}{}^{\beta}Ric_{*}{}^{\lambda}{}_{\eta}). \quad (2.57)$$

Combining the formulae (2.52), (2.55), (2.56), (2.57) and the Bianchi identity for the \* $\mathcal{W}$  we get the explicit representation (2.49) of the field equation (1.2).

#### 2.3.3 Variation with respect to connection

The variations  $\frac{\delta}{\delta\Gamma} \int (R^{(j)}, R^{(j)})_{YM}$  were calculated already in [99] and it was shown that

$$\frac{\delta}{\delta\Gamma} \int (R^{(j)}, R^{(j)})_{YM} = 4 \int \left( (\delta_{YM} R^{(j)})^{\mu} (\delta\Gamma)_{\mu} \right), \qquad (2.58)$$

where

$$\left(\delta_{\mathrm{YM}}R\right)^{\mu} := \frac{1}{\sqrt{|\det g|}} \left(\partial_{\nu} + \left[\Gamma_{\nu}, \cdot\right]\right) \left(\sqrt{|\det g|} R^{\mu\nu}\right)$$

is the Yang-Mills divergence. According to (2.58) and using the identity

$$\{\Gamma\}^{\xi}{}_{\xi\nu} = \frac{\partial_{\nu}|\det g|}{2|\det g|},\tag{2.59}$$

we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(1)}, R^{(1)})_{YM} = 4 \int (\delta_{YM} R^{(1)})^{\mu} (\delta\Gamma)_{\mu} =$$
  
=  $4 \int (\delta\Gamma^{\kappa}_{\ \mu\lambda}) \left( \partial_{\nu} R^{(1)}{}^{\lambda \ \mu\nu}_{\kappa} + \{\Gamma\}^{\xi}_{\ \xi\nu} R^{(1)}{}^{\lambda \ \mu\nu}_{\kappa} + \Gamma^{\lambda}_{\ \nu\eta} R^{(1)}{}^{\eta \ \mu\nu}_{\kappa} - \Gamma^{\eta}_{\ \nu\kappa} R^{(1)}{}^{\lambda \ \mu\nu}_{\eta} \right).$ 

Using the definition of covariant derivative (1.10), identity (1.14) and the fact that torsion is purely axial, we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(1)}, R^{(1)})_{YM} = 4 \int (\delta\Gamma^{\lambda\mu\kappa}) \left( \nabla_{\nu} R^{(1)}{}^{\nu}_{\kappa\lambda\mu} - \Gamma_{\mu\nu\eta} R^{(1)}{}^{\eta\nu}_{\kappa\lambda} \right).$$

Using metric compatibility, formula (1.31) and the fact that torsion is purely axial, we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(1)}, R^{(1)})_{YM} = 2 \int (\delta\Gamma^{\lambda\mu\kappa}) (\nabla_{\lambda}Ric_{\kappa\mu} - \nabla_{\kappa}Ric_{\lambda\mu} + g_{\kappa\mu}\nabla_{\xi}Ric_{\lambda}^{\ \xi} - g_{\mu\lambda}\nabla_{\xi}Ric_{\kappa}^{\ \xi} + 2Ric_{\kappa}^{\ \eta}K_{\mu\eta\lambda} + 2Ric_{\lambda}^{\ \eta}K_{\mu\kappa\eta}).$$
(2.60)

Using (2.58), (2.59) and the fact that torsion is purely axial, we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(3)}, R^{(3)})_{YM} = 4 \int (\delta_{YM} R^{(3)})^{\mu} (\delta\Gamma)_{\mu} 
= 4 \int (\delta\Gamma^{\kappa}_{\ \mu\lambda}) \left( \partial_{\nu} R^{(3)\lambda}_{\ \kappa}^{\ \mu\nu} + \{\Gamma\}^{\xi}_{\ \xi\nu} R^{(3)\lambda}_{\ \kappa}^{\ \mu\nu} + \Gamma^{\lambda}_{\ \nu\eta} R^{(3)\eta}_{\ \kappa}^{\ \mu\nu} - \Gamma^{\eta}_{\ \nu\kappa} R^{(3)\lambda}_{\ \eta}^{\ \mu\nu} \right) 
= 4 \int (\delta\Gamma^{\lambda\mu\kappa}) \left( \nabla_{\nu} \mathcal{W}_{\kappa\lambda\mu}^{\ \nu} - K_{\mu\nu\eta} \mathcal{W}_{\kappa\lambda}^{\ \eta\nu} \right).$$
(2.61)

Using (2.58), (2.59) and the fact that torsion is purely axial, we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(5)}, R^{(5)})_{YM} = 4 \int (\delta_{YM} R^{(5)})^{\mu} (\delta\Gamma)_{\mu} =$$

$$= 4 \int (\delta\Gamma^{\kappa}_{\ \mu\lambda}) \left( \partial_{\nu} R^{(5)}{}^{\lambda \ \mu\nu}_{\ \kappa} + \{\Gamma\}^{\xi}_{\ \xi\nu} R^{(5)}{}^{\lambda \ \mu\nu}_{\ \kappa} + \Gamma^{\lambda}_{\ \nu\eta} R^{(5)}{}^{\eta \ \mu\nu}_{\ \kappa} - \Gamma^{\eta}_{\ \nu\kappa} R^{(5)}{}^{\lambda \ \mu\nu}_{\ \eta} \right)$$

$$= 4 \int (\delta\Gamma^{\lambda\mu\kappa}) \left( \nabla_{\nu} R^{(5)}{}^{\kappa\lambda\mu}_{\ \kappa\lambda\mu} - \Gamma_{\mu\nu\eta} R^{(5)}{}^{\kappa\lambda\mu}_{\ \kappa\lambda} \right).$$

Using the explicit formula (B.3) for the  $R^{(5)}$  piece of curvature and Lemma 1.4.9, we get that

$$\frac{\delta}{\delta\Gamma} \int (R^{(5)}, R^{(5)})_{YM} = 2 \int (\delta\Gamma^{\lambda\mu\kappa}) \left( -\epsilon^{\eta}{}_{\lambda\mu}{}^{\nu}\nabla_{\nu}Ric_{*\kappa\eta} + \epsilon^{\eta}{}_{\kappa\mu}{}^{\nu}\nabla_{\nu}Ric_{*\lambda\eta} \right) + \epsilon^{\eta}{}_{\lambda}{}^{\xi\nu}K_{\mu\nu\xi}Ric_{*\kappa\eta} - \epsilon^{\eta}{}_{\kappa}{}^{\xi\nu}K_{\mu\nu\xi}Ric_{*\lambda\eta} \right).$$
(2.62)

Combining the formulae (2.52), (2.60), (2.61) and (2.62) we get that the second field equation (1.3) is given explicitly by

$$0 = c_1 (\nabla_{\lambda} Ric_{\kappa\mu} - \nabla_{\kappa} Ric_{\lambda\mu} + g_{\kappa\mu} \nabla_{\xi} Ric_{\lambda}^{\xi} - g_{\mu\lambda} \nabla_{\xi} Ric_{\kappa}^{\xi}) + 2c_1 (K_{\mu\eta\lambda} Ric_{\kappa}^{\eta} + K_{\mu\kappa\eta} Ric_{\lambda}^{\eta}) + 2c_3 (\nabla_{\nu} \mathcal{W}_{\kappa\lambda\mu}^{\nu} - K_{\mu\nu\eta} \mathcal{W}_{\kappa\lambda}^{\eta\nu}) + c_5 (-\epsilon^{\eta}{}_{\lambda\mu}^{\nu} \nabla_{\nu} Ric_{*\kappa\eta} + \epsilon^{\eta}{}_{\kappa\mu}^{\nu} \nabla_{\nu} Ric_{*\lambda\eta}) + c_5 (+\epsilon^{\eta}{}_{\lambda}^{\xi\nu} K_{\mu\nu\xi} Ric_{*\kappa\eta} - \epsilon^{\eta}{}_{\kappa}^{\xi\nu} K_{\mu\nu\xi} Ric_{*\lambda\eta}), \qquad (2.63)$$

where  $c_1, c_3, c_5$  are the coefficients of the quadratic form (1.4).

Now, using the Bianchi identity (B.5), we can explicitly express the terms  $\nabla Ric$  and  $\nabla W$  containing one contraction, see Appendix B for detailed calculations, to get that

$$\nabla_{\xi} Ric^{\xi}{}_{\lambda} = -K^{\eta}{}_{\xi\zeta} \mathcal{W}^{\xi\zeta}{}_{\lambda\eta} - \epsilon^{\vartheta\zeta}{}_{\eta\lambda} K^{\eta}{}_{\xi\zeta} Ric_{*}{}^{\xi}{}_{\vartheta}$$
(2.64)

and

$$\nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi} = -\frac{1}{2}\left(\nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\nu}Ric_{\lambda\xi} + 2Ric^{\mu}{}_{\xi}K_{\nu\mu\lambda} + 2Ric^{\mu}{}_{\nu}K_{\mu\xi\lambda} + \epsilon^{\vartheta\mu}{}_{\nu\xi}\nabla_{\mu}Ric_{*\lambda\vartheta} - \epsilon^{\vartheta}{}_{\lambda\nu\xi}\nabla_{\mu}Ric_{*}{}^{\mu}{}_{\vartheta}\right) \\
- K^{\eta}{}_{\mu\xi}\left(\epsilon^{\vartheta}{}_{\lambda\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*\lambda\vartheta}\right) - K^{\eta}{}_{\mu\nu}\left(\epsilon^{\vartheta}{}_{\lambda\xi\eta}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\xi\eta}Ric_{*\lambda\vartheta}\right) \\
+ \frac{1}{2}K^{\eta}{}_{\mu\zeta}\left(g_{\lambda\xi}\epsilon^{\vartheta\zeta}{}_{\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - g_{\lambda\nu}\epsilon^{\vartheta\zeta}{}_{\eta\xi}Ric_{*}{}^{\mu}{}_{\vartheta}\right) \\
+ \frac{1}{2}K^{\eta}{}_{\mu\zeta}\left(g_{\lambda\xi}\mathcal{W}^{\mu\zeta}{}_{\nu\eta} - g_{\lambda\nu}\mathcal{W}^{\mu\zeta}{}_{\xi\eta}\right) - 2K^{\eta}{}_{\nu\mu}\mathcal{W}^{\mu}{}_{\lambda\xi\eta} - 2K^{\eta}{}_{\mu\xi}\mathcal{W}^{\mu}{}_{\lambda\nu\eta}.$$
(2.65)

Substituting (2.64) and (2.65) into (2.63), we exclude the terms  $\nabla Ric$  and  $\nabla W$  from equation (2.63) and hence we get the explicit representation (2.50) of the field equation (1.3).

This concludes the proof of Theorem 2.3.10.

#### 2.3.4 PP-type solutions of the field equations

The aim of this section is to prove Theorem 2.3.4, i.e. that generalised ppwaves with purely axial torsion of parallel Ricci curvature are solutions of (1.2), (1.3) in the case (1.4).

**Proof** of Theorem 2.3.4. We will prove this theorem by the direct substitution of the explicit formulae for torsion (2.34), (2.39), Weyl curvature (2.13) and the Ricci curvature (2.44) and  $Ric_*$  curvature (2.45) in the explicit representation of the field equations (2.49), (2.50). Also, we will use equations (2.8), (2.9) in order to simplify our calculations.

The terms  $\mathcal{W} \times Ric$ ,  $\mathcal{W} \times Ric_*$  and  $Ric \times Ric_*$  with one contraction are zero since the vectors l,  $m_1$  and  $m_2$  are orthogonal, see (2.9). Hence, equation (2.49) is satisfied.

Let us now consider the equation (2.50). We will first consider the terms  $T \times W$  with two contractions. Using formulae (2.9), (2.13) and (2.39), we have that

$$T^{\eta}{}_{\lambda\xi}\mathcal{W}^{\xi}{}_{\mu\kappa\eta} = \mp k(x^3)(l \wedge m_1 \wedge m_2)^{\eta}{}_{\lambda\xi}\sum_{j,k=1}^2 w_{jk}(l \wedge m_j)^{\xi}{}_{\mu}(l \wedge m_k)_{\kappa\eta}$$
$$= \mp k(x^3)(-w_{12}l_{\lambda}l_{\mu}l_{\kappa} + w_{21}l_{\lambda}l_{\mu}l_{\kappa}) = 0,$$

since  $w_{12} = w_{21}$ . Consequently, the terms  $T \times W$  with three contractions are also zero.

The terms  $T \times Ric$  with one contraction are equal to zero since

$$T_{\mu\eta\lambda}Ric_{\kappa}^{\eta} = \frac{1}{2} \left( f_{11} + f_{22} - (k(x^3))^2 \right) k(x^3) l^{\xi} \epsilon_{\xi\mu\eta\lambda} l_{\kappa} l^{\eta}$$
$$= \frac{1}{2} \left( f_{11} + f_{22} - (k(x^3))^2 \right) k(x^3) l^{\xi} l^{\eta} \epsilon_{\xi\eta\mu\lambda} l_{\kappa} = 0$$

as the product of the symmetric tensor  $l^\xi l^\eta$  and the antisymmetric Levi-Civita tensor.

Similarly, the terms  $T \times Ric_*$  with one contraction are equal to zero, since we get the product of a symmetric tensor and an antisymmetric tensor.

Now consider the terms  $\epsilon \times T \times Ric_*$  with two contractions between  $\epsilon$  and T and one contraction between  $\epsilon$  and  $Ric_*$ . Using formulae (1.49), (2.34) and (2.45), we get that

$$\epsilon^{\vartheta\xi}{}_{\eta\lambda}T^{\eta}{}_{\xi\kappa}Ric_{*\mu\vartheta} = 2k(x^3)k'(x^3)l_{\kappa}l_{\lambda}l_{\mu}.$$

Hence

$$\epsilon^{\vartheta\xi}{}_{\eta\lambda}T^{\eta}{}_{\xi\kappa}Ric_{*\mu\vartheta} - \epsilon^{\vartheta\xi}{}_{\eta\kappa}T^{\eta}{}_{\xi\lambda}Ric_{*\mu\vartheta} = 0.$$

Now consider the terms  $\epsilon \times \nabla Ric_*$ . Since  $\Gamma^{\eta}_{\xi\mu}Ric_{*\eta\vartheta} = 0$ , we have that

$$\nabla_{\xi} Ric_{*\mu\vartheta} = \partial_{\xi} Ric_{*\mu\vartheta} - \Gamma^{\eta}{}_{\xi\mu} Ric_{*\eta\vartheta} - \Gamma^{\eta}{}_{\xi\vartheta} Ric_{*\mu\eta} = -k''(x^3) l_{\xi} l_{\mu} l_{\vartheta}.$$

Clearly  $\nabla_{\xi} Ric_*^{\xi}{}_{\eta} = -k''(x^3)l_{\xi}l^{\xi}l_{\eta} = 0$ . The only nonzero term of  $\nabla_{\xi} Ric_{*\mu\vartheta}$  is  $\nabla_3 Ric_{*33} = -k''(x^3)$  and consequently we have that the term  $\epsilon \times \nabla Ric_*$  with two contractions between  $\epsilon$  and  $\nabla Ric_*$  are zero. This is because

$$\epsilon^{\vartheta\xi}{}_{\lambda\kappa}\nabla_{\xi}Ric_{*\mu\vartheta} = \epsilon^{33}{}_{\lambda\kappa}\nabla_{3}Ric_{*33} = 0.$$

Equation (2.50) now reduces to checking that

$$\nabla_{\lambda} Ric_{\kappa\mu} - \nabla_{\kappa} Ric_{\lambda\mu} = 0. \tag{2.66}$$

Assuming that Ricci curvature is parallel, equation (2.66) is clearly satisfied. However, we see that parallel Ricci curvature is in fact not required, only the Cotton tensor of generalised pp-waves with purely axial torsion needs to vanish.

This ends the proof of Theorem 2.3.4.

**Remark 2.3.13.** The Cotton tensor  $\nabla_{\lambda} Ric_{\kappa\mu} - \nabla_{\kappa} Ric_{\lambda\mu}$  of generalised ppwaves with purely axial torsion vanishes if and only if  $f_{11} + f_{22} = \text{const.}$  This condition is equivalent to the condition  $\{\nabla\}\{Ric\} = 0$ , i.e. that the Ricci curvature of the classical pp-wave is parallel.

## 2.4 The massless Dirac operator in theories of gravity

In this section we give the mathematical and physical significance of the spacetimes which are considered in Section 2.2.2 and we attempt to give a physical interpretation of these spacetimes as was done similarly in [74]. As we stated in [74], classical pp-waves of parallel Ricci curvature do not have an obvious physical interpretation and therefore they should not be viewed separately. Our analysis of the generalised pp-waves of parallel Ricci curvature are just the part of much wider class of solutions. Analysing the formula for curvature of the generalised pp-waves with purely axial torsion, we notice that the curvature in our special local coordinates (2.1), (2.5) is the sum of the curvature of the underlying classical pp-space

$$-\frac{1}{2}(l \wedge \{\nabla\}) \otimes (l \wedge \{\nabla\})f \tag{2.67}$$

and the curvature

$$\frac{1}{4}(k(\varphi))^2 \operatorname{Re}\left((l \wedge m) \otimes (l \wedge \bar{m})\right) \mp \frac{1}{2}k'(\varphi) \operatorname{Im}\left((l \wedge m) \otimes (l \wedge \bar{m})\right) \quad (2.68)$$

generated by an axial torsion wave traveling over the pp-space. This remarkable property for curvature is not a trivial fact. Similarly, the property that curvatures just add up was also present in the case of generalised pp-waves with purely tensor torsion, see [74]. The physical interpretation of generalised pp-waves with purely tensor torsion introduced in Section 2.2.1 was given in [74] where it was proposed that generalised pp-waves with purely tensor torsion of parallel Ricci curvature represent a metric-affine model for the massless neutrino. Hence, in order to give physical interpretation of generalised pp-waves with purely axial torsion, we now compare these spacetimes to the solutions of Einstein-Weyl theory. Einstein-Weyl theory is the classical model which describes the interaction of gravitational and massless neutrino fields.

**Remark 2.4.1.** Our torsion and torsion generated curvature can be interpreted as waves traveling at speed of light. The underlying classical pp-space of parallel Ricci curvature can then be viewed as the gravitational imprint created by a wave of some massless matter field. As pointed out in [74], such a situation occurs in Einstein-Weyl theory.

In line with the traditions of quantum mechanics, we choose to complexify the curvature (2.68). The complexified curvature can be written as

$$\mathfrak{R}_A := r \, (l \wedge m) \otimes (l \wedge \overline{m}), \tag{2.69}$$

where

$$r := \frac{1}{4} (k(\varphi))^2 \pm \frac{i}{2} k'(\varphi)$$
 (2.70)

is a complex function. The function r is a function of the phase (2.4) and the curvature (2.68) is the real part of the complexified curvature (2.69). The curvature  $\mathfrak{R}_A$  is polarized, i.e.

$$^*\mathfrak{R}_A = -\mathfrak{R}_A^* = \pm \mathrm{i}\,\mathfrak{R}_A,$$

where the  $\pm$  sign depends on the sign of (2.5). It can also be written as

$$\Re_{A\,\alpha\beta\gamma\delta} = \sigma_{\alpha\beta ab}\,\omega^{abcd}\,\overline{\sigma}_{\gamma\delta cd} \tag{2.71}$$

where  $\omega$  is some symmetric rank 4 spinor and  $\sigma_{\alpha\beta}$  are second order Pauli matrices (2.16) where  $\overline{\sigma}$  denotes their complex conjugation. The complex conjugate matrices  $\overline{\sigma}$  are exactly the same set of matrices (2.16) only with the opposite sign chosen to correspond with the sign in (2.5).

Resolving (2.71) with respect to  $\omega$  yields

$$\omega = \xi \otimes \xi \otimes \xi \otimes \xi, \qquad (2.72)$$

where

$$\xi^a := r^{1/4} \chi^a. \tag{2.73}$$

Note that the spinor  $\chi^a$  which appears in formula (2.73) is the parallel spinor field of the underlying pp-space (2.5) and the complex function r is given by (2.70).

Formula (2.72) shows that the rank 4 spinor  $\omega$  is the 4th tensor power of a rank 1 spinor  $\xi$ . Hence, the curvature  $\Re_A$  is completely determined by the rank 1 spinor field  $\xi$ .

Interestingly, we can now establish a connection between the generalised pp-waves with purely axial torsion and a massless neutrino field. A massless neutrino field is a metric compatible spacetime (with or without torsion) described by the action

$$S_{\text{neutrino}} := 2i \int \left( \xi^a \, \sigma^{\mu}{}_{ab} \left( \nabla_{\mu} \bar{\xi}^b \right) - \left( \nabla_{\mu} \xi^a \right) \sigma^{\mu}{}_{ab} \, \bar{\xi}^b \right), \qquad (2.74)$$

see formula (11) of [44]. Varying the action (2.74) with respect to the spinor  $\xi$ , while keeping torsion and the metric fixed, we get the *massless Dirac* equation

$$\sigma^{\mu}{}_{ab}\nabla_{\mu}\xi^{a} - \frac{1}{2}T^{\eta}{}_{\eta\mu}\sigma^{\mu}{}_{ab}\xi^{a} = 0, \qquad (2.75)$$

which can equivalently be written as

$$\sigma^{\mu}{}_{ab}\{\nabla\}_{\mu}\xi^{a} \pm \frac{i}{4}\varepsilon_{\alpha\beta\gamma\delta}T^{\alpha\beta\gamma}\sigma^{\delta}{}_{ab}\xi^{a} = 0, \qquad (2.76)$$

where  $\{\nabla\}$  is the covariant derivative with respect to the Levi-Civita connection, see Appendix B of [74]. Also, using the massless Dirac operator, see Section 3.2, the massless Dirac equation can be obtained by varying the action (3.12) with respect to the spinor  $\xi$ . The massless Dirac operator describes a single massless neutrino living in a compact universe and physically its eigenvalues are interpreted to be the energy levels of that massless particle. An interesting property is stated by the following

Lemma 2.4.2. The spinor field (2.73) satisfies the massless Dirac equation.

*Proof.* Since the classical pp-wave spacetime admits the parallel spinor field  $\chi$  then

$$\sigma^{\mu}{}_{ab}\{\nabla\}_{\mu}\xi^{a} = (r^{1/4})'\sigma^{\mu}{}_{ab}l_{\mu}\chi^{a} = 0$$

for the Pauli matrices (2.15) and the special local coordinates (2.5). Also, according to formulae (1.48), (2.34) and the Pauli matrices (2.15), we have that

$$\varepsilon_{\alpha\beta\gamma\delta}T^{\alpha\beta\gamma}\sigma^{\delta}{}_{ab}\xi^{a} = \varepsilon_{\alpha\beta\gamma\delta}l_{\mu}k(\varphi)\varepsilon^{\mu\alpha\beta\gamma}\sigma^{\delta}{}_{ab}\xi^{a} = 6k(\varphi)l_{\mu}\sigma^{\mu}{}_{ab}\xi^{a} = 0,$$

i.e. the massless Dirac equation (2.76) is satisfied.

In an attempt to give a physical interpretation of our generalised ppwaves with purely axial torsion, we provide the pp-wave type solution of Einstein-Weyl theory and compare them to the pp-wave type solutions of our conformally invariant metric-affine model of gravity.

#### 2.4.1 Metric-affine vs Einstein-Weyl pp-wave type solutions

In Einstein-Weyl theory we consider the action as

$$S_{EW} := 2i \int \left( \xi^a \, \sigma^{\mu}_{\ ab} \left( \{ \nabla \}_{\mu} \overline{\xi}^b \right) - \left( \{ \nabla \}_{\mu} \xi^a \right) \sigma^{\mu}_{\ ab} \, \overline{\xi}^b \right) + K \int \mathcal{R}, \quad (2.77)$$

where the constant  $K = \frac{c^4}{16\pi G}$  is the universal constant where c is the speed of light and G is the gravitational constant, see [63]. In Einstein-Weyl theory the connection is assumed to be Levi-Civita, so we obtain the Einstein-Weyl field equations by varying the action (2.77) with respect to the metric and the spinor, i.e.

$$\frac{\delta S_{EW}}{\delta g} = 0, \qquad (2.78)$$

$$\frac{\delta S_{EW}}{\delta \xi} = 0. \tag{2.79}$$

The variation of the first term of the action (2.77) with respect to the metric yields the energy momentum tensor of the Weyl action (2.74). For the detailed derivation of formula for the energy momentum tensor see Appendix B of [74]. The explicit representation of the Einstein-Weyl field equations (2.78), (2.79) is given by

$$\begin{split} \frac{\mathrm{i}}{2} & \left[ \sigma^{\nu}{}_{ab} \left( \overline{\xi}^{b} \{ \nabla \}^{\mu} \xi^{a} - \xi^{a} \{ \nabla \}^{\mu} \overline{\xi}^{b} \right) + \sigma^{\mu}{}_{ab} \left( \overline{\xi}^{b} \{ \nabla \}^{\nu} \xi^{a} - \xi^{a} \{ \nabla \}^{\nu} \overline{\xi}^{b} \right) \right] \\ & + \mathrm{i} \left( \xi^{a} \sigma^{\eta}{}_{ab} \left( \{ \nabla \}_{\eta} \overline{\xi}^{b} \right) g^{\mu\nu} - \left( \{ \nabla \}_{\eta} \xi^{a} \right) \sigma^{\eta}{}_{ab} \overline{\xi}^{b} g^{\mu\nu} \right) \\ & - KRic^{\mu\nu} + \frac{K}{2} \mathcal{R} g^{\mu\nu} = 0, \quad (2.80) \\ & \sigma^{\mu}{}_{ab} \{ \nabla \}_{\mu} \xi^{a} = 0. \quad (2.81) \end{split}$$

The examination of the Einstein-Weyl field equations has a long history, see e.g. [7, 21, 22, 42, 43, 45, 46, 56]. A review of known solutions of Einstein-Weyl theory is given in [72, 74]. The nonlinear system of equations (2.80), (2.81) has solutions in the form of pp-waves, as stated in [72, 74]. We now wish to present a class of explicit solutions of (2.80), (2.81) where the metric g is in the form of a pp-metric (2.1) and the spinor  $\xi$  as in (2.73). The spinor (2.73) satisfies the massless Dirac equation (2.81), see Lemma 2.4.2. Because of this and the fact that the scalar curvature is zero in the setting of a pp-space, the equation (2.80) now becomes

$$\frac{\mathrm{i}}{2}\sigma^{\nu}{}_{ab}\left(\overline{\xi}^{b}\{\nabla\}^{\mu}\xi^{a}-\xi^{a}\{\nabla\}^{\mu}\overline{\xi}^{b}\right)+\frac{\mathrm{i}}{2}\sigma^{\mu}{}_{ab}\left(\overline{\xi}^{b}\{\nabla\}^{\nu}\xi^{a}-\xi^{a}\{\nabla\}^{\nu}\overline{\xi}^{b}\right)-KRic^{\mu\nu}=0.$$

Substituting formulae (2.44), (2.73) into the above equation, we get that

$$i(\sigma^{\nu}{}_{ab}l^{\mu} + \sigma^{\mu}{}_{ab}l^{\nu}) \Big( (r^{1/4})' \ \overline{r^{1/4}} - r^{1/4} \ (\overline{r^{1/4}})' \Big) \chi^a \overline{\chi}^{\dot{b}} = K l^{\mu} l^{\nu} \big( f_{11} + f_{22} - k(x^3)^2 \big) \,.$$

The condition for a pp-wave type solution needs to satisfy in order to be a solution of Einstein-Weyl is

$$f_{11} + f_{22} = k(\varphi)^2 + \frac{2i}{K} \left( (r^{1/4})' \ \overline{r^{1/4}} - r^{1/4} \ (\overline{r^{1/4}})' \right), \qquad (2.82)$$

since  $\sigma^{\mu}{}_{ab}\chi^{a}\overline{\chi}^{b} = l^{\mu}$ . Since the function  $k(\varphi)$  is arbitrary real function hence the complex function  $r(\varphi)$  can be chosen arbitrarily and it uniquely determines the RHS of (2.82).

**Remark 2.4.3.** As was stated in [74], the main difference between the two models is that in the metric-affine model the generalised pp-wave solutions have parallel *Ric* curvature, whereas in the Einstein-Weyl model the pp-wave type solutions do not necessarily have parallel Ricci curvature.

The comparison of this two types of solutions becomes much clearer in the case of the monochromatic solutions. Similarly as was done in [74], if we choose the function  $k(x^3)$  such that the function (2.70) is equal to

$$r = c^4 e^{4i(ax^3 + b)},$$

where  $a, b, c \in \mathbb{R}$ ,  $a \neq 0$ , then the spinor  $\xi$  form (2.73) is explicitly given by

$$\xi = c \begin{pmatrix} 1\\ 0 \end{pmatrix} e^{\mathbf{i}(ax^3 + b)}.$$
 (2.83)

The vector field A from Definition 2.2.5 is  $A = k(x^3)l$  where the function  $k(x^3)$  is the solution of the differential equation

$$\frac{1}{4}(k(x^3))^2 \pm \frac{\mathrm{i}}{2}k'(x^3) = c^4 e^{4\mathrm{i}(ax^3+b)}$$

where  $a, b, c \in \mathbb{R}$ ,  $a \neq 0$ . For the spinor field (2.83), the condition (2.82) reads

$$f_{11} + f_{22} = (k(x^3))^2 - \frac{4ac^2}{K}.$$

Hence, we conclude that in the metric-affine case the Laplacian of f can be any constant, while in the Einstein-Weyl case it is required for it to be a particular constant, which is the consequence of conformal invariance of the metric-affine model and the presence of the gravitational constant in the Einstein-Weyl.

The generalised pp-waves of parallel Ricci curvature are very similar to pp-type solutions of the Einstein-Weyl model. According this conclusion, similarly to [74], we propose that generalised pp-waves with purely axial torsion of parallel Ricci curvature represent a metric-affine model for the massless neutrino.

## Chapter 3

# Spectral Analysis of the Massless Dirac Operator on a 3-dimensional Manifold

Generalised pp-waves with purely axial and purely tensor torsion of parallel Ricci curvature considered in this thesis have their particular physical interpretation as was shown in the previous chapter. The spinor field  $\xi$ , which completely determines the complexified curvatures of those spacetimes, satisfies the massless Dirac equation. The axial torsion considered in previous sections is the irreducible piece of torsion which is usually used when one models the massless neutrino, see [20], or the electron, see [16], by means of Cosserat elasticity. Now we are interested in a more mathematical approach and the analysis of the massless Dirac equation and the massless Dirac operator in 3 dimensions.

In this chapter we consider the massless Dirac operator on a 3-manifold which describes a single massless neutrino living in a 3-dimensional compact universe. The eigenvalues of the massless Dirac operator are interpreted to be the energy levels of that massless particle. The eigenvalues of the massless Dirac operator can be explicitly calculated when we consider the unit torus  $\mathbb{T}^3$  equipped with Euclidean metric and the unit sphere  $\mathbb{S}^3$  equipped with metric induced by the natural embedding of  $\mathbb{S}^3$  in Euclidean space  $\mathbb{R}^4$ . It turns out that in these cases the spectrum of the massless Dirac operator is symmetric.

However, according to [3, 4, 5, 6], for a general oriented Riemannian 3manifold (M, g) there is no physical reason for the spectrum of the massless Dirac operator to be symmetric. Physically interpreted, this symmetry would mean that in these two examples, the massless neutrino and the massless antineutrino have the same properties. Our goal is to prove that it is possible break the spectral symmetry of the massless Dirac operator on a 3-torus using the perturbation of the Euclidean metric. We found justification for doing this in the numerical analysis of the spectrum of the massless Dirac operator. We use the Galerkin method, see e.g. [49], to calculate the spectrum explicitly for different perturbations of the Euclidean metric. Perturbing the Euclidean metric for some positive real parameter  $\epsilon$ , we derive the asymptotic formulae for the eigenvalues of the massless Dirac operator, which depends to the parameter  $\epsilon$ . We analyse under which perturbations of the metric it is possible to obtain the spectral asymmetry.

In the spectral analysis of the massless Dirac operator we will use some known results for an elliptic self-adjoint first-order differential operator, see [18, 19, 25].

## 3.1 Some properties of an elliptic self-adjoint first-order differential operator

One way to analyse the spectrum of an operator is to consider the distribution of the eigenvalues of that operator. Hence, we will be interested in studying the *spectral function*, see Definition 3.1.5, and *the counting function*, see Definition 3.1.6, of the massless Dirac operator. The massless Dirac operator is a self-adjoint first order differential operator and it has a discrete spectrum with eigenvalues accumulating to  $+\infty$  and  $-\infty$ , while the eigenfunctions of the operator are infinitely smooth, see [19, 24, 25].

Let M be a connected compact 3-dimensional manifold without boundary and let  $x = (x^1, x^2, x^3)$  be the local coordinates on the manifold M. Consider a first order differential operator A which is self-adjoint acting on 2-columns  $v = (v_1 \quad v_2)^T$  of complex-valued half-densities over a manifold M.

The *principal* symbol and the *subprincipal* symbol of the first order differential operator A, which will be used through this chapter, are defined as follows, see [85].

**Definition 3.1.1.** The principal symbol of the first order differential operator A is a matrix obtained by leaving in A only the leading first order derivatives and replacing each  $\partial/\partial x^{\alpha}$  by  $i\xi_{\alpha}$ ,  $\alpha = 1, 2, 3$ , where  $\xi = (\xi_1, \xi_2, \xi_3)$  is the variable dual to the position variable x which in physics literature is referred to as *momentum*. We denote the principal symbol of the operator A by  $A_1(x, \xi)$ .

As was shown in [19], the existence of a principal symbol  $A_1(x,\xi)$  implies that our manifold M is parallelisable and the principal symbol admits the metric and the teleparallel connection  $\Gamma^{\alpha}{}_{\beta\gamma}(x)$ . This allows us to express the results of our spectral analysis in a geometric language and to define the torsion tensor by (1.12).

**Definition 3.1.2.** The subprincipal symbol of the first order differential operator A is defined as

$$A_{\rm sub} := A_0 + \frac{i}{2} (A_1)_{x^{\alpha} \xi_{\alpha}}, \qquad (3.1)$$

where  $A_1(x,\xi)$  and  $A_0(x)$  are the components of the full symbol  $A(x,\xi) = A_1(x,\xi) + A_0(x)$  of our first order differential operator, with the subscript indicating the degree of homogeneity.

**Remark 3.1.3.** We assume that the principal symbol  $A_1(x,\xi)$  is trace free for all  $(x,\xi) \in T^*M$  and that det  $A_1(x,\xi) \neq 0, \forall (x,\xi) \in T'M$  where T'M := $T^*M \setminus \{\xi = 0\}$ . The principal symbol of our operator A is a 2 × 2 Hermitian matrix-function on the cotangent bundle  $T^*M$  and linear in  $\xi$ .

**Remark 3.1.4.** It is known, see [18, 19], that under these assumptions, the spectrum of the operator A is discrete with eigenvalues accumulating to  $\pm \infty$ .

We denote by  $\lambda_k$  the eigenvalues of the operator A and by  $v_k(x)$  the corresponding eigenvectors. The eigenvalues  $\lambda_k$  are enumerated in increasing order using  $k = 1, 2, \ldots$  for positive eigenvalues and  $k = 0, -1, -2, \ldots$  for non-positive eigenvalues.

For the purposes of further analysis of the spectrum of the operator, we are interested in the analysis of two functions, namely the spectral function and the counting function, which are defined as follows, see [19, 85].

**Definition 3.1.5.** The spectral function is the real density defined as

$$e(\lambda, x, x) := \sum_{0 < \lambda_k < \lambda} \parallel v_k(x) \parallel^2, \qquad (3.2)$$

where  $|| v_k(x) ||^2 := [v_k(x)]^* v_k(x)$  is the square of the Euclidean norm of the eigenfunction  $v_k$  evaluated at the point  $x \in M$  and  $\lambda$  is a positive parameter (spectral parameter).

**Definition 3.1.6.** The counting function is the function

$$N(\lambda) := \sum_{0 < \lambda_k < \lambda} 1 = \int_M e(\lambda, x, x) dx, \qquad (3.3)$$

where  $e(\lambda, x, x)$  is the spectral function (3.2).

The counting function  $N(\lambda)$  is actually the number of eigenvalues between zero and  $\lambda$ . Also, we are interested in the asymptotic formulae of the spectral function (3.2) and the counting function (3.3), i.e. we are interested in deriving the formulae of the type

$$e(\lambda, x, x) = a(x)\lambda^3 + b(x)\lambda^2 + o(\lambda^2), \qquad (3.4)$$

$$N(\lambda) = a\lambda^3 + b\lambda^2 + o(\lambda^2) \tag{3.5}$$

as  $\lambda \to +\infty$ , where the real constants a, b and real densities a(x), b(x) are related in accordance with

$$a = \int_{M} a(x)dx, \qquad (3.6)$$

$$b = \int_{M} b(x) dx. \tag{3.7}$$

The asymptotic formulae (3.4) and (3.5) for the first order differential operator are explicitly derived in the following theorem.

**Theorem 3.1.7.** The coefficients in the two-term asymptotic formula (3.4) are given by

$$a(x) = \frac{1}{6\pi^2} \sqrt{\det g_{\alpha\beta}(x)},$$
  
$$b(x) = \frac{1}{8\pi^2} \left( (3c * T^{\mathrm{ax}} - 2\mathrm{tr}A_{\mathrm{sub}}) \sqrt{\det g_{\alpha\beta}} \right)(x),$$

where

$$T_{\alpha\beta\gamma}^{\mathrm{ax}} := \frac{1}{3} \left( T_{\alpha\beta\gamma} + T_{\gamma\alpha\beta} + T_{\beta\gamma\alpha} \right)$$

is axial torsion, c is topological charge defined by

$$c := -\frac{i}{2}\sqrt{\det g_{\alpha\beta}} \operatorname{tr}((A_1)_{\xi_1}(A_1)_{\xi_2}(A_1)_{\xi_3}),$$

which can only take two values, +1 or -1 and where \* is the Hodge star operator (1.21).

For the proof of Theorem 3.1.7 see e.g. [19].

### 3.2 The massless Dirac operator

In this section we introduce the massless Dirac operator on a 3-dimensional manifold and we recall its main properties, as was similarly done in [19, 24].

For more on how the massless Dirac operator acts on a manifold of arbitrary dimension, see [36, 41].

Let M be a 3-dimensional connected compact oriented manifold equipped with a Riemannian metric  $g_{\alpha\beta}$ , where  $\alpha, \beta = 1, 2, 3$  are tensor indices. According to [54], a 3-dimensional oriented manifold is parallelisable and consequently, there exist smooth real vector fields  $e_j(x)$ , j = 1, 2, 3 that are linearly independent in every point x of the manifold M.

The triple of linearly independent vector fields  $e_j(x)$ , j = 1, 2, 3, is called a *frame*. We can assume that the vector fields  $e_j(x)$  are orthonormal, and if not, the orthonormality can always be achieved using the Gram-Schmidt process. The coordinate components of the vector  $e_j(x)$  are  $e_j^{\alpha}(x)$ ,  $\alpha = 1, 2, 3$ , where the so-called anholonomic or *frame index*, denoted by the Latin letter j, enumerates the vector field and the holonomic or *tensor index*, denoted by Greek letter  $\alpha$ , enumerates their components.

The *coframe* is defined as the triple of covector fields  $e^k(x)$ , k = 1, 2, 3, and the coordinate components of the vector  $e^k(x)$  are  $e^k_{\alpha}(x)$ ,  $\alpha = 1, 2, 3$ , where

$$e^k{}_\beta := \delta^{kj} g_{\beta\gamma} e_j{}^\gamma.$$

The frame is completely determined by the coframe, and vice versa, by the relation  $e_j^{\alpha} e^k_{\alpha} = \delta_j^k$ .

**Definition 3.2.1.** The massless Dirac operator is defined to be the matrix operator

$$W := -\mathrm{i}\sigma^{\alpha} \left( \frac{\partial}{\partial x^{\alpha}} + \frac{1}{4}\sigma_{\beta} \left( \frac{\partial\sigma^{\beta}}{\partial x^{\alpha}} + \left\{ \frac{\beta}{\alpha\gamma} \right\} \sigma^{\gamma} \right) \right), \qquad (3.8)$$

where summation is carried out over  $\alpha, \beta, \gamma = 1, 2, 3$ .

**Remark 3.2.2.** We denote the massless Dirac operator with the Latin letter "W" because in theoretical physics literature it is often referred to as the *Weyl operator*.

**Remark 3.2.3.** The massless Dirac operator can be thought of as a square root of the Laplacian.

The coefficients  $\begin{pmatrix} \alpha \\ \beta \gamma \end{pmatrix}$  appearing in the Definition 3.2.1 are the Christoffel symbols (1.11). The matrices  $\sigma$  are the Pauli matrices defined as

$$\sigma^{\alpha}(x) := s^{j} e_{j}^{\ \alpha}(x), \tag{3.9}$$

where summation is carried out over the repeated frame index j = 1, 2, 3, the index  $\alpha = 1, 2, 3$  is the free tensor index and the matrices  $s^{j}$  and  $s_{j}$ , j = 1, 2, 3 are defined with

$$s^{1} := \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} = s_{1}, \ s^{2} := \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} = s_{2}, \ s^{3} := \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} = s_{3}.$$
(3.10)

The massless Dirac operator (3.8) acts on 2-columns

$$v = \left(\begin{array}{cc} v_1 & v_2 \end{array}\right)^T \tag{3.11}$$

of complex-valued scalar functions which are referred to as the Weyl spinor. The spinor (3.11) transforms in a particular way under transformations of the orthonormal frame  $e_j(x)$ . We choose the frame *a priori* so that we can treat the components of the spinor as scalars.

The raising and of lowering indices follows the standard convention using the metric tensor.

Using the massless Dirac operator (3.8) we can construct the massless Dirac action - the variational functional corresponding to the operator (3.8) which we consider in Section 2.4.

**Definition 3.2.4.** The massless Dirac action is defined as

$$S(\zeta) := \int_M \operatorname{Re}(\zeta^* W \zeta) \sqrt{\det g_{\alpha\beta}} dx, \qquad (3.12)$$

where W is the massless Dirac operator (3.8) and the star indicates Hermitian conjugation.

The orientation (positive or negative) of the massless Dirac operator (3.8) is completely determined by the sign of the frame. Our frame has positive orientation if det  $e_j > 0$  and negative orientation if det  $e_j < 0$ . Note that the transformation  $W \mapsto -W$  changes the orientation of the massless Dirac operator.

Physically interpreted, the operator (3.8) describes a single massless neutrino living in a 3-dimensional compact universe M and the energy levels of such particle are determined by the eigenvalues of the operator. We now list the main properties of the massless Dirac operator (3.8), see [17, 29] for their proofs:

- The massless Dirac operator is invariant under changes of local coordinates.
- The massless Dirac operator is an elliptic operator.

• The massless Dirac operator is formally self-adjoint with respect to the inner product

$$\langle v, w \rangle := \int_{M} w^* v \sqrt{\det g_{\alpha\beta}} dx$$
 (3.13)

on 2-columns of smooth scalar functions  $v, w : M \to \mathbb{C}^2$ .

• It is of a special interest in theoretical physics to view the antilinear map called the *charge conjugation*, which is defined by

$$v \mapsto C(v) := \epsilon \overline{v}, \tag{3.14}$$

where

$$\epsilon := \left( \begin{array}{cc} 0 & -1 \\ 1 & 0 \end{array} \right).$$

The antilinear operator of charge conjugation (3.14) maps any element of  $L^2(M; \mathbb{C}^2)$  to an element of  $L^2(M; \mathbb{C}^2)$  and any element of  $H^1(M; \mathbb{C}^2)$ to an element of  $H^1(M; \mathbb{C}^2)$ .

**Remark 3.2.5.** An interesting property is that the massless Dirac operator W commutes with the charge conjugation operator, i.e.

$$C(Wv) = W(C(v)).$$
 (3.15)

• Let  $R: M \to SU(2)$  be an arbitrary smooth special unitary matrixfunction. Let us introduce new Pauli matrices

$$\widetilde{\sigma}^{\alpha} := R \sigma^{\alpha} R^* \tag{3.16}$$

and a new operator  $\widetilde{W}$  which is obtained by replacing the  $\sigma$  by  $\widetilde{\sigma}$  in (3.8). Then, the operators W and  $\widetilde{W}$  are related in the same way as the Pauli matrices  $\sigma$  and  $\widetilde{\sigma}$ , i.e.

$$W := RWR^*$$

If there exists a smooth matrix-function  $R: M \to SU(2)$  such that the corresponding Pauli matrices  $\sigma^{\alpha}$  and  $\tilde{\sigma}^{\alpha}$  are related in accordance with (3.16), we say that the operators W are  $\widetilde{W}$  are *equivalent*.

The charge conjugation operator also possesses some additional interesting properties, as stated by following

**Lemma 3.2.6.** The formulae (3.13) and (3.14) imply the following useful identities:

$$C(C(v)) = -v,$$
 (3.17)

$$\langle v, C(v) \rangle = 0, \tag{3.18}$$

$$\langle C(v), C(w) \rangle = \langle w, v \rangle.$$
 (3.19)

*Proof.* Let  $v = \begin{pmatrix} v_1 & v_2 \end{pmatrix}^T$  and  $w = \begin{pmatrix} w_1 & w_2 \end{pmatrix}^T$  be arbitrary smooth scalar functions and

$$C(v) = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} \overline{v_1} \\ \overline{v_2} \end{pmatrix} = \begin{pmatrix} -\overline{v_2} \\ \overline{v_1} \end{pmatrix}.$$

Then

$$C(C(v)) = C\begin{pmatrix} -\overline{v_2} \\ \overline{v_1} \end{pmatrix} = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} -v_2 \\ v_1 \end{pmatrix} = \begin{pmatrix} -v_1 \\ -v_2 \end{pmatrix} = -v.$$

Also

$$\langle v, C(v) \rangle = \int_M (C(v))^* v \sqrt{\det g_{\alpha\beta}} \, dx = \int_M \left( \begin{array}{cc} -v_2 & v_1 \end{array} \right) \left( \begin{array}{cc} v_1 \\ v_2 \end{array} \right) \sqrt{\det g_{\alpha\beta}} \, dx \\ = \int_M \left( -v_1 v_2 + v_1 v_2 \right) \sqrt{\det g_{\alpha\beta}} \, dx = 0,$$

and

$$\begin{aligned} \langle C(v), C(w) \rangle - \langle w, v \rangle &= \int_M (C(w))^* C(v) \sqrt{\det g_{\alpha\beta}} \, dx - \int_M (v)^* w \sqrt{\det g_{\alpha\beta}} \, dx \\ &= \int_M (w_2 \overline{v_2} + w_1 \overline{v_1} - \overline{v_1} w_1 - \overline{v_2} w_2) dx = 0. \end{aligned}$$

Hence,  $\langle C(v), C(w) \rangle = \langle w, v \rangle$ .

As was stated in [19], the justification for the introduction of the halfdensity  $\sqrt{\det g_{\alpha\beta}}$  in the formula (3.13) for the inner product lies in the fact that the massless Dirac operator (3.8) is an operator which acts as on 2columns of scalar functions, i.e. on 2-columns of quantities which do not change under changes of local coordinates. The density and the half-density are defined as follows, see [85].

**Definition 3.2.7.** We say that  $\mu$  is a *density* if  $\mu(x) = J(x)\widetilde{\mu}(\widetilde{x}(x))$  and a *half-density* if  $\mu(x) = J^{1/2}(x)\widetilde{\mu}(\widetilde{x}(x))$  where  $\widetilde{\mu}$  is the representation of  $\mu$  in coordinates  $\widetilde{x}$  and  $J(x) = |\det \partial \widetilde{x}/\partial x|$ .

An operator of special interest of us is the massless Dirac operator on half-densities:

**Definition 3.2.8.** The massless Dirac operator on half-densities is the operator

$$W_{1/2} := (\det g_{\kappa\lambda})^{1/4} W(\det g_{\mu\nu})^{-1/4}$$
(3.20)

which maps half-densities to half-densities.

**Remark 3.2.9.** The operator (3.20) is equivalent to the massless Dirac operator (3.8).

The domain of the operator  $W_{1/2}$  is  $H^1(M; \mathbb{C}^2)$ , which is the Sobolev space of 2-columns of half-densities that are square integrable together with their first partial derivatives.

## 3.3 The spectrum of the massless Dirac operator

We are now interested to apply the results from Section 3.1 for the first order differential operator to the spectral analysis of the massless Dirac operator (3.8).

We can now ask the question: which conditions need the first order differential operator A satisfy in order for it to be the massless Dirac operator on half-densities? The answer to that question is stated by following theorem.

**Theorem 3.3.1.** The operator A is a massless Dirac operator on half-densities if and only if the following two conditions are satisfied at every point of the manifold M:

- the subprincipal symbol (3.1) of the operator is proportional to the identity matrix;
- the second asymptotic coefficient of the spectral function b(x) is zero.

For the proofs of Theorem 3.1.7 and Theorem 3.3.1, see [19].

The explicit formula for the principal symbol of the massless Dirac operator on half-densities (3.20) is given by

$$A_{1}(x,\xi) = \begin{pmatrix} e_{3}^{\alpha} & e_{1}^{\alpha} - ie_{2}^{\alpha} \\ e_{1}^{\alpha} + ie_{2}^{\alpha} & -e_{3}^{\alpha} \end{pmatrix} \xi_{\alpha}$$
(3.21)

and the explicit formula for the its subprincipal symbol is given by

$$A_{\rm sub}(x) = \frac{3}{4} (*T^{\rm ax}(x))I, \qquad (3.22)$$

where

$$*T^{\mathrm{ax}}(x) = \frac{\delta_{lk}}{3} \sqrt{\det g^{\alpha\beta}} \left[ e^k_1 \partial e^l_3 / \partial x^2 + e^k_2 \partial e^l_1 / \partial x^3 + e^k_3 \partial e^l_2 / \partial x^1 - e^k_1 \partial e^l_2 / \partial x^3 - e^k_2 \partial e^l_3 / \partial x^1 - e^k_3 \partial e^l_1 / \partial x^2 \right]$$
(3.23)

is the explicit formula for the Hodge dual of the axial part of torsion, see Section 8 of [19].

The massless Dirac operator on half-densities (3.20) is a self-adjoint first order elliptic differential operator acting on 2-columns of complex-valued half-densities, det  $A_1(x,\xi) \neq 0$  and the principal symbol (3.21) is clearly trace free. Hence, the massless Dirac operator on half-densities satisfies the conditions for the differential operator A from Remark 3.1.3.

According to Theorem 3.1.7, the asymptotic formulae (3.4) and (3.5) for the massless Dirac operator on half-densities (3.20) read

$$e(\lambda, x, x) = \frac{\sqrt{\det g_{\alpha\beta}(x)}}{6\pi^2} \lambda^3 + o(\lambda^2), \qquad (3.24)$$

$$N(\lambda) = \frac{\text{Vol } M}{6\pi^2} \lambda^3 + o(\lambda^2), \qquad (3.25)$$

where Vol M is the volume of the Riemannian 3-manifold M.

**Remark 3.3.2.** We want to stress the remarkable simplicity of the asymptotic formula (3.25) for the counting function of the operator (3.20). The asymptotic coefficient (3.6) is determined only by the volume of the Riemannian 3-manifold M and does not depend on the shape of the manifold M. The asymptotic coefficient (3.7) is equal to zero.

**Remark 3.3.3.** Analysing the formula (3.25) we can conclude that the positive eigenvalues of the massless Dirac operator are distributed in same way as the negative eigenvalues.

**Remark 3.3.4.** The factor  $\sqrt{\det g_{\alpha\beta}(x)}$  appears in formula (3.24) because we are working with the massless Dirac operator on half-densities (3.20). Of course, for the massless Dirac operator on spinors (3.8) the spectral function is a scalar field and formula (3.24) reads

$$e(\lambda, x, x) = \frac{1}{6\pi^2}\lambda^3 + o(\lambda^2).$$

**Example 3.3.5.** Consider the unit torus  $\mathbb{T}^3$  and unit sphere  $\mathbb{S}^3$  as manifolds M. Since Vol  $\mathbb{T}^3 = (2\pi)^3$  and Vol  $\mathbb{S}^3 = 2\pi^2$  the counting functions (3.25) on a 3-torus and 3-sphere are respectively given by

$$N(\lambda) = \frac{4}{3}\pi\lambda^3 + o(\lambda^2), \qquad (3.26)$$

and

$$N(\lambda) = \frac{1}{3}\lambda^3 + o(\lambda^2). \tag{3.27}$$

The counting functions (3.26) and (3.27) show that for these two choices of manifolds the positive eigenvalues are distributed in same way as the negative eigenvalues.

The relation between the spectrum and the geometry of the manifold is an object of intense research today. It is a very difficult task to determine the spectrum of the massless Dirac operator on an arbitrary manifold M. To our knowledge, the first explicit calculation of the spectrum of the Dirac operator was done by Friedrich [35]. In the same paper, the dependence of the spectrum to the choice of the spin structure was also shown.

For the moment, we can only analyse the counting function (3.25) to see the distribution of the eigenvalues. There are only two examples where the spectrum is determined explicitly. The first example is the unit torus  $\mathbb{T}^3$ equipped with Euclidean metric and the second example is the unit sphere  $\mathbb{S}^3$ equipped with metric induced by the natural embedding of  $\mathbb{S}^3$  in Euclidean space  $\mathbb{R}^4$ . In both examples the spectrum is symmetric about zero, see [10, 19, 93].

To our knowledge, there are only two known examples where the spectrum of the massless Dirac operator can be calculated explicitly. The spectrum of the massless Dirac operator on the unit torus  $\mathbb{T}^3$  equipped with Euclidean metric is as follows: zero is an eigenvalue of multiplicity two and for each  $m \in \mathbb{Z}^3 \setminus \{0\}$  the eigenvalues are  $\pm ||m||$ . The spectrum of the massless Dirac operator on  $\mathbb{S}^3$  equipped with metric induced by the natural embedding of  $\mathbb{S}^3$  in Euclidean space  $\mathbb{R}^4$  can also be calculated explicitly, see [10, 93], and the eigenvalues are  $\pm (k + \frac{1}{2})$ ,  $(k = 1, 2, \ldots)$ , with multiplicity k(k + 1).

The eigenvalues of the massless Dirac operator have a very interesting property given by the following lemma, which is a consequence of the commuting of the massless Dirac operator and the charge conjugation operator.

**Lemma 3.3.6.** The eigenvalues of the massless Dirac operator have even multiplicity.

*Proof.* Let the vector v be the eigenvector corresponding to the eigenvalue  $\lambda$  of the massless Dirac operator, i.e.  $Wv = \lambda v$ . Let C be the charge conjugation operator. Then, according to the formula (3.15) we have that

$$W(C(v)) = C(Wv) = C(\lambda v) = \lambda C(v),$$

i.e. the vector C(v) is also the eigenvector of the massless Dirac operator corresponding the same eigenvalue  $\lambda$ .

**Remark 3.3.7.** As was stated in [24], if we introduce the magnetic field as Erdős and Solovej did in [29], the situation is different and the double eigenvalues split up. This indicates that the double eigenvalues of the massless Dirac operator correspond to two different spins.

An interesting result for the lower eigenvalue estimate on closed manifolds is given by the so-called *Friedrich's inequality*.

**Theorem 3.3.8.** If  $\lambda$  is an eigenvalue of the massless Dirac operator on an *n*-dimensional ( $n \ge 2$ ) closed Riemannian manifold ( $M^n, g$ ) with a spin structure, then

$$\lambda^2 \ge \frac{n}{4(n-1)} \inf_M(\mathcal{R}),\tag{3.28}$$

where  $\mathcal{R}$  is scalar curvature.

For the proof of Theorem 3.3.8 and the improvements of the formula (3.28) see [36].

According to [3, 4, 5, 6], for a general oriented Riemannian 3-manifold (M, g) there is no reason for the spectrum of the massless Dirac operator to be symmetric. In this chapter we break the spectral symmetry working on the unit torus  $\mathbb{T}^3$  as a 3-manifold with trivial topology perturbing the Euclidean metric. This approach was also used in [24].

## 3.4 The perturbation theory for the massless Dirac operator

One way to break the spectral symmetry of the massless Dirac operator is to use a trivial metric and a manifold with a nontrivial topology. In this chapter we use an opposite strategy: we use a manifold with a trivial topology and a nontrivial metric, i.e. we perturb the Euclidean metric in order to create the spectral asymmetry. In this section we develop the perturbation theory which is used for that purpose. We consider perturbations of the Euclidean metric with a small positive real parameter  $\epsilon$ . In order to have a simpler approach, we introduce a different approach to the frame and coframe because we are working in a specified coordinate system, in a similar fashion to [24]. For the Euclidean metric, it is easy to see that the massless Dirac operator (3.8) corresponding to the standard spin structure reads

$$W = -i \begin{pmatrix} \frac{\partial}{\partial x^3} & \frac{\partial}{\partial x^1} - i \frac{\partial}{\partial x^2} \\ \frac{\partial}{\partial x^1} + i \frac{\partial}{\partial x^2} & -\frac{\partial}{\partial x^3} \end{pmatrix}.$$
 (3.29)

Consider the perturbed metric  $g_{\alpha\beta}(x;\epsilon)$  whose components are smooth functions of coordinates  $x^{\alpha}$  and of small positive real parameter  $\epsilon$ , which satisfies

$$g_{\alpha\beta}(x;0) = \delta_{\alpha\beta}.\tag{3.30}$$

Let

$$h_{\alpha\beta}(x) := \left. \frac{\partial g_{\alpha\beta}(x)}{\partial \epsilon} \right|_{\epsilon=0}, \quad k_{\alpha\beta}(x) := 4 \left. \frac{\partial^2 g_{\alpha\beta}(x)}{\partial \epsilon^2} \right|_{\epsilon=0}.$$
(3.31)

A perturbed coframe is a smooth real-valued matrix-function  $e^{j}_{\alpha}(x;\epsilon)$ ,  $j, \alpha = 1, 2, 3$  satisfying the conditions

$$g_{\alpha\beta}(x;\epsilon) = \delta_{jk} e^{j}{}_{\alpha}(x;\epsilon) e^{k}{}_{\beta}(x;\epsilon), \qquad (3.32)$$

$$e^{j}_{\alpha}(x;0) = \delta^{j}_{\alpha}. \tag{3.33}$$

The reason for the condition (3.33) is that we want our unperturbed operator to have the form (3.29). The first index j of the matrix-function  $e^{j}{}_{\alpha}(x;\epsilon)$ enumerates the matrix rows and the second index  $\alpha$  enumerates the matrix columns. Since we are working in a specified coordinate system, we can require the coframe  $e^{j}{}_{\alpha}(x;\epsilon)$  to be symmetric, i.e.

$$e^{j}_{\alpha}(x;\epsilon) = e^{\alpha}_{\ i}(x;\epsilon). \tag{3.34}$$

The condition (3.34) will significantly simplify our calculations involving the massless Dirac operator. As was stated in [24], the asymptotic expansion of the subprincipal symbol of the massless Dirac operator on half-densities (3.22) in powers of  $\epsilon$  starts with a quadratic term and that coefficient has a very simple structure. For the perturbed metric  $g_{\alpha\beta}(x;\epsilon)$  the coframe  $e^{j}_{\alpha}(x;\epsilon)$  is not determined uniquely. We can multiply the matrix-function  $e^{j}_{\alpha}(x;\epsilon)$  from the left by an arbitrary smooth  $3 \times 3$  special orthogonal matrix-function  $O(x;\epsilon)$  satisfying the condition O(x;0) = I, with I denoting the  $3 \times 3$  identity matrix. The new coframe will also satisfy the conditions (3.32), (3.33). This choice of the coframe does not affect the spectrum of the massless Dirac operator, see [19] for more details.

The frame is a smooth real-valued matrix-function  $e_j^{\alpha}(x;\epsilon)$  defined by the system of linear algebraic equations

$$e_j^{\ \alpha}(x;\epsilon)e^k_{\ \alpha}(x;\epsilon) = \delta_j^{\ k}.$$
(3.35)

**Remark 3.4.1.** In matrix notation, formula (3.35) reads as the frame is the transpose of the inverse of the coframe.

As we chose our coframe to be symmetric, our frame is symmetric as well. According to formulae (3.30) and (3.31) we have that

$$g_{\alpha\beta}(x;\epsilon) = \delta_{\alpha\beta} + \epsilon h_{\alpha\beta}(x) + \frac{\epsilon^2}{4}k_{\alpha\beta}(x) + O(\epsilon^3).$$
(3.36)

Using Taylor's formula for the function  $\sqrt{1+z}$ , the coframe is then given by

$$e^{j}{}_{\alpha}(x;\epsilon) = \delta^{j}{}_{\alpha} + \frac{\epsilon}{2}h^{j}{}_{\alpha}(x) - \frac{\epsilon^{2}}{8}\left(h^{2}\right)^{j}{}_{\alpha}(x) + \frac{\epsilon^{2}}{8}k^{j}{}_{\alpha}(x) + O(\epsilon^{3}).$$
(3.37)

Using formula (3.37) and the Taylor's formula for  $(1+z)^{-1}$ , the frame is now given by

$$e_j^{\alpha}(x;\epsilon) = \delta_j^{\alpha} - \frac{\epsilon}{2}h_j^{\alpha}(x) + \frac{3\epsilon^2}{8}\left(h^2\right)_j^{\alpha}(x) - \frac{\epsilon^2}{8}k_j^{\alpha}(x) + O(\epsilon^3).$$
(3.38)

**Remark 3.4.2.** Note that  $h^2$  appearing in formulae (3.37) and (3.38) denotes the square of the perturbation matrix h, i.e.  $(h^2)_j^{\alpha}(x) = h_j^{\beta}(x)h_{\beta}^{\alpha}(x)$ , where the summation is over the repeated index  $\beta$ .

We raise and lower indices in the matrices h and k using the Euclidean metric, which means that raising or lowering an index does not change anything.

We cannot apply standard perturbation theory for the massless Dirac operator, because the massless Dirac operator commutes with the charge conjugation operator and the eigenvalues have even multiplicity, see Lemma 3.3.6. In this section we present the perturbation theory which takes care of multiplicity of the eigenvalues. In the presence of a magnetic field, we can apply standard perturbation theory because the magnetic field splits up the double eigenvalues, see Remark 3.3.7.

Let  $W_{1/2}(\epsilon)$  therefore be the massless Dirac operator on half-densities (3.20) corresponding to the perturbed metric  $g_{\alpha\beta}(x;\epsilon)$ . We prefer to deal with the massless Dirac operator on half-densities  $W_{1/2}(\epsilon)$  rather than the massless Dirac operator  $W(\epsilon)$ , although the spectra of these operators are the same, because  $W_{1/2}(\epsilon)$  and  $W(\epsilon)$  are equivalent. Let

$$W_{1/2}(\epsilon) = W_{1/2}^{(0)} + \epsilon W_{1/2}^{(1)} + \epsilon^2 W_{1/2}^{(2)} + \cdots$$
(3.39)

be the asymptotic expansion of the perturbed massless Dirac operator in powers of the small parameter  $\epsilon$ . The operator  $W_{1/2}^{(0)} = W_{1/2}(0)$  is the unperturbed massless Dirac operator on half-densities (3.20). We denote by  $\lambda^{(0)}$  the eigenvalue of this operator and by  $v^{(0)}$  the corresponding eigenvector. Each eigenvalue  $\lambda^{(0)}$  has even multiplicity because the massless Dirac operator commutes with the antilinear operator of charge conjugation (3.14).

The perturbation of an isolated eigenvalue of finite multiplicity of a bounded operator was described by Rellich [84] and that procedure can be applied in our case with some additional conditions.

applied in our case with some additional conditions. The operators  $W_{1/2}^{(k)}$ , k = 0, 1, 2, ... are formally self-adjoint first order differential operators which also commute with the antilinear operator of charge conjugation (3.14). We suppose that the series (3.39) is convergent for a sufficiently small  $\epsilon$ . Then, see [84], there exist power series

$$\lambda(\epsilon) = \lambda^{(0)} + \epsilon \lambda^{(1)} + \epsilon^2 \lambda^{(2)} + \cdots, \qquad (3.40)$$

$$v(\epsilon) = v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \cdots .$$
(3.41)

which are convergent for sufficiently small  $\epsilon$ , which satisfy the condition

$$W_{1/2}(\epsilon)v(\epsilon) = \lambda(\epsilon)v(\epsilon).$$

In the perturbation process that we use in this chapter we deal with a formally self-adjoint linear operator A which commutes with the antilinear operator of charge conjugation (3.14). Such an operator possess a special property stated by following lemma, see [24].

**Lemma 3.4.3.** Let  $A : C^{\infty}(M; \mathbb{C}^2) \to C^{\infty}(M; \mathbb{C}^2)$  be a (possibly unbounded) formally self-adjoint linear operator which commutes with the antilinear operator of charge conjugation (3.14). Then for any  $v \in C^{\infty}(M; \mathbb{C}^2)$  we have

$$\langle Av, C(v) \rangle = 0. \tag{3.42}$$

*Proof.* We prove of this lemma using the properties (3.17)-(3.19) of the charge conjugation operator (3.14). Let  $v, w \in C^{\infty}(M; \mathbb{C}^2)$  be arbitrary elements. Then we have

$$\langle A(C(w)), C(v) \rangle = \langle C(A(w)), C(v) \rangle = \langle v, A(w) \rangle = \langle A(v), w \rangle.$$
(3.43)

For the w = C(v) formula (3.43) reads

$$\langle A(C(C(v))), C(v) \rangle = \langle A(v), C(v) \rangle.$$
(3.44)
Since we have C(C(v)) = -v, the formula (3.44) becomes

$$-\langle A(v), C(v) \rangle = \langle A(v), C(v) \rangle$$

which gives us (3.42).

We will use the *pseudoinverse* operator which we will define similarly to how it was done in [24, 84].

### 3.4.1 Pseudoinverse operator construction

Let  $v^{(0)}$  be a normalised eigenvector of the operator A corresponding to the eigenvalue  $\lambda^{(0)}$ . Then, see Lemma 3.3.6, the vector  $C(v^{(0)})$  is also a normalised eigenvector corresponding to the eigenvalue  $\lambda^{(0)}$ . Consider the problem

$$(A - \lambda^{(0)}) v = f, (3.45)$$

for a given function  $f \in L^2(M; \mathbb{C}^2)$  where we need to find the function  $v \in H^1(M; \mathbb{C}^2)$ . Suppose that the function f satisfies the conditions

$$\langle f, v^{(0)} \rangle = \langle f, C(v^{(0)}) \rangle = 0,$$

where C is the charge conjugation operator (3.14). The problem (3.45) can be resolved for the function v and the uniqueness of this function is achieved by the conditions

$$\langle v, v^{(0)} \rangle = \langle v, C(v^{(0)}) \rangle = 0.$$

We define the operator Q as

 $Q: f \mapsto v.$ 

The operator Q is a bounded linear operator acting on the orthogonal complement of the eigenspace of the operator A corresponding to the eigenvalue  $\lambda^{(0)}$ . Also, the bounded linear operator Q is self-adjoint and commutes with the antilinear operator of charge conjugation (3.14). We can extend the acting of the pseudoinverse operator Q to the whole Hilbert space  $L^2(M; \mathbb{C}^2)$  in accordance with  $Qv^{(0)} = QC(v^{(0)}) = 0$ .

Now we construct the pseudoinverse operator Q itself, as was explained in [84]. Let  $\lambda$  be an eigenvalue of multiplicity k of a Hermitian operator A. Then the homogenous equation

$$(A - \lambda^{(0)})v = 0$$

has k linearly independent solutions  $\phi^{(1)}, \phi^{(2)}, \ldots, \phi^{(k)}$  for which we can assume orthonormality, i.e.

$$\langle \phi^{(i)}, \phi^{(j)} \rangle = \delta_{ij}, \ (i, j = 1, 2, \dots, k).$$

The operator  $A - \lambda^{(0)}$  has no inverse, but there is a unique bounded Hermitian operator Q such that  $Q\phi^{(i)} = 0$ , (i = 1, 2, ..., k) and

$$Q(A - \lambda^{(0)})u = u - \sum_{i=1}^{k} \langle \phi^{(i)}, u \rangle \phi^{(i)}.$$

Define the projector operator P by

$$Pu = \sum_{i=1}^{k} \langle \phi^{(i)}, u \rangle \phi^{(i)}.$$
 (3.46)

Then the properties of the above operator Q can be written as QP = 0 and  $Q(A - \lambda^{(0)}) = I - P$ .

The operator Q is called the pseudoinverse operator of the operator  $A - \lambda$ .

Now, we can complete the eigenvectors  $\phi^{(1)}, \phi^{(2)}, \ldots, \phi^{(k)}$  with eigenvectors  $\phi^{(k+1)}, \phi^{(k+2)}, \ldots, \phi^{(n)}$ , which correspond to eigenvalues  $\lambda_{k+1}, \lambda_{k+2}, \ldots, \lambda_n$ , respectively, such that

$$\langle \phi^{(i)}, \phi^{(j)} \rangle = \delta_{ij}, \ (i, j = 1, 2, \dots, n).$$

Expanding v and f as

$$v = \sum_{i=1}^{n} \langle \phi^{(i)}, v \rangle \phi^{(i)}, \quad f = \sum_{i=1}^{n} \langle \phi^{(i)}, f \rangle \phi^{(i)},$$

from equation (3.45) we get that

$$\langle \phi^{(i)}, f \rangle = 0, \quad (i = 1, 2, \dots, k), \langle \phi^{(i)}, f \rangle = (\lambda_i - \lambda) \langle \phi^{(i)}, v \rangle, \quad (i = k + 1, k + 2, \dots, n).$$

If the equation for v has a solution, it is necessary for f to be orthogonal to all solutions of the homogeneous equation. Hence we set

$$v = \sum_{i=1}^{k} v_i \phi^{(i)} + \sum_{\lambda_i \neq \lambda} \frac{\langle \phi^{(i)}, f \rangle}{\lambda_i - \lambda} \phi^{(i)},$$

where  $v_i$  are arbitrary constants. The vector v defines the complete solution of the equation.

Let the operator P be the projector operator into the space spanned by the vectors  $\phi^{(1)}, \phi^{(2)}, \ldots, \phi^{(k)}$ , and  $P_i$  the projector into the one-dimensional space spanned by  $\phi^{(i)}$ ,  $(i = k + 1, k + 2, \ldots, n)$ .

Definition 3.4.4. The operator

$$Q = \sum_{\lambda_i \neq \lambda^{(0)}} \frac{P_i}{\lambda_i - \lambda^{(0)}}$$
(3.47)

is the *pseudoinverse operator* of the operator  $A - \lambda^{(0)}$ .

### 3.4.2 Explicit formulae for the asymptotic coefficients

Now we will derive the explicit formulae for the coefficients  $\lambda^{(1)}$  and  $\lambda^{(2)}$  in the asymptotic expansion (3.40). Consider the perturbed eigenvalue problem

$$W_{1/2}(\epsilon)v(\epsilon) = \lambda(\epsilon)v(\epsilon)$$

Using formulae (3.39), (3.40) and (3.41), we get that

$$(W_{1/2}^{(0)} + \epsilon W_{1/2}^{(1)} + \epsilon^2 W_{1/2}^{(2)} + \cdots)(v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \cdots)$$
  
=  $(\lambda^{(0)} + \epsilon \lambda^{(1)} + \epsilon^2 \lambda^{(2)} + \cdots)(v^{(0)} + \epsilon v^{(1)} + \epsilon^2 v^{(2)} + \cdots).$ 

By grouping together the elements not containing  $\epsilon$ , we get that

$$W_{1/2}^{(0)}v^{(0)} = \lambda^{(0)}v^{(0)},$$

which is the unperturbed eigenvalue problem. By grouping together the elements containing  $\epsilon$ , we get that

$$W_{1/2}^{(0)}v^{(1)} + W_{1/2}^{(1)}v^{(0)} = \lambda^{(0)}v^{(1)} + \lambda^{(1)}v^{(0)}$$

and hence  $(W_{1/2}^{(0)} - \lambda^{(0)})v^{(1)} = (\lambda^{(1)} - W_{1/2}^{(1)})v^{(0)}$ , i.e.  $v^{(1)} = Q(\lambda^{(1)} - W_{1/2}^{(1)})v^{(0)}$ where Q is the pseudoinverse operator of the operator  $W_{1/2}^{(0)} - \lambda^{(0)}$ . We denote by

$$f^{(1)} = (\lambda^{(1)} - W^{(1)}_{1/2})v^{(0)}.$$
(3.48)

By grouping together the elements containing  $\epsilon^2$ , we get that

$$W_{1/2}^{(0)}v^{(2)} + W_{1/2}^{(1)}v^{(1)} + W_{1/2}^{(2)}v^{(0)} = \lambda^{(0)}v^{(2)} + \lambda^{(1)}v^{(1)} + \lambda^{(2)}v^{(0)}$$

and hence

$$(W_{1/2}^{(0)} - \lambda^{(0)})v^{(2)} = (\lambda^{(2)} - W_{1/2}^{(2)})v^{(0)} + (\lambda^{(1)} - W_{1/2}^{(1)})v^{(1)}.$$

We denote by

$$f^{(2)} = (\lambda^{(2)} - W^{(2)}_{1/2})v^{(0)} + (\lambda^{(1)} - W^{(1)}_{1/2})v^{(1)}$$
  
=  $(\lambda^{(2)} - W^{(2)}_{1/2})v^{(0)} + (\lambda^{(1)} - W^{(1)}_{1/2})Q(\lambda^{(1)} - W^{(1)}_{1/2})v^{(0)}.$  (3.49)

Continuing this process, the vectors  $f^{(k)}$  and the coefficients  $\lambda^{(k)}$  are obtained from the conditions

$$\langle f^{(k)}, v^{(0)} \rangle = 0$$
 (3.50)

and

$$\langle f^{(k)}, C(v^{(0)}) \rangle = 0.$$
 (3.51)

**Remark 3.4.5.** The eigenvalues have even multiplicity, so the condition (3.51) is an additional condition which need to be satisfied. This is the part where our perturbation process differs from the standard perturbation process for single eigenvalues.

The components  $v^{(k)}$  are given by

$$v^{(k)} = Qf^{(k)},$$

where Q is the pseudoinverse operator of the operator  $W_{1/2}^{(0)} - \lambda^{(0)}$ . Substituting formulae (3.48), (3.49) into formulae (3.50), (3.51) we obtain the result stated by the following

**Lemma 3.4.6.** The explicit formulae for the coefficients  $\lambda^{(1)}$  and  $\lambda^{(2)}$  in the asymptotic expansion of the eigenvalue  $\lambda(\epsilon)$  are given by

$$\lambda^{(1)} = \langle W_{1/2}^{(1)} v^{(0)}, v^{(0)} \rangle, \qquad (3.52)$$

$$\lambda^{(2)} = \langle W_{1/2}^{(2)} v^{(0)}, v^{(0)} \rangle - \langle (W_{1/2}^{(1)} - \lambda^{(1)}) Q(W_{1/2}^{(1)} - \lambda^{(1)}) v^{(0)}, v^{(0)} \rangle.$$
(3.53)

where  $\langle \cdot, \cdot \rangle$  is the inner product defined by (3.13).

## **3.5** Spectral asymmetry of the massless Dirac operator on a 3-torus

We now want to apply the results and the approach from the previous sections in this chapter in order to obtain and demonstrate spectral asymmetry of the massless Dirac operator (3.8) on a particular manifold. Hence, consider the unit tours  $\mathbb{T}^3$  parameterised by cyclic coordinates  $x^{\alpha}$ ,  $\alpha = 1, 2, 3$  of period  $2\pi$ . For the Euclidean metric, the massless Dirac operator corresponding to the standard spin structure is given by the formula (3.29) and the eigenvalues can be evaluated explicitly. The spectrum is symmetric about zero and zero itself is an eigenvalue of the operator. But, as we stated before and according to [3, 4, 5, 6], for a general oriented Riemannian 3-manifold (M, g) there is no reason for the spectrum of the massless Dirac operator (3.8) to be symmetric.

We obtain the spectral asymmetry of the massless Dirac operator on a unit torus  $\mathbb{T}^3$  by perturbing the Euclidean metric, as described in Section 3.4. By perturbing the metric we get the perturbed coframe (3.37) and frame (3.38), and hence the massless Dirac operator on half-densities (3.20) is also perturbed accordingly and we investigate under which perturbations of the metric the spectral symmetry of this operator is broken. **Definition 3.5.1.** For a given function  $f : \mathbb{T}^3 \to \mathbb{C}$ , we denote by

$$\widehat{f}(m) := \frac{1}{(2\pi)^3} \int_{\mathbb{T}^3} e^{-im_{\alpha}x^{\alpha}} f(x) dx, \quad m \in \mathbb{Z}^3,$$
(3.54)

its Fourier coefficients. Here  $dx := dx^1 dx^2 dx^3$ .

**Remark 3.5.2.** An important special case that we will consider is when the metric is a function of the coordinate  $x^1$  only. In this case one can choose the coframe and frame so that they depend on the coordinate  $x^1$  only and seek eigenfunctions in the form  $v(x^1)$ . We call this case the *axisymmetric case*. In the above setting the original eigenvalue problem for a partial differential operator reduces to an eigenvalue problem for an ordinary differential operator.

**Remark 3.5.3.** For the Euclidean metric the massless Dirac operator (3.29) corresponding to the standard spin structure in the axisymmetric case reads

$$W = -i \begin{pmatrix} 0 & \frac{d}{dx^1} \\ \frac{d}{dx^1} & 0 \end{pmatrix}.$$
 (3.55)

A first order differential operator A is completely determined by its principal and subprincipal symbols, see Definition 3.1.1 and Definition 3.1.2. The principal symbol has the form  $A_1(x,\xi) := M^{(\alpha)}(x)\xi_{\alpha}$  where  $M^{(\alpha)}(x)$  are matrix-functions depending only on the position variable x and the differential operator A is given by

$$A = -\frac{\mathrm{i}}{2}M^{(\alpha)}(x)\frac{\partial}{\partial x^{\alpha}} - \frac{\mathrm{i}}{2}\frac{\partial}{\partial x^{\alpha}}M^{(\alpha)}(x) + A_{\mathrm{sub}}.$$
 (3.56)

Let us now consider the axisymmetric case, see Remark 3.5.2. For  $\alpha = 1$  in (3.56), using the formulae for the principal symbol (3.21) and subprincipal symbol (3.22), we straightforwardly get the explicit formula for the axisymmetric massless Dirac operator on half-densities in terms of the coframe and the frame

$$W_{1/2}(\epsilon) = -\frac{i}{2} \begin{pmatrix} e_3^1 & e_1^{1} - ie_2^1 \\ e_1^{1} + ie_2^{1} & -e_3^1 \end{pmatrix} \frac{d}{dx^1} \\ -\frac{i}{2} \frac{d}{dx^1} \begin{pmatrix} e_3^1 & e_1^{1} - ie_2^1 \\ e_1^{1} + ie_2^{1} & -e_3^1 \end{pmatrix} \\ + \frac{\delta_{jk}}{4\sqrt{\det g_{\alpha\beta}}} \left( e_3^j \left( \frac{de_2^k}{dx^1} \right) - e_2^j \left( \frac{de_3^k}{dx^1} \right) \right) I, \quad (3.57)$$

corresponding to the perturbed metric  $g(x^1, \epsilon)$  and where I is the  $2 \times 2$  identity matrix and

$$\sqrt{\det g_{\alpha\beta}} = \frac{1}{\sqrt{\det g^{\alpha\beta}}} = \det e^j{}_{\alpha} = \frac{1}{\det e_j{}^{\alpha}}.$$

The coframe  $e_{\alpha}^{j}(x^{1};\epsilon)$  and the frame  $e_{j}^{\alpha}(x^{1};\epsilon)$  are defined in accordance with (3.37) and (3.38), respectively.

For an eigenvalue  $n \in \mathbb{Z}$  the corresponding normalised eigenvector of the massless Dirac operator in the axisymmetric case (3.55) is

$$v_n(x^1) = \frac{1}{2\sqrt{\pi}} \begin{pmatrix} 1\\1 \end{pmatrix} e^{inx^1}.$$
(3.58)

Note that the vector

$$w_n(x^1) = C(v_n(x^1)) = \frac{1}{2\sqrt{\pi}} \begin{pmatrix} -1 \\ 1 \end{pmatrix} e^{-inx^1}$$
(3.59)

is also a normalised eigenvector of the operator (3.55) corresponding to eigenvalue  $n \in \mathbb{Z}$ , where C is the charge conjugation operator (3.14), since each eigenvalue of the massless Dirac operator in the axisymmetric case is of multiplicity two, see Lemma 3.3.6.

For the eigenvalue  $\lambda = 0$  of the massless Dirac operator on half-densities the asymptotic formula (3.40) was obtained in the general case by Downes, Levitin and Vassiliev in [24], where the perturbations of the metric (3.36) with  $k_{\alpha\beta}(x) = 0$  were considered. The linear coefficient  $\lambda^{(1)}$  is equal to zero and the explicit formula for the quadratic coefficient  $\lambda^{(2)}$  was obtained. The formula for the quadratic coefficient  $\lambda^{(2)}$  was expressed via the components of perturbation matrix  $h_{\alpha\beta}(x)$ :

$$\lambda^{(2)} = \frac{\mathrm{i}}{16} \varepsilon_{\alpha\beta\gamma} \sum_{m \in \mathbb{Z}^3 \setminus \{0\}} \left( \delta_{\mu\nu} - \frac{m_{\mu}m_{\nu}}{\|m\|^2} \right) m_{\alpha} \widehat{h}_{\beta\mu}(m) \overline{\widehat{h}_{\gamma\nu}(m)}.$$
(3.60)

Here  $\varepsilon_{\alpha\beta\gamma}$  is the totally antisymmetric quantity,  $\varepsilon_{123} := +1$ , and the overline stands for complex conjugation.

In the axisymmetric case, see Remark 3.5.2, formula (3.60) is simpler and reads

$$\lambda^{(2)} = -\frac{1}{8} \sum_{m_1 \in \mathbb{N}} m_1 \operatorname{tr} \left( \left( \begin{array}{cc} \widehat{h}_{22} & \widehat{h}_{23} \\ \widehat{h}_{32} & \widehat{h}_{33} \end{array} \right) \left( \begin{array}{cc} 0 & -\mathrm{i} \\ \mathrm{i} & 0 \end{array} \right) \left( \begin{array}{cc} \widehat{h}_{22} & \widehat{h}_{23} \\ \widehat{h}_{32} & \widehat{h}_{33} \end{array} \right)^* \right),$$

where  $\hat{h}_{\alpha\beta} = \hat{h}_{\alpha\beta}(m_1)$  and '\*' stands for Hermitian conjugation.

In [24] the authors also determined the conditions under which the constant  $\lambda^{(2)}$  is nonzero, which tells us that for a sufficiently small nonzero  $\epsilon$ the spectrum of our massless Dirac operator is asymmetric about zero. Also, the authors give two explicit examples of perturbation of the Euclidean metric, for which the eigenvalues of massless Dirac operator on half-densities in the axisymmetric case have been evaluated explicitly. One is an example of quadratic dependence and the second is an example of quartic dependence on the parameter  $\epsilon$ .

In this chapter we take a similar approach to that taken in [24], but crucially we consider the eigenvalues  $\pm 1$  of the massless Dirac operator in the axisymmetric case (3.57) and for the perturbed Euclidean metric (3.36) we derive their asymptotic formulae of type (3.40), i.e. we seek the perturbations of the Euclidean metric for which it is possible to shift the eigenvalues  $\pm 1$  in an asymmetric way.

#### 3.5.1 Numerical analysis of the spectrum

In this section we first numerically analyse the spectrum of the massless Dirac operator (3.55) using the *Galerkin method* to discretise the eigenvalue problem of the massless Dirac operator. We work on the unit torus  $\mathbb{T}^3$  equipped with the Euclidean metric. Then for the standard spin structure we can calculate the spectrum of the massless Dirac operator (3.55) explicitly.

Consider the 2m + 1 eigenvalues  $\lambda_i = i$ ,  $(i = 0, \pm 1, \ldots, \pm m)$  of the unperturbed massless Dirac operator on half-densities  $W_{1/2}(0)$ . Each eigenvalue  $\lambda_i$   $(i = 0, \pm 1, \ldots, \pm m)$  has multiplicity two and the corresponding eigenvectors  $v_i(x^1)$  and  $w_i(x^1)$   $(i = 0, \pm 1, \ldots, \pm m)$  are given by (3.58) and (3.59). Now, we have that

$$W_{1/2}(0)v_i(x^1) = \lambda_i v_i(x^1), \qquad (3.61)$$

$$W_{1/2}(0)w_i(x^1) = \lambda_i w_i(x^1), \qquad (3.62)$$

where  $i = 0, \pm 1, \ldots, \pm m$ . The eigenvectors  $v_i(x^1)$  and  $w_i(x^1)$  are orthonormal with respect to the inner product (3.13), i.e.

$$\langle v_i, v_j \rangle = \langle w_i, w_j \rangle = \delta_{ij},$$

$$\langle v_i, w_j \rangle = \langle w_i, v_j \rangle = 0, \ (i, j = 0, \pm 1, \dots, \pm m).$$

$$(3.63)$$

According to (3.63), from equations (3.61) and (3.62), for  $i, j = 0, \pm 1, \ldots, \pm m$ we have that

$$\lambda_i = \langle W_{1/2}(0)v_i(x^1), v_i(x^1) \rangle = \langle W_{1/2}(0)w_i(x^1), w_i(x^1) \rangle$$

and

$$\langle W_{1/2}(0)v_i(x^1), w_j(x^1) \rangle = \langle W_{1/2}(0)v_j(x^1), w_i(x^1) \rangle = 0, \langle W_{1/2}(0)w_i(x^1), v_j(x^1) \rangle = \langle W_{1/2}(0)w_j(x^1), v_i(x^1) \rangle = 0.$$

Let us now construct the matrices

$$H_{i,j} := \begin{pmatrix} \langle W_{1/2}(0)v_i, v_j \rangle & \langle W_{1/2}(0)v_i, w_j \rangle \\ \langle W_{1/2}(0)w_i, v_j \rangle & \langle W_{1/2}(0)w_i, w_j \rangle \end{pmatrix}$$
(3.64)

where  $i, j = 0, \pm 1, ..., \pm m$ . Using the matrices (3.64) let us construct the block matrix H as follows

$$H := \begin{pmatrix} H_{-m,m} & H_{0,m} & H_{m,m} \\ & \ddots & \vdots & \ddots & \\ & & H_{0,1} & & \\ & & & H_{0,0} & H_{1,0} & \cdots & \\ & & & & H_{0,-1} & & \\ & & & & \vdots & \ddots & \\ H_{-m,-m} & & H_{0,-m} & & H_{m,-m} \end{pmatrix}.$$

The matrix H is a quadratic matrix of order 2(2m + 1) and by construction it is a Hermitian matrix.

The eigenvalues of the matrix H are  $\lambda = 0, \pm 1, \ldots, \pm m$  and each eigenvalue has multiplicity two. It means that we reduced the eigenvalue problem (3.61)-(3.62) to the eigenvalue problem of the matrix H, using the Galerkin method.

Now we can consider the matrix  $H(\epsilon)$  with the perturbed massless Dirac operator  $W_{1/2}(\epsilon)$  instead of the unperturbed operator  $W_{1/2}(0)$ . Then the matrix  $H(\epsilon)$  is a Hermitian matrix whose entries depend on the parameter  $\epsilon$ and H(0) = H. Using the perturbation process described by McCartin [61] for perturbed Hermitian matrices, we can get the asymptotic expansions of the eigenvalues of the perturbed matrix  $H(\epsilon)$ , and specially the asymptotic expansions of the eigenvalues  $\lambda = \pm 1$ . The eigenvalues of the matrix  $H(\epsilon)$ will clearly converge to the eigenvalues of the matrix H as  $\epsilon \to 0$ .

**Remark 3.5.4.** Throughout this chapter we denote by  $\lambda_+(\epsilon)$  and  $\lambda_-(\epsilon)$  the asymptotic expansions of the eigenvalues  $\lambda = 1$  and  $\lambda = -1$ , respectively, and with  $\lambda_+^{(i)}$  and  $\lambda_-^{(i)}$  their asymptotic coefficients.

The examination of the spectrum of the perturbed massless Dirac operator will reduce to the examination of the spectrum of the Hermitian matrix  $H(\epsilon)$ . **Example 3.5.5.** Consider the coframe

$$e^{j}{}_{\alpha} = \delta^{j}{}_{\alpha} + \epsilon \begin{pmatrix} 0 & 0 & 0 \\ 0 & \cos x^{1} & \sin x^{1} \\ 0 & \sin x^{1} & -\cos x^{1} \end{pmatrix}.$$
 (3.65)

The explicit formula for the perturbed massless Dirac operator corresponding to the coframe (3.65) was calculated in [24] and it reads

$$W(\epsilon) = -i \begin{pmatrix} 0 & 1\\ 1 & 0 \end{pmatrix} \frac{d}{dx^1} - \frac{\epsilon^2}{2(1-\epsilon^2)}I.$$
(3.66)

The eigenvalues of the operator (3.66) are explicitly given by

$$\lambda_n(\epsilon) = n - \frac{\epsilon^2}{2(1-\epsilon^2)} = n - \frac{\epsilon^2}{2} - \frac{\epsilon^4}{2} + O(\epsilon^6), \quad n \in \mathbb{Z}$$
(3.67)

and all eigenvalues have multiplicity two.

Now we will use the coframe (3.65) to analyse the spectrum of the massless Dirac operator using the Galerkin method described at the beginning of this section in order to numerically confirm these results. We explicitly constructed the matrix  $H(\epsilon)$  of order  $102 \times 102$  and numerically analysed the part of its spectrum. The eigenvalues  $0, \pm 1, \pm 2$  of the matrix  $H(\epsilon)$  are perturbed as follows

	$\lambda = -2$	$\lambda = -1$	$\lambda = 0$	$\lambda = 1$	$\lambda = 2$
for $\epsilon = 0.2$	-2.02083	-1.02083	-0.0208333	0.979167	1.97917
for $\epsilon = 0.1$	-2.00505	-1.00505	-0.00505051	0.994949	1.994950
for $\epsilon = 0.01$	-2.00005	-1.00005	-0.000050005	0.99995	1.99995

and each eigenvalue has multiplicity two. Analysing the data given in the above table we see that for this choice of the coframe the spectral symmetry of the matrix  $H(\epsilon)$  is broken and consequently we obtain spectral asymmetry of the massless Dirac operator in the axisymmetric case.

Using the perturbation process for the matrices with double eigenvalues described in [61], we get that the asymptotic formulae for the eigenvalues  $\pm 1$  are given by

$$\lambda_{+}(\epsilon) = 1 - \frac{1}{2}\epsilon^{2} + O(\epsilon^{3}),$$
  
$$\lambda_{-}(\epsilon) = -1 - \frac{1}{2}\epsilon^{2} + O(\epsilon^{3}).$$

which is in accordance with formula (3.67).

Now, we will consider the coframe which is not symmetric to show that in this case it is also possible to obtain spectral asymmetry.

**Example 3.5.6.** Consider the coframe

$$e^{j}_{\ \alpha} = \delta^{j}_{\ \alpha} + \epsilon \left( \begin{array}{ccc} 0 & \cos x^{1} - \cos 2x^{1} + \cos 3x^{1} & \sin x^{1} + \sin 2x^{1} - \sin 3x^{1} \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{array} \right).$$

Analysing the spectrum of the matrix  $H(\epsilon)$  of order  $102 \times 102$  we get that the eigenvalues  $0, \pm 1, \pm 2$  of the matrix are perturbed as follows

	$\lambda = -2$	$\lambda = -1$	$\lambda = 0$	$\lambda = 1$	$\lambda = 2$
for $\epsilon = 0.2$	-2.10913	-1.05372	0.00169489	1.0571	2.11252
for $\epsilon = 0.1$	-2.02923	-1.01456	0.000119453	1.0148	2.02947
for $\epsilon = 0.01$	-2.0003	-1.00015	$1.24941 \times 10^{-8}$	1.00015	2.0003

and each eigenvalue has multiplicity two. Analysing the data given in the above table we see that the spectral symmetry is broken.

Using the method described in [61], we obtain that the asymptotic formulae for the eigenvalues  $\pm 1$  are given by

$$\lambda_{+}(\epsilon) = 1 + \frac{3}{2}\epsilon^{2} - \frac{17}{8}\epsilon^{4} + O(\epsilon^{5}),$$
  
$$\lambda_{-}(\epsilon) = -1 - \frac{3}{2}\epsilon^{2} + \frac{37}{8}\epsilon^{4} + O(\epsilon^{5}),$$

and we see that spectral asymmetry is achieved in the quartic term.

### 3.5.2 Analytical formulae for perturbed eigenvalues

The numerical analysis of the spectrum of massless Dirac operator shows that for the appropriate choice of the coframe spectral asymmetry can be obtained. Now we want to prove this statement analytically, hence we derive explicitly the asymptotic formulae (3.40) for the eigenvalues  $\pm 1$  for an arbitrary perturbation of the Euclidean metric. The main result of this section is following

**Theorem 3.5.7.** Under arbitrary perturbations of the metric (3.36), we have that

$$\lambda_{+}(\epsilon) = 1 + \lambda_{+}^{(1)}\epsilon + \lambda_{+}^{(2)}\epsilon^{2} + O(\epsilon^{3}) \quad as \quad \epsilon \to 0,$$
(3.68)

$$\lambda_{-}(\epsilon) = -1 + \lambda_{-}^{(1)}\epsilon + \lambda_{-}^{(2)}\epsilon^{2} + O(\epsilon^{3}) \quad as \quad \epsilon \to 0,$$
(3.69)

where the constants  $\lambda_+^{(1)}$ ,  $\lambda_-^{(1)}$ ,  $\lambda_+^{(2)}$  and  $\lambda_-^{(2)}$  are given by the formulae

$$\lambda_{+}^{(1)} = -\frac{1}{2}\hat{h}_{11}(0), \qquad (3.70)$$

$$\lambda_{-}^{(1)} = \frac{1}{2}\hat{h}_{11}(0), \qquad (3.71)$$

$$\lambda_{+}^{(2)} = \frac{3}{8} \widehat{(h^2)}_{11}(0) - \frac{1}{8} \widehat{k}_{11}(0) - \frac{i}{16} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \overline{\widehat{h}_{\alpha\beta}(m)} \widehat{h}_{\alpha\gamma}(m) - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} (m+1)^2 \widehat{h}_{11}(m-1) \overline{\widehat{h}_{11}(m-1)} - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \widehat{h}_{31}(m+1) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right) - \frac{i}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \widehat{h}_{21}(m+1) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right), \quad (3.72)$$

$$\lambda_{-}^{(2)} = -\frac{3}{8} \widehat{(h^2)}_{11}(0) + \frac{1}{8} \widehat{k}_{11}(0) - \frac{i}{16} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \overline{\widehat{h}_{\alpha\beta}(m)} \widehat{h}_{\alpha\gamma}(m) - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} \frac{1}{m+1} (m-1)^2 \widehat{h}_{11}(m+1) \overline{\widehat{h}_{11}(m+1)} - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \widehat{h}_{31}(m-1) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right) - \frac{i}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \widehat{h}_{21}(m-1) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right), \quad (3.73)$$

where  $\varepsilon_{\alpha\beta\gamma}$  is the totally antisymmetric quantity,  $\varepsilon_{123} := +1$ , and the overline stands for complex conjugation.

**Remark 3.5.8.** We consider the massless Dirac operator on half-densities in the axisymmetric case (3.57), hence the metric  $g_{\alpha\beta}$  depends on the coordinate  $x^1$  only and for a given function  $h_{ij}$ , its Fourier coefficients appearing in Theorem 3.5.7 are given by

$$\widehat{h_{ij}}(m_1) := \frac{1}{2\pi} \int_0^{2\pi} e^{-im_1 x^1} h_{ij}(x^1) dx^1, \quad m_1 \in \mathbb{Z}.$$
 (3.74)

Theorem 3.5.7 warrants the following facts.

**Remark 3.5.9.** According to the formulae (3.70) and (3.71), we have that

$$\lambda_{+}^{(1)} + \lambda_{-}^{(1)} = 0,$$

so we conclude that the spectral asymmetry can't be achieved in the linear term.

**Remark 3.5.10.** The length of the arc of the  $x^1$  circle is given by

$$l(\epsilon) = \int_0^{2\pi} \sqrt{g_{11}(x^1)} dx^1 = 2\pi \left(1 + \frac{1}{2}\hat{h}_{11}(0)\epsilon\right) + O(\epsilon^2).$$

Hence the coefficients  $\lambda_{+}^{(1)}$  and  $\lambda_{-}^{(1)}$  are determined by the change of the length of the arc of  $x^1$  circle.

Remark 3.5.11. According to formulae (3.72) and (3.73), we get that

$$\begin{split} \lambda_{+}^{(2)} + \lambda_{-}^{(2)} &= -\frac{i}{8} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \widehat{h}_{\alpha\beta}(m) \widehat{h}_{\alpha\gamma}(m) \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} (m+1)^2 \, \widehat{h}_{11}(m-1) \overline{\widehat{h}_{11}(m-1)} \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \widehat{h}_{31}(m+1) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right) \\ &- \frac{i}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \widehat{h}_{21}(m+1) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right) \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} \frac{1}{m+1} (m-1)^2 \, \widehat{h}_{11}(m+1) \overline{\widehat{h}_{11}(m+1)} \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \widehat{h}_{31}(m-1) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right) \\ &- \frac{i}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \widehat{h}_{21}(m-1) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right) . \end{split}$$

Hence, we conclude that one way to obtain spectral asymmetry is to choose such a perturbation matrix  $h_{\alpha\beta}(x^1)$  so that  $\widehat{h}_{11}(m) = \widehat{h}_{21}(m) = \widehat{h}_{31}(m) = 0$  for all  $m \in \mathbb{Z}$  such that the term

$$\varepsilon_{\beta\gamma1}\sum_{m\in\mathbb{Z}\setminus\{0\}}m\widehat{\widehat{h}}_{\alpha\beta}(m)\widehat{h}_{\alpha\gamma}(m)\neq 0.$$

In that case we have that  $\lambda_{+}^{(2)} + \lambda_{-}^{(2)} \neq 0$ .

*Proof of Theorem 3.5.7.* In order to prove Theorem 3.5.7 we need to first write down explicitly the differential operators  $W_{1/2}^{(1)}$  i  $W_{1/2}^{(2)}$  appearing in the asymptotic expansion of the perturbed massless Dirac operator on half-densities

$$W_{1/2}(\epsilon) = W_{1/2}^{(0)} + \epsilon W_{1/2}^{(1)} + \epsilon^2 W_{1/2}^{(2)} + O(\epsilon^3).$$

Substituting formulae (3.37) and (3.38) into (3.57), we get that

$$W_{1/2}^{(1)} = \frac{i}{4} \begin{pmatrix} h_3^{1} & h_1^{1} - ih_2^{1} \\ h_1^{1} + ih_2^{1} & -h_3^{1} \end{pmatrix} \frac{d}{dx^{1}} \\ + \frac{i}{4} \frac{d}{dx^{1}} \begin{pmatrix} h_3^{1} & h_1^{1} - ih_2^{1} \\ h_1^{1} + ih_2^{1} & -h_3^{1} \end{pmatrix}$$
(3.75)

and

$$W_{1/2}^{(2)} = -\frac{3\mathrm{i}}{16} \begin{pmatrix} (h^2)_3^{\ 1} & (h^2)_1^{\ 1} - \mathrm{i} (h^2)_2^{\ 1} \\ (h^2)_1^{\ 1} + \mathrm{i} (h^2)_2^{\ 1} & -(h^2)_3^{\ 1} \end{pmatrix} \frac{d}{dx^1} \\ -\frac{3\mathrm{i}}{16} \frac{d}{dx^1} \begin{pmatrix} (h^2)_3^{\ 1} & (h^2)_1^{\ 1} - \mathrm{i} (h^2)_2^{\ 1} \\ (h^2)_1^{\ 1} + \mathrm{i} (h^2)_2^{\ 1} & -(h^2)_3^{\ 1} \end{pmatrix} \\ +\frac{\mathrm{i}}{16} \begin{pmatrix} k_3^{\ 1} & k_1^{\ 1} - \mathrm{i} k_2^{\ 1} \\ k_1^{\ 1} + \mathrm{i} k_2^{\ 1} & -k_3^{\ 1} \end{pmatrix} \frac{d}{dx^1} \\ +\frac{\mathrm{i}}{16} \frac{d}{dx^1} \begin{pmatrix} k_3^{\ 1} & k_1^{\ 1} - \mathrm{i} k_2^{\ 1} \\ k_1^{\ 1} + \mathrm{i} k_2^{\ 1} & -k_3^{\ 1} \end{pmatrix} - \frac{1}{16} \varepsilon_{\beta\gamma1} h_{\alpha\beta} \frac{dh_{\alpha\gamma}}{dx^1} I. \quad (3.76)$$

The explicit formulae for the coefficients  $\lambda^{(1)}$  and  $\lambda^{(2)}$  are given by (3.52) and (3.53).

Now consider the eigenvalue  $\lambda = 1$  of the massless Dirac operator in axisymmetric case (3.57). The normalised eigenvector  $v^{(0)}$  which corresponds to the eigenvalue  $\lambda^{(0)} = 1$  is given by

$$v^{(0)}(x^{1}) = \frac{1}{2\sqrt{\pi}} \begin{pmatrix} 1\\1 \end{pmatrix} e^{ix^{1}}.$$
 (3.77)

Of course, the vector

$$w^{(0)}(x^{1}) = C(v^{(0)}(x^{1})) = \frac{1}{2\sqrt{\pi}} \begin{pmatrix} -1 \\ 1 \end{pmatrix} e^{-ix^{1}}$$

is also an eigenvector of the massless Dirac operator corresponding to the eigenvalue  $n \in \mathbb{Z}$ , where C is the charge conjugation operator (3.14), see formulae (3.58) and (3.59). We calculate the coefficients  $\lambda_{+}^{(1)}$  and  $\lambda_{+}^{(2)}$  for the

eigenvector  $v^{(0)}(x^1)$ . Note that the result would be the same if we chose the eigenvector  $w^{(0)}(x^1)$ .

Using formulae (3.52), (3.75) for the eigenvector (3.77) and integrating by parts, we get that

$$\lambda_{+}^{(1)} = \langle W_{1/2}^{(1)} v^{(0)}, v^{(0)} \rangle = -\frac{1}{2} \hat{h}_{11}(0),$$

see Appendix D for detailed calculations.

We will calculate the coefficient  $\lambda_{+}^{(2)}$ , see (3.53), in several stages, for sake of simplicity and readability. Let us first calculate the part  $\langle W_{1/2}^{(2)}v^{(0)}, v^{(0)} \rangle$ . Using (3.76), for the eigenvector (3.77) we get that

$$\langle W_{1/2}^{(2)}v^{(0)}, v^{(0)} \rangle = \frac{3}{8} \widehat{(h^2)}_{11}(0) - \frac{1}{8} \widehat{k}_{11}(0) - \frac{1}{16} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \overline{\widehat{h}}_{\alpha\beta}(m) \widehat{h}_{\alpha\gamma}(m).$$

To calculate the part  $\langle (W_{1/2}^{(1)} - \lambda^{(1)})Q(W_{1/2}^{(1)} - \lambda^{(1)})v^{(0)}, v^{(0)} \rangle$  we need the pseudoinverse operator Q. The construction of the pseudoinverse operator Q is explained in Section 3.4.1. For the eigenvectors (3.58) and (3.59) the projection operator (3.46) is given by

$$P = \frac{1}{4\pi} \sum_{m \in \mathbb{Z}} \left[ \begin{pmatrix} 1\\1 \end{pmatrix} e^{imx^1} \int_0^{2\pi} \begin{pmatrix} 1\\1 \end{pmatrix} \begin{pmatrix} \cdot \end{pmatrix} e^{-imy^1} dy^1 + \begin{pmatrix} -1\\1 \end{pmatrix} e^{-imx^1} \int_0^{2\pi} \begin{pmatrix} -1&1 \end{pmatrix} \begin{pmatrix} \cdot \end{pmatrix} e^{imy^1} dy^1 \right]$$

and hence using (3.47), the explicit formula for the pseudoinverse Q of the operator  $W_{1/2}(0) - I$  is given by

$$Q = \frac{1}{4\pi} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \left[ e^{imx^1} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix} \int_0^{2\pi} e^{-imy^1} (\cdot) dy^1 + e^{-imx^1} \begin{pmatrix} 1 & -1 \\ -1 & 1 \end{pmatrix} \int_0^{2\pi} e^{imy^1} (\cdot) dy^1 \right]$$
(3.78)

Now, using (3.78), lengthy but straightforward calculations produce

$$\langle (W_{1/2}^{(1)} - \lambda^{(1)})Q(W_{1/2}^{(1)} - \lambda^{(1)})v^{(0)}, v^{(0)} \rangle =$$

$$\frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} (m+1)^2 \, \widehat{h}_{11}(m-1) \overline{\widehat{h}_{11}(m-1)} +$$

$$\frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \, \left( \widehat{h}_{31}(m+1) + i \widehat{h}_{21}(m+1) \right) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right) .$$

So, finally we get the formula (3.72).

The procedure of deriving of the formulae (3.71) and (3.73) is analogous to the derivations of the formulae (3.70) and (3.72). The only difference is that in calculations we use the eigenvector

$$v^{(0)}(x^1) = \frac{1}{2\sqrt{\pi}} \begin{pmatrix} 1\\1 \end{pmatrix} e^{-ix^1}.$$
 (3.79)

which correspond to the eigenvalue  $\lambda = -1$  and the formula for the pseudoinverse Q of the operator  $W_{1/2}(0) + I$  is given by

$$Q = \frac{1}{4\pi} \sum_{m \in \mathbb{Z} \setminus \{-1\}} \frac{1}{m+1} \left[ e^{imx^1} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix} \int_0^{2\pi} e^{-imy^1} (\cdot) dy^1 + e^{-imx^1} \begin{pmatrix} 1 & -1 \\ -1 & 1 \end{pmatrix} \int_0^{2\pi} e^{imy^1} (\cdot) dy^1 \right].$$
 (3.80)

Using formulae (3.52), (3.75) for the eigenvector (3.79) and integrating by parts, we get that

$$\lambda_{-}^{(1)} = \langle W_{1/2}^{(1)} v^{(0)}, v^{(0)} \rangle = \frac{1}{2} \hat{h}_{11}(0).$$

We have that

$$\langle W_{1/2}^{(2)}v^{(0)}, v^{(0)} \rangle = -\frac{3}{8}\widehat{(h^2)}_{11}(0) + \frac{1}{8}\widehat{k}_{11}(0) - \frac{i}{16}\varepsilon_{\beta\gamma1}\sum_{m\in\mathbb{Z}\setminus\{0\}} m\overline{\widehat{h}_{\alpha\beta}(m)}\widehat{h}_{\alpha\gamma}(m)$$

and

$$\langle (W_{1/2}^{(1)} - \lambda^{(1)})Q(W_{1/2}^{(1)} - \lambda^{(1)})v^{(0)}, v^{(0)} \rangle =$$

$$\frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} \frac{1}{m+1} (m-1)^2 \, \widehat{h}_{11}(m+1) \overline{\widehat{h}_{11}(m+1)} +$$

$$\frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \, \left( \widehat{h}_{31}(m-1) + i \widehat{h}_{21}(m-1) \right) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right)$$

Combining the above results we obtain the formula (3.73).

#### 

### 3.5.3 Explicit examples of spectral asymmetry

In this section we present two explicit examples of perturbation matrices for which the massless Dirac operator in the axisymmetric case (3.57) has an

asymmetric spectrum. The explicit formulae for the massless Dirac operator are also derived for two different coframes.

The asymptotic formulae for the eigenvalues  $\pm 1$  are derived in two different ways, thus confirming our new results for the asymptotic coefficients from Theorem 3.5.7. First we derive the asymptotic coefficient of the eigenvalues  $\pm 1$  of the massless Dirac operator in the axisymmetric case by definition, using the formulae (3.52) and (3.53) and secondly, we use our new formulae (3.70), (3.71), (3.72) and (3.73) which are expressed only by using the Fourier's coefficients of the perturbation matrices h and k.

In the first example we present the perturbation matrices of the coframe for which the linear coefficients in the asymptotic expansions of the eigenvalues  $\pm 1$  are zero. In the second example, the linear factor is non-zero. These examples give us the guidelines on how to choose the perturbed coframe in order to obtain the spectral asymmetry of the massless Dirac operator.

**Example 3.5.12.** We choose the following perturbation matrices

$$h_{\alpha\beta}(x^{1}) = 2 \begin{pmatrix} 0 & 0 & 0 \\ 0 & \cos x^{1} & \sin x^{1} \\ 0 & \sin x^{1} & -\cos x^{1} \end{pmatrix}, \quad k_{\alpha\beta}(x^{1}) = \begin{pmatrix} \sin x^{1} & \cos x^{1} & 0 \\ \cos x^{1} & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}.$$

Using formulae (3.75) and (3.76), we get that

$$\begin{aligned} A^{(1)} &= 0, \\ A^{(2)} &= \frac{i}{16} \begin{pmatrix} 0 & -i\cos x^{1} + \sin x^{1} \\ i\cos x^{1} + \sin x^{1} & 0 \end{pmatrix} \frac{d}{dx^{1}} \\ &+ \frac{i}{16} \frac{d}{dx^{1}} \begin{pmatrix} 0 & -i\cos x^{1} + \sin x^{1} \\ i\cos x^{1} + \sin x^{1} & 0 \end{pmatrix} - \frac{1}{2}I, \end{aligned}$$

so the corresponding perturbed massless Dirac operator in the axisymmetric case (3.57) is explicitly given by

$$W(\epsilon) = -i \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \frac{d}{dx^{1}} + \frac{i\epsilon^{2}}{16} \begin{pmatrix} 0 & -i\cos x^{1} + \sin x^{1} \\ i\cos x^{1} + \sin x^{1} & 0 \end{pmatrix} \frac{d}{dx^{1}} \\ + \frac{i\epsilon^{2}}{16} \frac{d}{dx^{1}} \begin{pmatrix} 0 & -i\cos x^{1} + \sin x^{1} \\ i\cos x^{1} + \sin x^{1} & 0 \end{pmatrix} - \frac{\epsilon^{2}}{2}I.$$

Using formulae (3.13), (3.52), (3.53), (3.78) and (3.80) we get that

$$\lambda_{+}(\epsilon) = 1 - \frac{1}{2}\epsilon^{2} + O(\epsilon^{3}), \qquad (3.81)$$

$$\lambda_{-}(\epsilon) = -1 - \frac{1}{2}\epsilon^{2} + O(\epsilon^{3}).$$
 (3.82)

Application of the Fourier transform (3.74) gives us

$$\widehat{h}_{\alpha\beta}(m) = \begin{cases} \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1 & -i \\ 0 & -i & -1 \end{pmatrix} & \text{for } m = 1, \\ 0 & & \text{for } m = 2, 3, \dots, \end{cases}$$
(3.83)

$$\widehat{k}_{\alpha\beta}(m) = \begin{cases} \begin{pmatrix} -i/2 & 1/2 & 0\\ 1/2 & 0 & 0\\ 0 & 0 & 0 \end{pmatrix} & \text{for } m = 1, \\ (3.84)$$

Substituting (3.83), (3.84) and (3.85) into (3.70), (3.71), (3.72) and (3.73) we get that

$$\lambda_{\pm}^{(1)} = 0, \ \lambda_{+}^{(2)} = \lambda_{-}^{(2)} = -\frac{1}{2}$$

which is in accordance with the asymptotic formulae (3.81) and (3.82).

Example 3.5.13. We choose the following perturbation matrices

$$h_{\alpha\beta}(x^{1}) = \begin{pmatrix} 1 & \cos x^{1} & \sin x^{1} \\ \cos x^{1} & \cos x^{1} & \sin x^{1} \\ \sin x^{1} & \sin x^{1} & -\cos x^{1} \end{pmatrix}, \quad k_{\alpha\beta}(x^{1}) = \begin{pmatrix} \sin x^{1} & \cos x^{1} & 0 \\ \cos x^{1} & -\sin x^{1} & 0 \\ 0 & 0 & 0 \end{pmatrix}.$$

Using formulae (3.75) and (3.76), we get that

$$\begin{aligned} A^{(1)} &= \frac{\mathrm{i}}{4} \left( \begin{array}{cc} \sin x^1 & 1 - \mathrm{i} \cos x^1 \\ 1 + \mathrm{i} \cos x^1 & -\sin x^1 \end{array} \right) \frac{d}{dx^1} + \frac{\mathrm{i}}{4} \frac{d}{dx^1} \left( \begin{array}{cc} \sin x^1 & 1 - \mathrm{i} \cos x^1 \\ 1 + \mathrm{i} \cos x^1 & -\sin x^1 \end{array} \right) \\ A^{(2)} &= -\frac{3\mathrm{i}}{16} \left( \begin{array}{cc} \sin x^1 & 2 - \mathrm{i} - \mathrm{i} \cos x^1 \\ 2 + \mathrm{i} + \mathrm{i} \cos x^1 & -\sin x^1 \end{array} \right) \frac{d}{dx^1} \\ &- \frac{3\mathrm{i}}{16} \frac{d}{dx^1} \left( \begin{array}{cc} \sin x^1 & 2 - \mathrm{i} - \mathrm{i} \cos x^1 \\ 2 + \mathrm{i} + \mathrm{i} \cos x^1 & -\sin x^1 \end{array} \right) \\ &+ \frac{\mathrm{i}}{16} \left( \begin{array}{cc} 0 & -\mathrm{i} \cos x^1 + \sin x^1 \\ \mathrm{i} \cos x^1 + \sin x^1 & 0 \end{array} \right) \frac{d}{dx^1} \\ &+ \frac{\mathrm{i}}{16} \frac{d}{dx^1} \left( \begin{array}{cc} 0 & -\mathrm{i} \cos x^1 + \sin x^1 \\ \mathrm{i} \cos x^1 + \sin x^1 & 0 \end{array} \right) - \frac{3}{16} I, \end{aligned}$$

so the corresponding perturbed massless Dirac operator in the axisymmetric case is given by

$$\begin{split} W(\epsilon) &= -\mathrm{i} \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \frac{d}{dx^{1}} + \frac{\mathrm{i}\epsilon}{4} \begin{pmatrix} \sin x^{1} & 1 - \mathrm{i} \cos x^{1} \\ 1 + \mathrm{i} \cos x^{1} & -\sin x^{1} \end{pmatrix} \frac{d}{dx^{1}} \\ &+ \frac{\mathrm{i}\epsilon}{4} \frac{d}{dx^{1}} \begin{pmatrix} \sin x^{1} & 1 - \mathrm{i} \cos x^{1} \\ 1 + \mathrm{i} \cos x^{1} & -\sin x^{1} \end{pmatrix} \\ &- \frac{3\mathrm{i}\epsilon^{2}}{16} \begin{pmatrix} \sin x^{1} & 2 - \mathrm{i} - \mathrm{i} \cos x^{1} \\ 2 + \mathrm{i} + \mathrm{i} \cos x^{1} & -\sin x^{1} \end{pmatrix} \frac{d}{dx^{1}} \\ &- \frac{3\mathrm{i}\epsilon^{2}}{16} \frac{d}{dx^{1}} \begin{pmatrix} \sin x^{1} & 2 - \mathrm{i} - \mathrm{i} \cos x^{1} \\ 2 + \mathrm{i} + \mathrm{i} \cos x^{1} & -\sin x^{1} \end{pmatrix} \\ &+ \frac{\mathrm{i}\epsilon^{2}}{16} \begin{pmatrix} 0 & -\mathrm{i} \cos x^{1} + \sin x^{1} \\ \mathrm{i} \cos x^{1} + \sin x^{1} & 0 \end{pmatrix} \frac{d}{dx^{1}} \\ &+ \frac{\mathrm{i}\epsilon^{2}}{16} \frac{d}{dx^{1}} \begin{pmatrix} 0 & -\mathrm{i} \cos x^{1} + \sin x^{1} \\ \mathrm{i} \cos x^{1} + \sin x^{1} & 0 \end{pmatrix} - \frac{3\epsilon^{2}}{16} I. \end{split}$$

Again using the formulae (3.13), (3.52), (3.53), (3.78) and (3.80) we get that

$$\lambda_{+}(\epsilon) = 1 - \frac{1}{2}\epsilon + \frac{3}{4}\epsilon^{2} + O(\epsilon^{3}), \qquad (3.86)$$

$$\lambda_{-}(\epsilon) = -1 + \frac{1}{2}\epsilon - \epsilon^{2} + O(\epsilon^{3}).$$
(3.87)

Application of the Fourier transform (3.74) gives us

$$\widehat{h}_{\alpha\beta}(m) = \begin{cases}
\begin{pmatrix}
1 & 0 & 0 \\
0 & 0 & 0 \\
0 & 0 & 0
\end{pmatrix} & \text{for } m = 0, \\
\begin{pmatrix}
0 & 1/2 & -i/2 \\
1/2 & 1/2 & -i/2 \\
-i/2 & -i/2 & -1/2
\end{pmatrix} & \text{for } m = 1, \\
0 & \text{for } m = 2, 3, \dots, \end{cases}$$

$$\widehat{k}_{\alpha\beta}(m) = \begin{cases}
\begin{pmatrix}
-i/2 & 1/2 & 0 \\
1/2 & i/2 & 0 \\
0 & 0 & 0
\end{pmatrix} & \text{for } m = 1, \\
0 & \text{for } m = 2, 3, \dots, \end{cases}$$
(3.89)
$$0 & \text{for } m = 2, 3, \dots, \end{cases}$$

$$\widehat{(h^2)}_{\alpha\beta}(m) = \begin{cases} \begin{pmatrix} 2 & 1 & 0 \\ 1 & 3/2 & 0 \\ 0 & 0 & 3/2 \end{pmatrix} & \text{for } m = 0, \\ \begin{pmatrix} 0 & 1/2 & -i/2 \\ 1/2 & 0 & 0 \\ -i/2 & 0 & 0 \end{pmatrix} & \text{for } m = 1, \\ \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1/2 & -i/4 \\ 0 & -i/4 & -1/4 \end{pmatrix} & \text{for } m = 2, \\ \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1/2 & -i/4 \\ 0 & -i/4 & -1/4 \end{pmatrix} & \text{for } m = 3, 4, \dots. \end{cases}$$
(3.90)

Substituting (3.88), (3.89) and (3.90) into (3.70), (3.71), (3.72) and (3.73) we get that

$$\lambda_{\pm}^{(1)} = \mp \frac{1}{2}, \quad \lambda_{+}^{(2)} = \frac{3}{4}, \text{ and } \lambda_{-}^{(2)} = -1,$$

which is in accordance with the asymptotic formulae (3.86) and (3.87).

**Remark 3.5.14.** In Example 3.5.12 we have shown that if we choose the perturbation matrix  $h_{\alpha\beta}(x^1)$  such that  $h_{11}(x^1) = 0$ , the linear coefficients  $\lambda_{\pm}^{(1)}$  are zero, which is in accordance with (3.70) and (3.71). However, if we choose the perturbation matrix  $h_{\alpha\beta}(x^1)$  as in Example 3.5.13 with  $h_{11}(x^1) \neq 0$ , the linear coefficient is nonzero and for the eigenvalues  $\pm 1$  they have opposite sign. This is in accordance with Remark 3.5.9. The spectral asymmetry in both examples is obtained in the quadratic term.

## Appendix A

## Einstein and Yang-Mills Equations

In this appendix we present the derivation of the Einstein field equations, the Yang-Mills equation and the complementary Yang-Mills equation. These results are well known, see e.g. [53, 59, 72, 75, 99], but here we provide the detailed derivation of these equations.

### A.1 The Einstein field equations

The Einstein field equations lie at the center of general relativity. It relates a spacetime geometry and a matter. The vacuum Einstein equations

$$Ric_{\alpha\beta} - \frac{1}{2}\mathcal{R}g_{\alpha\beta} = 0 \tag{A.1}$$

are obtained by varying the Einstein-Hilbert action

$$S_{EH} := \frac{c^4}{16\pi G} \int \mathcal{R}\sqrt{|\det g|},\tag{A.2}$$

with respect to the metric g, as was previously stated in [75]. Here  $\mathcal{R}$  is the scalar curvature (1.19), *Ric* the Ricci curvature (1.18), g is the metric, cis the speed of light and G is the gravitational constant, the recommended numerical value of which is 6.673 84(80)  $\times 10^{-11} \text{m}^3 \text{kg}^{-1} \text{s}^{-2}$ , with relative standard uncertainty  $1.2 \times 10^{-4}$ , see [63]. The full field equations are then obtained by adding the matter Lagrangian to the Einstein-Hilbert action, which gives us the Einstein equations in tensor form

$$Ric_{\mu\nu} - \frac{1}{2}\mathcal{R}g_{\mu\nu} = \frac{8\pi G}{c^4} T_{\mu\nu},$$

where T is the stress energy tensor that arises from the matter Lagrangian, see e.g. [59]. The simplest solution of this equations is the Minkowski space-time from special relativity.

**Proposition A.1.1.** Let (M, g) be a (pseudo-)Riemannian manifold. Under the variation  $g_{\mu\nu} \rightarrow g_{\mu\nu} + \delta g_{\mu\nu}$  and  $g_{\mu\nu}$ , det g change as

- $\delta g^{\mu\nu} = -g^{\mu\kappa}g^{\lambda\nu}\delta g_{\kappa\lambda},$
- $\delta \det g_{\mu\nu} = \det g_{\mu\nu} g^{\mu\nu} \delta g_{\mu\nu}$ ,
- $\delta \sqrt{|\det g_{\mu\nu}|} = \frac{1}{2} \sqrt{|\det g_{\mu\nu}|} g^{\mu\nu} \delta g_{\mu\nu}.$

For the proof of Proposition A.1.1 see [65]. Since  $\mathcal{R} = R^{\kappa\mu}_{\ \ \kappa\mu}$ , we have that

$$\frac{\delta S_{EH}}{\delta g} = \delta \int R^{\kappa\mu}{}_{\kappa\mu} \sqrt{|\det g|} = \delta \int R^{\kappa}{}_{\lambda\kappa\mu} g^{\lambda\mu} \sqrt{|\det g|} \\
= \int \delta \left( R^{\kappa}{}_{\lambda\kappa\mu} g^{\lambda\mu} \sqrt{|\det g|} \right) = \int R^{\kappa}{}_{\lambda\kappa\mu} \delta \left( g^{\lambda\mu} \sqrt{|\det g|} \right) \\
= \int R^{\kappa}{}_{\lambda\kappa\mu} \delta \left( -g^{\lambda\alpha} g^{\beta\mu} \delta g_{\alpha\beta} \sqrt{|\det g|} + g^{\lambda\mu} \frac{1}{2} \sqrt{|\det g|} g^{\alpha\beta} \delta g_{\alpha\beta} \right) \\
= \int (\delta g_{\alpha\beta}) \left( -R^{\kappa}{}_{\lambda\kappa\mu} g^{\lambda\alpha} g^{\beta\mu} + \frac{1}{2} R^{\kappa}{}_{\lambda\kappa\mu} g^{\lambda\mu} g^{\alpha\beta} \right) \sqrt{|\det g|}.$$

So, we get that

$$\frac{\delta S_{EH}}{\delta g} = \int (\delta g_{\alpha\beta}) \left( -R^{\kappa\alpha}{}^{\beta}{}_{\kappa} + \frac{1}{2}\mathcal{R}g^{\alpha\beta} \right) = \int (\delta g_{\alpha\beta}) \left( -Ric^{\alpha\beta} + \frac{1}{2}\mathcal{R}g^{\alpha\beta} \right) d\beta$$

which gives us equation (A.1).

### A.2 The Yang-Mills equations

We define the Yang-Mills action as

$$S_{YM} := \int R^{\kappa}{}_{\lambda\mu\nu} R^{\lambda}{}_{\kappa}{}^{\mu\nu} \sqrt{|\det g|}.$$
(A.3)

The variation of the action (A.3) with respect to the metric produces the socalled *complementary Yang-Mills equation* for the affine connection. Using Proposition A.1.1, we get that

$$\frac{\delta S_{YM}}{\delta g} = \int \delta \left( R^{\kappa}_{\ \lambda\mu\nu} R^{\lambda}_{\ \kappa\mu'\nu'} g^{\mu\mu'} g^{\nu\nu'} \sqrt{|\det g|} \right) \\
= \int (\delta g_{\alpha\beta}) R^{\kappa}_{\ \lambda\mu\nu} R^{\lambda}_{\ \kappa\mu'\nu'} \left( -g^{\mu\alpha} g^{\beta\mu'} g^{\nu\nu'} - g^{\nu\alpha} g^{\beta\nu'} g^{\mu\mu'} + \frac{1}{2} g^{\mu\mu'} g^{\nu\nu'} g^{\alpha\beta} \right) \\
= -\int (\delta g_{\alpha\beta}) \left( R^{\kappa}_{\ \lambda\nu} R^{\lambda}_{\ \kappa} R^{\lambda} + R^{\kappa}_{\ \lambda\mu} R^{\lambda}_{\ \kappa} R^{\mu\beta} - \frac{1}{2} g^{\alpha\beta} R^{\kappa}_{\ \lambda\mu\nu} R^{\lambda}_{\ \kappa} R^{\mu\nu} \right) \\
= -2 \int (\delta g_{\alpha\beta}) \left( R^{\kappa}_{\ \lambda\mu} R^{\lambda}_{\ \kappa} R^{\mu\beta} - \frac{1}{4} g^{\alpha\beta} R^{\kappa}_{\ \lambda\mu\nu} R^{\lambda}_{\ \kappa} R^{\mu\nu} \right), \quad (A.4)$$

hence the complementary Yang-Mils equation is

$$R^{\kappa}_{\ \lambda\mu}{}^{\alpha}R^{\lambda}_{\ \kappa}{}^{\mu\beta} - \frac{1}{4}g^{\alpha\beta}R^{\kappa}_{\ \lambda\mu\nu}R^{\lambda}_{\ \kappa}{}^{\mu\nu} = 0.$$
(A.5)

Using the notation  $H = H_{\alpha}{}^{\beta} := R^{\kappa}{}_{\lambda\mu\alpha}R^{\lambda}{}_{\kappa}{}^{\mu\beta}$  and  $\delta = \delta_{\alpha}{}^{\beta}$  is the identity tensor, the equation (A.5) is equivalent to the equation

$$H - \frac{1}{4}(\operatorname{tr} \, H)\delta = 0,$$

as was obtained in [53].

By varying the action (A.3) with respect to the connection we get the Yang - Mills equation for the affine connection. As we vary the curvature independently, we have that

$$\frac{\delta S_{YM}}{\delta \Gamma} = \int \delta \left( R^{\kappa}_{\ \lambda\mu\nu} R^{\lambda}_{\ \kappa}^{\ \mu\nu} \right) \sqrt{|\det g|} 
= \int \delta \left( R^{\kappa}_{\ \lambda\mu\nu} \right) R^{\lambda}_{\ \kappa}^{\ \mu\nu} \sqrt{|\det g|} + \int \delta \left( R^{\lambda}_{\ \kappa}^{\ \mu\nu} \right) R^{\kappa}_{\ \lambda\mu\nu} \sqrt{|\det g|} 
= 2 \int \delta \left( R^{\kappa}_{\ \lambda\mu\nu} \right) R^{\lambda}_{\ \kappa}^{\ \mu\nu} \sqrt{|\det g|}.$$
(A.6)

Using the formula for the curvature tensor (1.17), we get that

$$\delta R^{\kappa}{}_{\lambda\mu\nu} = \partial_{\mu} \left(\delta \Gamma^{\kappa}{}_{\nu\lambda}\right) - \partial_{\nu} \left(\delta \Gamma^{\kappa}{}_{\mu\lambda}\right) + \left(\delta \Gamma^{\kappa}{}_{\mu\eta}\right) \Gamma^{\eta}{}_{\nu\lambda} + \Gamma^{\kappa}{}_{\mu\eta} \delta \Gamma^{\eta}{}_{\nu\lambda} - \left(\delta \Gamma^{\kappa}{}_{\nu\eta}\right) \Gamma^{\eta}{}_{\mu\lambda} - \Gamma^{\kappa}{}_{\nu\eta} \delta \Gamma^{\eta}{}_{\mu\lambda}.$$
(A.7)

Substituting equation (A.7) into equation (A.6), we get that

$$\begin{split} \frac{1}{2} \frac{\delta S_{YM}}{\delta \Gamma} &= -\int \left(\delta \Gamma^{\kappa}_{\ \nu\lambda}\right) \partial_{\mu} \left(R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|}\right) \frac{\sqrt{|\det g|}}{\sqrt{|\det g|}} \\ &+ \int \left(\delta \Gamma^{\kappa}_{\ \mu\lambda}\right) \partial_{\nu} \left(R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|}\right) \frac{\sqrt{|\det g|}}{\sqrt{|\det g|}} \\ &+ \int \left(\delta \Gamma^{\kappa}_{\ \mu\eta}\right) \Gamma^{\eta}_{\ \nu\lambda} R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|} + \int \Gamma^{\kappa}_{\ \mu\eta} \left(\delta \Gamma^{\eta}_{\ \mu\lambda}\right) R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|} \\ &- \int \left(\delta \Gamma^{\kappa}_{\ \nu\eta}\right) \Gamma^{\eta}_{\ \mu\lambda} R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|} - \int \Gamma^{\kappa}_{\ \nu\eta} \left(\delta \Gamma^{\eta}_{\ \mu\lambda}\right) R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|}. \end{split}$$

Using the fact that for any curvature the antisymmetry  $R^{\kappa}_{\ \lambda\mu\nu} = -R^{\kappa}_{\ \lambda\nu\mu}$  holds and renaming some indices, we get that

$$\frac{1}{2} \frac{\delta S_{YM}}{\delta \Gamma} = 2 \int \left( \delta \Gamma^{\kappa}_{\ \mu\lambda} \right) \partial_{\nu} \left( R^{\lambda}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|} \right) \frac{\sqrt{|\det g|}}{\sqrt{|\det g|}} + 2 \int \left( \delta \Gamma^{\kappa}_{\ \mu\lambda} \right) \Gamma^{\lambda}_{\ \nu\eta} R^{\eta}_{\ \kappa} \ ^{\mu\nu} \sqrt{|\det g|} - 2 \int \Gamma^{\eta}_{\ \nu\kappa} \left( \delta \Gamma^{\kappa}_{\ \mu\lambda} \right) R^{\lambda}_{\ \eta} \ ^{\mu\nu} \sqrt{|\det g|},$$

Then

$$\frac{\delta S_{YM}}{\delta \Gamma} = 4 \int \left(\delta \Gamma_{\mu}\right) \frac{1}{\sqrt{|\det g|}} \left(\partial_{\nu} + \left[\Gamma_{\nu}, \cdot\right]\right) \left(\sqrt{|\det g|} R^{\mu\nu}\right).$$

Hence, the Yang-Mills equation is

$$\left(\partial_{\nu} + \left[\Gamma_{\nu}, \cdot\right]\right) \left(\sqrt{\left|\det g\right|} R^{\mu\nu}\right) = 0.$$

## Appendix B

### **Bianchi Identity for Curvature**

In this appendix we present the derivation of the Bianchi identity for curvature which is used in the derivation of the explicit form of the second field equation (2.50). We will use the following assumptions:

- (i) Our spacetime is metric compatible;
- (ii) Torsion is purely axial;
- (*iii*) Ricci curvature is symmetric;
- (iv) Scalar curvature  $\mathcal{R}$  and pseudoscalar curvature  $\mathcal{R}_*$  are zero.

**Remark B.0.1.** The antisymmetry  $R_{\kappa\lambda\mu\nu} = -R_{\kappa\lambda\nu\mu}$  is true for any curvature and the antisymmetry  $R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu}$  is a consequence of the metric compatibility. Note that we will *not* use the symmetry  $R_{\kappa\lambda\mu\nu} = R_{\mu\nu\kappa\lambda}$ , since the curvature of generalised pp-waves with purely axial torsion does not possess this property.

## **B.1** Explicit formula for $R^{(5)}$ piece of curvature

Now we will derive the explicit formula for the  $R^{(5)}$  irreducible piece of curvature (1.35), see Section 1.4.1 and [99] for more on this irreducible decomposition. It is known that the piece  $R^{(5)}$  and the pieces  $R^{(9,l)}_*$ , see (1.41), are related as

$$R^{(5)} = -(R^{(9,1)}_* + R^{(9,2)}_*)^*, \tag{B.1}$$

where \* denotes the right Hodge star (1.23). Using equation (B.1), we have that

$$R_{\kappa\lambda\mu\nu}^{(5)} = -\left(\frac{3}{8}(g_{\kappa\mu}\mathcal{S}_{*}^{(1)}{}_{\lambda\nu} - g_{\kappa\nu}\mathcal{S}_{*}^{(1)}{}_{\lambda\mu}) - \frac{1}{8}(g_{\lambda\mu}\mathcal{S}_{*}^{(1)}{}_{\kappa\nu} - g_{\lambda\nu}\mathcal{S}_{*}^{(1)}{}_{\kappa\mu}) - \frac{1}{8}(g_{\kappa\mu}\mathcal{S}_{*}^{(2)}{}_{\lambda\nu} - g_{\kappa\nu}\mathcal{S}_{*}^{(2)}{}_{\lambda\mu}) + \frac{3}{8}(g_{\lambda\mu}\mathcal{S}_{*}^{(2)}{}_{\kappa\nu} - g_{\lambda\nu}\mathcal{S}_{*}^{(2)}{}_{\kappa\mu})\right)^{*} \\ = \frac{1}{8}\left(\left(3\mathcal{S}_{*}^{(1)}{}_{\lambda\eta} - \mathcal{S}_{*}^{(2)}{}_{\lambda\eta}\right)\epsilon^{\eta}{}_{\kappa\mu\nu} + \left(3\mathcal{S}_{*}^{(2)}{}_{\kappa\eta} - \mathcal{S}_{*}^{(1)}{}_{\kappa\eta}\right)\epsilon^{\eta}{}_{\lambda\mu\nu}\right).$$

For metric compatible spacetimes we have that  $R_{\kappa\lambda\mu\nu} = -R_{\lambda\kappa\mu\nu}$  and consequently  $\mathcal{S}^{(1)}_{*\ \mu\nu} = -\mathcal{S}^{(2)}_{*\ \mu\nu}$ . Hence, we get that

$$R^{(5)}_{\kappa\lambda\mu\nu} = \frac{1}{4} \left( \left( \mathcal{R}ic^{(1)}_{*\lambda\eta} + \mathcal{R}ic^{(1)}_{*\eta\lambda} \right) \epsilon^{\eta}_{\kappa\mu\nu} - \left( \mathcal{R}ic^{(1)}_{*\kappa\eta} + \mathcal{R}ic^{(1)}_{*\eta\kappa} \right) \epsilon^{\eta}_{\lambda\mu\nu} \right).$$
(B.2)

Under the assumptions that Ric is symmetric and pseudoscalar curvature  $\mathcal{R}_*$  is zero, we have the simplified version of formula (B.2)

$$R^{(5)}_{\kappa\lambda\mu\nu} = \frac{1}{2} \left( \epsilon^{\eta}{}_{\kappa\mu\nu} Ric^{(1)}_{*\lambda\eta} - \epsilon^{\eta}{}_{\lambda\mu\nu} Ric^{(1)}_{*\kappa\eta} \right).$$
(B.3)

### **B.2** The derivation of the Bianchi identity

Assumption (i) implies that the piece of curvature  $R^{(6)}$  (1.36) is zero and assumption (iv) clearly implies that the  $R^{(2)}$  (1.32) and  $R^{(4)}$  (1.34) pieces of curvature are zero. Hence, under the above assumptions, the curvature (1.17) has only *three* nonzero irreducible pieces, namely  $R^{(1)}$ ,  $R^{(3)}$  and  $R^{(5)}$ . It can therefore be represented as

$$R_{\kappa\lambda\mu\nu} = \frac{1}{2} (g_{\kappa\mu}Ric_{\lambda\nu} - g_{\lambda\mu}Ric_{\kappa\nu} + g_{\lambda\nu}Ric_{\kappa\mu} - g_{\kappa\nu}Ric_{\lambda\mu}) + \mathcal{W}_{\kappa\lambda\mu\nu} + \frac{1}{2} (-\epsilon^{\eta}{}_{\lambda\mu\nu}Ric_{*\kappa\eta} + \epsilon^{\eta}{}_{\kappa\mu\nu}Ric_{*\lambda\eta}).$$
(B.4)

The Bianchi identity for curvature is

$$\nabla_{\xi} R^{\kappa}{}_{\lambda\mu\nu} + \nabla_{\nu} R^{\kappa}{}_{\lambda\xi\mu} + \nabla_{\mu} R^{\kappa}{}_{\lambda\nu\xi} = 0,$$

which can be written as

$$(\partial_{\xi} + [\Gamma_{\xi}, \cdot])R_{\mu\nu} + (\partial_{\nu} + [\Gamma_{\nu}, \cdot])R_{\xi\mu} + (\partial_{\mu} + [\Gamma_{\mu}, \cdot])R_{\nu\xi} = 0, \qquad (B.5)$$

where we hide the Lie algebra indices of curvature by using matrix notation

$$[\Gamma_{\xi}, R_{\mu\nu}]^{\kappa}_{\ \lambda} = \Gamma^{\kappa}_{\ \xi\eta} R^{\eta}_{\ \lambda\mu\nu} - R^{\kappa}_{\ \eta\mu\nu} \Gamma^{\eta}_{\ \xi\lambda}.$$

Substituting (B.4) into (B.5) we get that

$$\begin{split} 0 &= \partial_{\xi} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{\lambda\nu} - g_{\lambda\mu}Ric^{\kappa}{}_{\nu} + g_{\lambda\nu}Ric^{\kappa}{}_{\mu} - \delta^{\kappa}{}_{\nu}Ric_{\lambda\mu}) + \mathcal{W}^{\kappa}{}_{\lambda\mu\nu} \right] \\ &+ \frac{1}{2} \partial_{\xi} (-\epsilon^{\vartheta}{}_{\lambda\mu\nu}Ric_{*}{}^{\kappa}{}_{\vartheta} + \epsilon^{\vartheta\kappa}{}_{\mu\nu}Ric_{*\lambda\vartheta}) \\ &+ \partial_{\nu} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\xi}Ric_{\lambda\mu} - g_{\lambda\xi}Ric^{\kappa}{}_{\mu} + g_{\lambda\mu}Ric^{\kappa}{}_{\xi} - \delta^{\kappa}{}_{\mu}Ric_{\lambda\xi}) + \mathcal{W}^{\kappa}{}_{\lambda\xi\mu} \right] \\ &+ \frac{1}{2} \partial_{\nu} (-\epsilon^{\vartheta}{}_{\lambda\xi\mu}Ric_{*}{}^{\kappa}{}_{\vartheta} + \epsilon^{\vartheta\kappa}{}_{\xi}Ric_{*\lambda\vartheta}) \\ &+ \partial_{\mu} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\nu}Ric_{\lambda\xi} - g_{\lambda\nu}Ric^{\kappa}{}_{\xi} + g_{\lambda\xi}Ric^{\kappa}{}_{\nu} - \delta^{\kappa}{}_{\xi}Ric_{\lambda\nu}) + \mathcal{W}^{\kappa}{}_{\lambda\nu\xi} \right] \\ &+ \frac{1}{2} \partial_{\mu} (-\epsilon^{\vartheta}{}_{\lambda\nu\xi}Ric_{*}{}^{\kappa}{}_{\vartheta} + \epsilon^{\vartheta\kappa}{}_{\nu\xi}Ric_{*\lambda\vartheta}) \\ &+ \Gamma^{\kappa}{}_{\nu\eta} \left[ \frac{1}{2} (\delta^{\eta}{}_{\xi}Ric_{\lambda\mu} - g_{\lambda\xi}Ric^{\eta}{}_{\mu} + g_{\lambda\mu}Ric^{\eta}{}_{\xi} - \delta^{\eta}{}_{\mu}Ric_{\lambda\xi}) + \mathcal{W}^{\eta}{}_{\lambda\xi\mu} \right] \\ &+ \frac{1}{2} \Gamma^{\kappa}{}_{\nu\eta} (-\epsilon^{\vartheta}{}_{\lambda\xi\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\xi\mu}Ric_{*\eta\vartheta}) \\ &- \Gamma^{\eta}{}_{\nu\lambda} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\xi}Ric_{\eta\mu} - g_{\eta\xi}Ric^{\kappa}{}_{\mu} + g_{\eta\mu}Ric^{\kappa}{}_{\xi} - \delta^{\kappa}{}_{\mu}Ric_{\eta\xi}) + \mathcal{W}^{\eta}{}_{\mu\mu\nu} \right] \\ &+ \frac{1}{2} \Gamma^{\kappa}{}_{\nu\eta} (-\epsilon^{\vartheta}{}_{\eta\xi\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\xi\mu}Ric_{*\eta\vartheta}) \\ &+ \Gamma^{\kappa}{}_{\xi\eta} \left[ \frac{1}{2} (\delta^{\eta}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\nu}Ric_{*\lambda\vartheta}) \\ &- \Gamma^{\eta}{}_{\xi\lambda} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\nu}Ric_{*\lambda\vartheta}) \\ &- \Gamma^{\eta}{}_{\xi\lambda} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\nu}Ric_{*\lambda\vartheta}) \\ &+ \Gamma^{\kappa}{}_{\mu\eta} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\nu}Ric_{*\lambda\vartheta}) \\ &- \Gamma^{\eta}{}_{\xi\lambda} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\nu}Ric_{*\eta\vartheta}) \\ &+ \Gamma^{\kappa}{}_{\mu\eta} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &+ \Gamma^{\kappa}{}_{\mu\eta} \left[ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &+ \Gamma^{\eta}{}_{\mu\eta} \left( -\epsilon^{\vartheta}{}_{\mu\nu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &- \Gamma^{\eta}{}_{\mu} \left\{ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &- \Gamma^{\eta}{}_{\mu} \left\{ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &- \Gamma^{\eta}{}_{\mu} \left\{ \frac{1}{2} (\delta^{\kappa}{}_{\mu}Ric_{*}{}^{\eta}{}_{\vartheta} + \epsilon^{\vartheta\eta}{}_{\mu\xi}Ric_{*\eta\vartheta}) \\ &- \Gamma^{$$

A lengthy but straightforward calculation yields

$$\begin{split} 0 &= \frac{1}{2} \left\{ \delta^{\kappa}_{\ \mu} (\nabla_{\xi} Ric_{\lambda\nu} + \Gamma^{\eta}_{\xi\nu} Ric_{\lambda\eta}) + g_{\lambda\mu} (-\nabla_{\xi} Ric^{\kappa}_{\ \nu} - \Gamma^{\eta}_{\xi\nu} Ric^{\kappa}_{\ \eta}) \right. \\ &+ g_{\lambda\nu} (\nabla_{\xi} Ric^{\kappa}_{\ \mu} + \Gamma^{\eta}_{\xi\mu} Ric^{\kappa}_{\ \eta}) + \delta^{\kappa}_{\ \nu} (-\nabla_{\xi} Ric_{\lambda\mu} - \Gamma^{\eta}_{\xi\mu} Ric_{\lambda\eta}) \\ &+ \delta^{\kappa}_{\ \xi} (\nabla_{\nu} Ric_{\lambda\mu} + \Gamma^{\eta}_{\nu\mu} Ric_{\lambda\eta}) + \delta^{\kappa}_{\ \mu} (-\nabla_{\nu} Ric_{\lambda\xi} - \Gamma^{\eta}_{\nu\mu} Ric_{\lambda\eta}) \\ &+ g_{\lambda\mu} (\nabla_{\nu} Ric^{\kappa}_{\ \xi} + \Gamma^{\eta}_{\ \nu\xi} Ric^{\kappa}_{\ \eta}) + g_{\lambda\xi} (-\nabla_{\nu} Ric^{\kappa}_{\ \mu} - \Gamma^{\eta}_{\nu\mu} Ric^{\kappa}_{\ \eta}) \\ &+ \delta^{\kappa}_{\ \nu} (\nabla_{\mu} Ric_{\lambda\xi} + \Gamma^{\eta}_{\ \mu\xi} Ric_{\lambda\eta}) + \delta^{\kappa}_{\ \xi} (-\nabla_{\mu} Ric_{\lambda\nu} - \Gamma^{\eta}_{\ \mu\nu} Ric^{\kappa}_{\ \eta}) \\ &+ g_{\lambda\xi} (\nabla_{\mu} Ric^{\kappa}_{\ \nu} + \Gamma^{\eta}_{\mu\nu} Ric^{\kappa}_{\ \eta}) + g_{\lambda\nu} (-\nabla_{\mu} Ric^{\kappa}_{\ \xi} - \Gamma^{\eta}_{\ \mu\xi} Ric^{\kappa}_{\ \eta}) \\ &+ g_{\lambda\xi} (\nabla_{\mu} Ric^{\kappa}_{\ \nu} + \Gamma^{\eta}_{\ \mu\nu} Ric^{\kappa}_{\ \eta}) + g_{\lambda\nu} (-\nabla_{\mu} Ric^{\kappa}_{\ \xi} - \Gamma^{\eta}_{\ \mu\xi} Ric^{\kappa}_{\ \eta}) \\ &+ g_{\lambda\xi} (\nabla_{\mu} Ric^{\kappa}_{\ \nu} + \Gamma^{\eta}_{\ \mu\nu} Ric^{\kappa}_{\ \eta}) + g_{\lambda\nu} (-\nabla_{\mu} Ric^{\kappa}_{\ \xi} - \Gamma^{\eta}_{\ \mu\xi} Ric^{\kappa}_{\ \eta}) \\ &+ g_{\lambda\xi} (\nabla_{\mu} Ric^{\kappa}_{\ \nu} + \Gamma^{\eta}_{\ \mu\nu} Ric^{\kappa}_{\ \eta}) + g_{\lambda\nu} (-\nabla_{\mu} Ric^{\kappa}_{\ \xi} - \Gamma^{\eta}_{\ \mu\xi} Ric^{\kappa}_{\ \eta}) \\ &+ G^{\kappa}_{\ \nu} (\delta^{\eta}_{\ \mu} Ric_{\lambda\nu} - \delta^{\eta}_{\ \mu} Ric_{\lambda\xi}) + \Gamma^{\eta}_{\ \nu\lambda} (g_{\eta\xi} Ric^{\kappa}_{\ \nu} - g_{\eta\mu} Ric^{\kappa}_{\ \xi}) \\ &+ \Gamma^{\kappa}_{\ \kappa\eta} (\delta^{\eta}_{\ \mu} Ric_{\lambda\xi} - \delta^{\eta}_{\ \xi} Ric_{\lambda\nu}) + \Gamma^{\eta}_{\ \mu\lambda} (g_{\eta\nu} Ric^{\kappa}_{\ \xi} - g_{\eta\xi} Ric^{\kappa}_{\ \nu}) \\ &+ \partial_{\xi} \left[ (-\epsilon^{\vartheta}_{\ \lambda\mu\nu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\vartheta}_{\ \mu\nu}} Ric_{\kappa\lambda\vartheta}) \right] + \partial_{\nu} \left[ (-\epsilon^{\vartheta}_{\ \lambda\xi\mu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\eta}_{\ \xi\mu}} Ric_{\kappa\lambda\vartheta}) \right] \\ &- \Gamma^{\eta}_{\ \nu\lambda} \left[ (-\epsilon^{\vartheta}_{\ \eta\mu\nu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\vartheta}_{\ \mu\mu}} Ric_{\kappa\eta\vartheta}) \right] + \Gamma^{\kappa}_{\ \eta\eta} \left[ (-\epsilon^{\vartheta}_{\ \lambda\mu\nu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\eta}_{\ \mu\nu}} Ric_{\kappa\lambda\vartheta}) \right] \\ &- \Gamma^{\eta}_{\ \lambda\lambda} \left[ (-\epsilon^{\vartheta}_{\ \eta\mu\nu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\vartheta}_{\ \mu\mu}} Ric_{\kappa\eta\vartheta}) \right] \right] \\ &- \Gamma^{\eta}_{\ \lambda\lambda} \left[ (-\epsilon^{\vartheta}_{\ \eta\nu\mu} Ric^{\kappa}_{\ \vartheta} + \epsilon^{\vartheta^{\vartheta}_{\ \mu\mu}} Ric_{\kappa\eta\vartheta}) \right] \right\} \\ &+ \partial_{\xi} \mathcal{W}^{\kappa}_{\ \mu\nu} + \partial_{\nu} \mathcal{W}^{\kappa}_{\ \lambda\xi\mu} + \Gamma^{\kappa}_{\ \eta\eta} \mathcal{W}^{\eta}_{\ \lambda\xi\mu} - \Gamma^{\eta}_{\ \lambda\nu} \mathcal{W}^{\kappa}_{\ \eta\xi\mu} \\ &+ \Gamma^{\kappa}_{\ \eta\eta} \mathcal{W}^{\eta}_{\ \mu\nu} - \Gamma^{\eta}_{\ \lambda\lambda} \mathcal{W}^{\kappa}_{\ \eta\mu\nu} + \Gamma^{\kappa}_{\ \eta\eta} \mathcal{W}^{\eta}_{\ \lambda\xi\mu}$$

We will now contract the indices  $\kappa$  and  $\mu$  in the above equation.

Since the above equation is too long, first we consider only the terms involving the Weyl curvature  $\mathcal{W}$ . Contracting the indices  $\kappa$  and  $\mu$  and using the formula for the covariant derivative of Weyl tensor, we get that

$$\begin{split} \nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi}\Gamma^{\mu}{}_{\nu\eta}\mathcal{W}^{\eta}{}_{\lambda\xi\mu}+\Gamma^{\mu}{}_{\xi\eta}\mathcal{W}^{\eta}{}_{\lambda\mu\nu}+\Gamma^{\eta}{}_{\mu\nu}\mathcal{W}^{\mu}{}_{\lambda\eta\xi}+\Gamma^{\eta}{}_{\mu\xi}\mathcal{W}^{\mu}{}_{\lambda\nu\eta} \\ =\nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi}+\mathcal{W}^{\mu}{}_{\lambda\xi\eta}(\Gamma^{\eta}{}_{\nu\mu}-\Gamma^{\eta}{}_{\mu\nu})+\mathcal{W}^{\mu}{}_{\lambda\nu\eta}(\Gamma^{\eta}{}_{\mu\xi}-\Gamma^{\eta}{}_{\xi\mu}) \\ =\nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi}+\mathcal{W}^{\mu}{}_{\lambda\xi\eta}(K^{\eta}{}_{\nu\mu}-K^{\eta}{}_{\mu\nu})+\mathcal{W}^{\mu}{}_{\lambda\nu\eta}(K^{\eta}{}_{\mu\xi}-K^{\eta}{}_{\xi\mu}), \end{split}$$

by the symmetry property of the Levi-Civita connection and equation (1.14).

Now we deal with the terms involving the covariant derivative of Ricci curvature, which after the contraction of the indices  $\kappa$  and  $\mu$  become

$$4\nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\xi}Ric_{\lambda\nu} + \nabla_{\nu}Ric_{\lambda\xi} - 4\nabla_{\nu}Ric_{\lambda\xi} + \nabla_{\nu}Ric_{\lambda\xi} + \nabla_{\nu}Ric_{\lambda\xi} - \nabla_{\xi}Ric_{\lambda\nu} + g_{\lambda\xi}\nabla_{\mu}Ric^{\mu}{}_{\nu} - g_{\lambda\nu}\nabla_{\mu}Ric^{\mu}{}_{\xi} = \nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\nu}Ric_{\lambda\xi} + g_{\lambda\xi}\nabla_{\mu}Ric^{\mu}{}_{\nu} - g_{\lambda\nu}\nabla_{\mu}Ric^{\mu}{}_{\xi}.$$

The other terms from the Bianchi identity involving Ricci tensor terms, after the contraction of the indices  $\kappa$  and  $\mu$  and using the symmetry property of the Levi-Civita connection and equation (1.14) become

$$\begin{aligned} &\frac{1}{2} [4Ric_{\lambda\eta}(K^{\eta}{}_{\xi\nu} - K^{\eta}{}_{\nu\xi}) + Ric_{\lambda\eta}(K^{\eta}{}_{\nu\xi} - K^{\eta}{}_{\xi\nu}) + g_{\lambda\nu}Ric^{\mu}{}_{\eta}(K^{\eta}{}_{\xi\mu} - K^{\eta}{}_{\mu\xi}) \\ &+ Ric_{\lambda\eta}(K^{\eta}{}_{\nu\xi} - K^{\eta}{}_{\xi\nu}) + Ric_{\lambda\eta}(K^{\eta}{}_{\nu\xi} - K^{\eta}{}_{\xi\nu}) + g_{\lambda\xi}Ric^{\mu}{}_{\eta}(K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\nu\mu}) \\ &+ Ric_{\lambda\mu}(K^{\mu}{}_{\nu\xi} - K^{\mu}{}_{\xi\nu}) + Ric_{\lambda\xi}(K^{\mu}{}_{\mu\nu} - K^{\mu}{}_{\nu\mu}) + Ric^{\mu}{}_{\mu}(K_{\xi\nu\lambda} - K_{\nu\xi\lambda}) \\ &+ Ric^{\mu}{}_{\xi}(K_{\nu\mu\lambda} - K_{\mu\nu\lambda}) + Ric_{\lambda\nu}(K^{\mu}{}_{\xi\mu} - K^{\mu}{}_{\mu\xi}) + Ric^{\mu}{}_{\nu}(K_{\mu\xi\lambda} - K_{\xi\mu\lambda})] \\ &= \frac{1}{2} [g_{\lambda\nu}Ric^{\mu}{}_{\eta}(K^{\eta}{}_{\xi\mu} - K^{\eta}{}_{\mu\xi}) + g_{\lambda\xi}Ric^{\mu}{}_{\eta}(K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\nu\mu}) \\ &+ Ric^{\mu}{}_{\xi}(K_{\nu\mu\lambda} - K_{\mu\nu\lambda}) + Ric^{\mu}{}_{\nu}(K_{\mu\xi\lambda} - K_{\xi\mu\lambda}) \\ &+ Ric^{\mu}{}_{\lambda\nu}(K^{\mu}{}_{\xi\mu} - K^{\mu}{}_{\mu\xi}) + Ric_{\lambda\xi}(K^{\mu}{}_{\mu\nu} - K^{\mu}{}_{\nu\mu})]. \end{aligned}$$

The terms from the Bianchi identity involving  $Ric_*$  tensor terms, after contracting the indices  $\kappa$  and  $\mu$  and using the definition of the covariant derivative become

$$\frac{1}{2} \left\{ (K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu}) (\epsilon^{\vartheta}{}_{\lambda\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*\lambda\vartheta}) \right. \\ \left. + (K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\nu\mu}) (\epsilon^{\vartheta}{}_{\lambda\xi\eta}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\xi\eta}Ric_{*\lambda\vartheta}) \right. \\ \left. - \nabla_{\mu} (\epsilon^{\vartheta}{}_{\lambda\nu\xi}Ric_{*}{}^{\mu}{}_{\vartheta}) + \nabla_{\mu} (\epsilon^{\vartheta\mu}{}_{\nu\xi}Ric_{*\lambda\vartheta}) \right\}.$$

Putting all these calculations together and using the fact that  $Ric^{\mu}{}_{\eta}K^{\eta}{}_{\xi\mu} = 0$ , the Bianchi identity for curvature becomes

$$0 = \nabla_{\xi} Ric_{\lambda\nu} - \nabla_{\nu} Ric_{\lambda\xi} + g_{\lambda\xi} \nabla_{\mu} Ric^{\mu}{}_{\nu} - g_{\lambda\nu} \nabla_{\mu} Ric^{\mu}{}_{\xi} + Ric^{\mu}{}_{\eta} (g_{\lambda\xi} K^{\eta}{}_{\mu\nu} - g_{\lambda\nu} K^{\eta}{}_{\mu\xi}) + Ric^{\mu}{}_{\xi} (K_{\nu\mu\lambda} - K_{\mu\nu\lambda}) + Ric^{\mu}{}_{\nu} (K_{\mu\xi\lambda} - K_{\xi\mu\lambda}) + Ric_{\lambda\nu} (K^{\mu}{}_{\xi\mu} - K^{\mu}{}_{\mu\xi}) + Ric_{\lambda\xi} (K^{\mu}{}_{\mu\nu} - K^{\mu}{}_{\nu\mu}) + (K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu}) (\epsilon^{\vartheta}{}_{\lambda\eta\nu} Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu} Ric_{*\lambda\vartheta}) + (K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\nu\mu}) (\epsilon^{\vartheta}{}_{\lambda\xi\eta} Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\xi\eta} Ric_{*\lambda\vartheta}) - \nabla_{\mu} (\epsilon^{\vartheta}{}_{\lambda\nu\xi} Ric_{*}{}^{\mu}{}_{\vartheta}) + \nabla_{\mu} (\epsilon^{\vartheta\mu}{}_{\nu\xi} Ric_{*\lambda\vartheta}) + 2\nabla_{\mu} \mathcal{W}^{\mu}{}_{\lambda\nu\xi} + 2\mathcal{W}^{\mu}{}_{\lambda\xi\eta} (K^{\eta}{}_{\nu\mu} - K^{\eta}{}_{\mu\nu}) + 2\mathcal{W}^{\mu}{}_{\lambda\nu\eta} (K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu}).$$
(B.6)

Now we want to eliminate the  $\nabla_{\mu} Ric^{\mu}{}_{\nu}$  term. Contracting the indices  $\lambda$  and  $\xi$  in (B.6), we get that

$$0 = \nabla_{\xi} Ric^{\xi}{}_{\nu} - \nabla_{\nu}R + 4\nabla_{\mu}Ric^{\mu}{}_{\nu} - \nabla_{\mu}Ric^{\mu}{}_{\nu} + Ric^{\mu}{}_{\eta}(4K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\mu\nu}) - Ric^{\mu}{}_{\xi}(K^{\xi}{}_{\mu\nu} - K^{\xi}{}_{\nu\mu}) + Ric^{\mu}{}_{\nu}(K^{\xi}{}_{\mu\xi} - K^{\xi}{}_{\xi\mu}) + Ric^{\xi}{}_{\nu}(K^{\mu}{}_{\xi\mu} - K^{\mu}{}_{\mu\xi}) + Ric^{\xi}{}_{\xi}(K^{\mu}{}_{\mu\nu} - K^{\mu}{}_{\nu\mu}) + (K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu})(\epsilon^{\vartheta\xi}{}_{\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*}{}^{\xi}{}_{\vartheta}) + 2\mathcal{W}^{\mu\xi}{}_{\nu\eta}(K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu}).$$

Hence we can express the  $\nabla Ric$  term as

$$\nabla_{\mu}Ric^{\mu}{}_{\nu} = -\frac{1}{2}Ric^{\mu}{}_{\eta}K^{\eta}{}_{\mu\nu} - \frac{1}{2}Ric^{\beta}{}_{\nu}(K^{\mu}{}_{\beta\mu} - K^{\mu}{}_{\mu\beta}) - \frac{1}{2}\mathcal{W}^{\mu\alpha}{}_{\nu\eta}(K^{\eta}{}_{\mu\alpha} - K^{\eta}{}_{\alpha\mu}) - \frac{1}{2}K^{\eta}{}_{\mu\beta}(\epsilon^{\vartheta\beta}{}_{\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*}{}^{\beta}{}_{\vartheta}).$$
(B.7)

Substituting (B.7) into (B.6), we get that

$$\begin{aligned} \nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi} &= -\frac{1}{2} [\nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\nu}Ric_{\lambda\xi} + \nabla_{\mu}(\epsilon^{\vartheta\mu}{}_{\nu\xi}Ric_{*\lambda\vartheta} - \epsilon^{\vartheta}{}_{\lambda\nu\xi}Ric_{*}{}^{\mu}{}_{\vartheta}) \\ &+ Ric^{\mu}{}_{\xi}(K_{\nu\mu\lambda} - K_{\mu\nu\lambda}) + Ric^{\mu}{}_{\nu}(K_{\mu\xi\lambda} - K_{\xi\mu\lambda}) + Ric_{\lambda\nu}(K^{\mu}{}_{\xi\mu} - K^{\mu}{}_{\mu\xi}) \\ &+ Ric_{\lambda\xi}(K^{\mu}{}_{\mu\nu} - K^{\mu}{}_{\nu\mu}) + (K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu})(\epsilon^{\vartheta}{}_{\lambda\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*\lambda\vartheta}) \\ &+ (K^{\eta}{}_{\mu\nu} - K^{\eta}{}_{\nu\mu})(\epsilon^{\vartheta}{}_{\lambda\xi\eta}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\xi\eta}Ric_{*\lambda\vartheta})] \\ &+ \frac{1}{4}Ric^{\mu}{}_{\eta}(g_{\lambda\xi}K^{\eta}{}_{\mu\nu} - g_{\lambda\nu}K^{\eta}{}_{\mu\xi}) + \frac{1}{4}(g_{\lambda\xi}Ric^{\beta}{}_{\nu} - g_{\lambda\nu}Ric^{\beta}{}_{\xi})(K^{\mu}{}_{\beta\mu} - K^{\mu}{}_{\mu\beta}) \\ &+ \frac{1}{4}g_{\lambda\xi}K^{\eta}{}_{\mu\beta}(\epsilon^{\vartheta\beta}{}_{\eta\nu}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\nu}Ric_{*}{}^{\beta}{}_{\vartheta}) \\ &- \frac{1}{4}g_{\lambda\nu}K^{\eta}{}_{\mu\beta}(\epsilon^{\vartheta\beta}{}_{\eta\xi}Ric_{*}{}^{\mu}{}_{\vartheta} - \epsilon^{\vartheta\mu}{}_{\eta\xi}Ric_{*}{}^{\beta}{}_{\vartheta}) \\ &+ \frac{1}{4}(g_{\lambda\xi}\mathcal{W}^{\mu\alpha}{}_{\nu\eta} - g_{\lambda\nu}\mathcal{W}^{\mu\alpha}{}_{\xi\eta})(K^{\eta}{}_{\mu\alpha} - K^{\eta}{}_{\alpha\mu}) \\ &- \mathcal{W}^{\mu}{}_{\lambda\xi\eta}(K^{\eta}{}_{\nu\mu} - K^{\eta}{}_{\mu\nu}) - \mathcal{W}^{\mu}{}_{\lambda\nu\eta}(K^{\eta}{}_{\mu\xi} - K^{\eta}{}_{\xi\mu}). \end{aligned}$$

**Remark B.2.1.** Note that the equation (B.8) represents the Bianchi identity for curvature without using any assumptions on the nature of the torsion.

Now we apply assumption (ii) that the torsion is purely axial and the relationship between the Levi-Civita and Weyl tensors given in Lemma 1.4.11. Equation (B.7) now becomes

$$\nabla_{\mu}Ric^{\mu}{}_{\nu} = \epsilon^{\vartheta\beta}{}_{\nu\eta}K^{\eta}{}_{\mu\beta}Ric_{*}{}^{\mu}{}_{\vartheta}, \qquad (B.9)$$

and equation (B.8) becomes

$$\nabla_{\mu}\mathcal{W}^{\mu}{}_{\lambda\nu\xi} = -\frac{1}{2} \left[ \nabla_{\xi}Ric_{\lambda\nu} - \nabla_{\nu}Ric_{\lambda\xi} + 2Ric^{\mu}{}_{\xi}K_{\nu\mu\lambda} + 2Ric^{\mu}{}_{\nu}K_{\mu\xi\lambda} \right. \\ \left. + \epsilon^{\vartheta\mu}{}_{\nu\xi}\nabla_{\mu}Ric_{*\lambda\vartheta} - \epsilon^{\vartheta}{}_{\lambda\nu\xi}\nabla_{\mu}Ric_{*}{}^{\mu}{}_{\vartheta} \right] \\ \left. - \epsilon^{\vartheta}{}_{\lambda\eta\nu}K^{\eta}{}_{\mu\xi}Ric_{*}{}^{\mu}{}_{\vartheta} + \epsilon^{\vartheta\mu}{}_{\eta\nu}K^{\eta}{}_{\mu\xi}Ric_{*\lambda\vartheta} \right. \\ \left. - \epsilon^{\vartheta}{}_{\lambda\xi\eta}K^{\eta}{}_{\mu\nu}Ric_{*}{}^{\mu}{}_{\vartheta} + \epsilon^{\vartheta\mu}{}_{\xi\eta}K^{\eta}{}_{\mu\nu}Ric_{*\lambda\vartheta} \right. \\ \left. + \frac{1}{2}g_{\lambda\xi}\epsilon^{\vartheta\zeta}{}_{\eta\nu}K^{\eta}{}_{\mu\zeta}Ric_{*}{}^{\mu}{}_{\vartheta} - \frac{1}{2}g_{\lambda\nu}\epsilon^{\vartheta\zeta}{}_{\eta\xi}K^{\eta}{}_{\mu\zeta}Ric_{*}{}^{\mu}{}_{\vartheta} \right. \\ \left. - 2\mathcal{W}^{\mu}{}_{\lambda\xi\eta}K^{\eta}{}_{\nu\mu} - 2\mathcal{W}^{\mu}{}_{\lambda\nu\eta}K^{\eta}{}_{\mu\xi}. \tag{B.10}$$

## Appendix C

## Explicit Variations of Some Quadratic Forms on Curvature

In this appendix we provide the explicit variations of certain quadratic forms on curvature which are used in this thesis. We provide only the variations with respect to the metric following Proposition A.1.1.

### C.1 Variation of $\int Ric_{\mu\nu}Ric^{\mu\nu}$

Since

$$\frac{\delta}{\delta g} \int Ric_{\mu\nu} Ric^{\mu\nu} \sqrt{|\det g|} = \int \delta \left( Ric_{\mu\nu} Ric_{\mu'\nu'} g^{\mu\mu'} g^{\nu\nu'} \sqrt{|\det g|} \right),$$

we get that

$$\frac{\delta}{\delta g} \int Ric_{\mu\nu}Ric^{\mu\nu}\sqrt{|\det g|} \\
= \int (\delta g_{\alpha\beta}) Ric_{\mu\nu}Ric_{\mu'\nu'} \left(-g^{\nu\nu'}g^{\mu\alpha}g^{\beta\mu'} - g^{\mu\mu'}g^{\nu\alpha}g^{\beta\nu'} + \frac{1}{2}g^{\mu\mu'}g^{\nu\nu'}g^{\alpha\beta}\right) \\
= \int (\delta g_{\alpha\beta}) \left(-Ric^{\alpha}{}_{\nu}Ric^{\beta\nu} - Ric_{\mu}{}^{\alpha}Ric^{\mu\beta} + \frac{1}{2}Ric_{\mu\nu}Ric^{\mu\nu}g^{\alpha\beta}\right).$$

**Remark C.1.1.** Under the assumption that the spacetime is metric compatible, Ricci curvature becomes symmetric and the last equation simplifies to

$$\frac{\delta}{\delta g} \int Ric_{\mu\nu}Ric^{\mu\nu}\sqrt{|\det g|} = -2\int \left(\delta g_{\alpha\beta}\right) \left(Ric_{\mu}{}^{\alpha}Ric^{\mu\beta} - \frac{1}{4}Ric_{\mu\nu}Ric^{\mu\nu}g^{\alpha\beta}\right).$$

## C.2 Variation of $\int Ric^{(2)}_{\mu\nu}Ric_{\mu\nu}$

Using Proposition A.1.1, we get that

**Remark C.2.1.** Under the assumption that the spacetime is metric compatible,  $Ric^{(2)} = -Ric$  and the last equation simplifies to

$$\frac{\delta}{\delta g} \int Ric^{(2)\kappa}{}_{\nu}Ric_{\kappa}{}^{\nu}\sqrt{|\det g|} \\ = \int (\delta g_{\alpha\beta}) \left( -R^{\kappa\alpha\beta}{}_{\nu}Ric_{\kappa}{}^{\nu} + Ric^{\kappa\alpha}Ric_{\kappa}{}^{\beta} - \frac{1}{2}g^{\alpha\beta}Ric^{\kappa}{}_{\nu}Ric_{\kappa}{}^{\nu} \right).$$

## C.3 Variation of $\int Ric^{(2)}_{\mu\nu}Ric^{(2)\mu\nu}$

Since 
$$Ric^{(2)}_{\kappa\nu} = R^{\ \mu}_{\kappa\ \mu\nu}$$
, we have that  
 $\frac{\delta}{\delta g} \int Ric^{(2)}_{\kappa\nu} Ric^{(2)\kappa\nu} = \frac{\delta}{\delta g} \int Ric^{(2)}_{\kappa\nu} Ric^{(2)\kappa\nu} \sqrt{|\det g|}$   
 $= \frac{\delta}{\delta g} \int R^{\ \mu}_{\kappa\ \mu\nu} R^{\kappa\lambda}_{\ \lambda}^{\ \nu} \sqrt{|\det g|} = \frac{\delta}{\delta g} \int R^{\kappa'}_{\ \mu'\mu\nu} R^{\kappa}_{\ \lambda'\lambda\nu'} g_{\kappa\kappa'} g^{\mu\mu'} g^{\lambda\lambda'} g^{\nu\nu'} \sqrt{|\det g|}.$ 
Using Proposition A.1.1, we get that

Using Proposition A.1.1, we get that

$$\begin{split} \frac{\delta}{\delta g} \int Ric^{(2)}_{\kappa\nu} Ric^{(2)\kappa\nu} &= \int (\delta g_{\alpha\beta}) \left( R^{\beta\mu}{}_{\mu\nu} R^{\alpha\lambda}{}_{\lambda}{}^{\nu} - R^{\kappa\beta\alpha\nu} R^{\lambda}{}_{\kappa}{}^{\lambda}{}_{\nu\nu} \right. \\ &\left. - R^{\mu}{}_{\kappa}{}_{\mu\nu} R^{\kappa\beta\alpha\nu} - R^{\mu}{}_{\kappa}{}^{\alpha} R^{\kappa\lambda}{}_{\lambda}{}^{\beta} + \frac{1}{2} R^{\mu}{}_{\kappa}{}_{\mu\nu} R^{\kappa\lambda}{}_{\lambda}{}^{\nu} g^{\alpha\beta} \right) \\ &= \int (\delta g_{\alpha\beta}) \left( Ric^{(2)\beta}{}_{\nu} Ric^{(2)\alpha\nu} - Ric^{(2)}{}_{\kappa}{}^{\alpha} Ric^{(2)\kappa\beta} \right. \\ &\left. - 2Ric^{(2)}{}_{\kappa\nu} R^{\kappa\beta\alpha\nu} + \frac{1}{2} g^{\alpha\beta} Ric^{(2)}{}_{\kappa\nu} Ric^{(2)\kappa\nu} \right). \end{split}$$

**Remark C.3.1.** Under the assumption that the spacetime is metric compatible, Ricci curvature is symmetric and  $Ric^{(2)} = -Ric$ , so the last equation simplifies to

$$\frac{\delta}{\delta g} \int Ric^{(2)}_{\kappa\nu} Ric^{(2)\kappa\nu} \sqrt{|\det g|} = \int (\delta g_{\alpha\beta}) \left( 2Ric_{\kappa\nu} R^{\kappa\beta\alpha\nu} + \frac{1}{2} g^{\alpha\beta} Ric_{\kappa\nu} Ric^{\kappa\nu} \right).$$

## Appendix D

# Detailed calculations of the asymptotic coefficients

In this appendix we provide detailed calculations of the formulae for the asymptotic coefficients (3.70), (3.71), (3.72) and (3.73) from Theorem 3.5.7 which correspond to the eigenvalues  $\lambda = \pm 1$ . We will use the perturbation theory which is described in Section 3.4 and the explicitly derived formulae for the asymptotic coefficients given by (3.52), (3.53). We will also use the concept of the pseudoinverse of the massless Dirac operator whose construction is given in Section 3.4.1.

### **D.1** The calculations of the $\lambda_{\pm}^{(1)}$ coefficients

We first want to show formulae (3.70), (3.71) from Theorem 3.5.7. Using the formula for the eigenvectors (3.77), (3.79) corresponding to the eigenvalues  $\lambda = 1$  and  $\lambda = -1$  respectively, as well as formula (3.75) for the differential operator  $W_{1/2}^{(1)}$ , integrating by parts, we get that the equation (3.52) for the eigenvalues  $\lambda = \pm 1$  becomes

$$\begin{split} \lambda_{+}^{(1)} &= \langle W_{1/2}^{(1)} v^{(0)}, v^{(0)} \rangle = \int_{0}^{2\pi} [v^{(0)}]^{*} W_{1/2}^{(1)} v^{(0)} dx^{1} \\ &= \frac{i}{4} \int_{0}^{2\pi} [v^{(0)}]^{*} \begin{pmatrix} h_{3}^{1} & h_{1}^{1} - ih_{2}^{1} \\ h_{1}^{1} + ih_{2}^{1} & -h_{3}^{1} \end{pmatrix} \frac{d}{dx^{1}} v^{(0)} dx^{1} \\ &- \frac{i}{4} \int_{0}^{2\pi} \frac{d}{dx^{1}} [v^{(0)}]^{*} \begin{pmatrix} h_{3}^{1} & h_{1}^{1} - ih_{2}^{1} \\ h_{1}^{1} + ih_{2}^{1} & -h_{3}^{1} \end{pmatrix} v^{(0)} dx^{1} \\ &= -\frac{1}{4\pi} \int_{0}^{2\pi} h_{1}^{1} (x^{1}) dx^{1} = -\frac{1}{2} \frac{1}{2\pi} \int_{0}^{2\pi} h_{1}^{1} (x^{1}) dx^{1} = -\frac{1}{2} \widehat{h}_{11} (0), \quad (D.1) \end{split}$$

$$\begin{split} \lambda_{-}^{(1)} &= \langle W_{1/2}^{(1)} v^{(0)}, v^{(0)} \rangle = \int_{0}^{2\pi} [v^{(0)}]^* W_{1/2}^{(1)} v^{(0)} dx^1 \\ &= \frac{i}{4} \int_{0}^{2\pi} [v^{(0)}]^* \begin{pmatrix} h_3^{1} & h_1^{1} - ih_2^{1} \\ h_1^{1} + ih_2^{1} & -h_3^{1} \end{pmatrix} \frac{d}{dx^1} v^{(0)} dx^1 \\ &- \frac{i}{4} \int_{0}^{2\pi} \frac{d}{dx^1} [v^{(0)}]^* \begin{pmatrix} h_3^{1} & h_1^{1} - ih_2^{1} \\ h_1^{1} + ih_2^{1} & -h_3^{1} \end{pmatrix} v^{(0)} dx^1 \\ &= \frac{1}{4\pi} \int_{0}^{2\pi} h_1^{1} (x^1) dx^1 = \frac{1}{2} \frac{1}{2\pi} \int_{0}^{2\pi} h_1^{1} (x^1) dx^1 = \frac{1}{2} \hat{h}_{11} (0), \end{split}$$
(D.2)

where  $\hat{h}(m)$  denotes the Fourier coefficient of the function  $h(x^1)$ , see Definition 3.5.1.

## **D.2** The calculations of the $\lambda^{(2)}_{\pm}$ coefficients

First, we will prove the formula (3.72) from Theorem 3.5.7 for the coefficient  $\lambda_{+}^{(2)}$  in the asymptotic expansion (3.68) of the eigenvalue  $\lambda = 1$ . We will separate the calculation of this coefficient into several parts, for sake of simplicity and readability, using the explicit formula (3.53). Let us therefore first evaluate the term  $\langle W_{1/2}^{(2)}v^{(0)}, v^{(0)} \rangle$ . Using the formula for the eigenvector (3.77) corresponding to the eigenvalue  $\lambda = 1$ , as well as formula (3.76) for the differential operator  $W_{1/2}^{(2)}$ , integrating by parts we get that

$$\begin{split} \langle W_{1/2}^{(2)}v^{(0)}, v^{(0)} \rangle &= \int_{0}^{2\pi} [v^{(0)}]^{*} W_{1/2}^{(2)}v^{(0)} dx^{1} \\ &= -\frac{3i}{16} \int_{0}^{2\pi} [v^{(0)}]^{*} \left( \begin{array}{c} (h^{2})_{3}^{1} & (h^{2})_{1}^{1} - i(h^{2})_{2}^{1} \\ (h^{2})_{1}^{1} + i(h^{2})_{2}^{1} & -(h^{2})_{3}^{1} \end{array} \right) \frac{d}{dx^{1}}v^{(0)} dx^{1} \\ &- \frac{3i}{16} \int_{0}^{2\pi} [v^{(0)}]^{*} \frac{d}{dx^{1}} \left( \begin{array}{c} (h^{2})_{3}^{1} & (h^{2})_{1}^{1} - i(h^{2})_{2}^{1} \\ (h^{2})_{1}^{1} + i(h^{2})_{2}^{1} & -(h^{2})_{3}^{1} \end{array} \right) v^{(0)} dx^{1} \\ &+ \frac{i}{16} \int_{0}^{2\pi} [v^{(0)}]^{*} \left( \begin{array}{c} k_{3}^{1} & k_{1}^{1} - ik_{2}^{1} \\ k_{1}^{1} + ik_{2}^{1} & -k_{3}^{1} \end{array} \right) \frac{d}{dx^{1}}v^{(0)} dx^{1} \\ &+ \frac{i}{16} \int_{0}^{2\pi} [v^{(0)}]^{*} \frac{d}{dx^{1}} \left( \begin{array}{c} k_{3}^{1} & k_{1}^{1} - ik_{2}^{1} \\ k_{1}^{1} + ik_{2}^{1} & -k_{3}^{1} \end{array} \right) v^{(0)} dx^{1} \\ &+ \frac{i}{16} \int_{0}^{2\pi} [v^{(0)}]^{*} \left( -\frac{1}{16} \right) \varepsilon_{\beta\gamma1} h_{\alpha\beta} \frac{dh_{\alpha\gamma}}{dx^{1}} Iv^{(0)} \\ &= \frac{3}{8} (\widehat{h^{2}})_{11}(0) - \frac{1}{8} \widehat{k}_{11}(0) - \frac{i}{16} \varepsilon_{\beta\gamma1} \sum_{m_{1} \in \mathbb{Z} \setminus \{0\}} m_{1} \overline{h}_{\alpha\beta}(m_{1}) \widehat{h}_{\alpha\gamma}(m_{1}). \end{split}$$
(D.3)

**Remark D.2.1.** We simplified equation (D.3) using the Fourier coefficients, see Definition 3.5.1, and the Parseval's formula

$$\frac{1}{2\pi} \int_0^{2\pi} p(x)\overline{q(x)} dx = \sum_{m \in \mathbb{Z}} \widehat{p}(m)\overline{\widehat{q}(m)}.$$

Secondly, we will evaluate the term  $\langle (W_{1/2}^{(1)} - \lambda^{(1)})Q(W_{1/2}^{(1)} - \lambda^{(1)})v^{(0)}, v^{(0)} \rangle$ . The pseudoinverse operator, see Section 3.4.1, corresponding to the operator  $W_{1/2} - I$  is given by

$$Q_{+} = \frac{1}{4\pi} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \left[ e^{imx^{1}} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix} \int_{0}^{2\pi} e^{-imy^{1}} (\cdot) dy^{1} + e^{-imx^{1}} \begin{pmatrix} 1 & -1 \\ -1 & 1 \end{pmatrix} \int_{0}^{2\pi} e^{imy^{1}} (\cdot) dy^{1} \right].$$
(D.4)

Using (3.75), (3.77) and (D.1), we have that

$$(W_{1/2}^{(1)} - \lambda_{+}^{(1)})v^{(0)} = -\frac{1}{4\sqrt{\pi}} \begin{pmatrix} h_1^{\ 1} - ih_2^{\ 1} + h_3^{\ 1} \\ h_1^{\ 1} + ih_2^{\ 1} - h_3^{\ 1} \end{pmatrix} e^{ix^1} \\ + \frac{i}{8\sqrt{\pi}} e^{ix^1} \frac{d}{dx^1} \begin{pmatrix} h_1^{\ 1} - ih_2^{\ 1} + h_3^{\ 1} \\ h_1^{\ 1} + ih_2^{\ 1} - h_3^{\ 1} \end{pmatrix} + \frac{\hat{h}_{11}(0)}{4\sqrt{\pi}} \begin{pmatrix} 1 \\ 1 \end{pmatrix} e^{ix^1}.$$
(D.5)

Using the explicit formula for the pseudoinverse (D.4), now we will evaluate the term  $Q_+((W_{1/2}^{(1)} - \lambda_+^{(1)})v^{(0)})$  in three parts. First we will act with the pseudoinverse  $Q_+$  on the first term on the RHS of equation (D.5). Using the well known property of the Fourier coefficient that  $\hat{h}(-m) = \overline{\hat{h}(m)}$ , we obtain

$$\begin{split} Q_{+} & \left( -\frac{1}{4\sqrt{\pi}} \begin{pmatrix} h_{1}^{\ 1} - ih_{2}^{\ 1} + h_{3}^{\ 1} \\ h_{1}^{\ 1} + ih_{2}^{\ 1} - h_{3}^{\ 1} \end{pmatrix} e^{ix^{1}} \right) \\ &= -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left[ e^{imx^{1}} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix} \frac{1}{2\pi} \int_{0}^{2\pi} \begin{pmatrix} h_{1}^{\ 1} - ih_{2}^{\ 1} + h_{3}^{\ 1} \\ h_{1}^{\ 1} + ih_{2}^{\ 1} - h_{3}^{\ 1} \end{pmatrix} e^{-i(m-1)y^{1}} dy^{1} \\ & + e^{-imx^{1}} \begin{pmatrix} 1 & -1 \\ -1 & 1 \end{pmatrix} \frac{1}{2\pi} \int_{0}^{2\pi} \begin{pmatrix} h_{1}^{\ 1} - ih_{2}^{\ 1} + h_{3}^{\ 1} \\ h_{1}^{\ 1} + ih_{2}^{\ 1} - h_{3}^{\ 1} \end{pmatrix} e^{-i(-(m+1))y^{1}} dy^{1} \\ \end{split}$$

$$\begin{split} &= -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left[ e^{imx^1} \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix} \begin{pmatrix} \hat{h}_{11}(m-1) - i\hat{h}_{21}(m-1) + \hat{h}_{31}(m-1) \\ \hat{h}_{11}(m-1) + i\hat{h}_{21}(m-1) - \hat{h}_{31}(m-1) \end{pmatrix} \right. \\ & \left. + e^{-imx^1} \begin{pmatrix} 1 & -1 \\ -1 & 1 \end{pmatrix} \begin{pmatrix} \overline{\hat{h}_{11}(m+1)} - i\overline{\hat{h}_{21}(m+1)} + \overline{\hat{h}_{31}(m+1)} \\ \overline{\hat{h}_{11}(m+1)} + i\overline{\hat{h}_{21}(m+1)} - \overline{\hat{h}_{31}(m+1)} \end{pmatrix} \right] \\ & = -\frac{1}{4\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left( \begin{array}{c} \hat{h}_{11}(m-1)e^{imx^1} - i \ \overline{\hat{h}_{21}(m+1)}e^{-imx^1} + \overline{\hat{h}_{31}(m+1)}e^{-imx^1} \\ \overline{\hat{h}_{11}(m-1)e^{imx^1} + i \ \overline{\hat{h}_{21}(m+1)}e^{-imx^1} - \overline{\hat{h}_{31}(m+1)}e^{-imx^1} \end{array} \right). \end{split}$$

Secondly, we act with the pseudoinverse  $Q_+$  on the second term on the RHS of (D.5). Integrating by parts, we obtain

$$\begin{split} Q_{+} \left( \frac{i}{8\sqrt{\pi}} e^{ix^{1}} \frac{d}{dx^{1}} \left( \begin{array}{c} h_{1}^{1} - ih_{2}^{1} + h_{3}^{1} \\ h_{1}^{1} + ih_{2}^{1} - h_{3}^{1} \end{array} \right) \right) &= -\frac{1}{16\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ \left[ e^{imx^{1}} \left( \begin{array}{c} 1 & 1 \\ 1 & 1 \end{array} \right) \frac{m-1}{2\pi} \int_{0}^{2\pi} e^{-i(m-1)y^{1}} \left( \begin{array}{c} h_{1}^{1} - ih_{2}^{1} + h_{3}^{1} \\ h_{1}^{1} + ih_{2}^{1} - h_{3}^{1} \end{array} \right) dy^{1} \\ &- e^{-imx^{1}} \left( \begin{array}{c} 1 & -1 \\ -1 & 1 \end{array} \right) \frac{m+1}{2\pi} \int_{0}^{2\pi} e^{-i(-(m+1))y^{1}} \left( \begin{array}{c} h_{1}^{1} - ih_{2}^{1} + h_{3}^{1} \\ h_{1}^{1} + ih_{2}^{1} - h_{3}^{1} \end{array} \right) dy^{1} \\ &= -\frac{1}{16\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ \left[ \left( m-1 \right) e^{imx^{1}} \left( \begin{array}{c} 1 & 1 \\ 1 & 1 \end{array} \right) \left( \begin{array}{c} \hat{h}_{11}(m-1) - i\hat{h}_{21}(m-1) + \hat{h}_{31}(m-1) \\ \hat{h}_{11}(m-1) + i\hat{h}_{21}(m-1) - \hat{h}_{31}(m-1) \end{array} \right) \\ &- \left( m+1 \right) e^{-imx^{1}} \left( \begin{array}{c} 1 & -1 \\ -1 & 1 \end{array} \right) \left( \begin{array}{c} \overline{\hat{h}_{11}(m+1)} - i\tilde{h}_{21}(m+1) \\ \overline{\hat{h}_{11}(m+1)} - h_{31}(m+1) \end{array} \right) \right) \\ &= -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ &\left( \begin{array}{c} (m-1)\hat{h}_{11}(m-1)e^{imx^{1}} + i(m+1)\overline{\hat{h}_{21}(m+1)}e^{-imx^{1}} - (m+1)\overline{\hat{h}_{31}(m+1)}e^{-imx^{1}} \\ (m-1)\hat{h}_{11}(m-1)e^{imx^{1}} - i(m+1)\overline{\hat{h}_{21}(m+1)}e^{-imx^{1}} + (m+1)\overline{\hat{h}_{31}(m+1)}e^{-imx^{1}} \end{array} \right). \end{split} \right] \end{split}$$

Thirdly, we act with the pseudoinverse  $Q_+$  on the third and final term on
the RHS of (D.5) to obtain

$$Q_{+}\left(\frac{\hat{h}_{11}(0)}{4\sqrt{\pi}}\begin{pmatrix}1\\1\end{pmatrix}e^{ix^{1}}\right)$$

$$=\frac{\hat{h}_{11}(0)}{16\pi\sqrt{\pi}}\sum_{m\in\mathbb{Z}\setminus\{1\}}\frac{1}{m-1}\left[e^{imx^{1}}\begin{pmatrix}1&1\\1&1\end{pmatrix}\int_{0}^{2\pi}\begin{pmatrix}1\\1\end{pmatrix}e^{i(1-m)y^{1}}dy^{1}\right]$$

$$+e^{-imx^{1}}\begin{pmatrix}1&-1\\-1&1\end{pmatrix}\int_{0}^{2\pi}\begin{pmatrix}1\\1\end{pmatrix}e^{i(m+1)y^{1}}dy^{1}\right].$$
 (D.6)

For  $m \in \mathbb{Z}$  we have that

$$\int_0^{2\pi} e^{i(1-m)y^1} dy^1 = \begin{cases} 2\pi, & m=1, \\ 0, & m\neq 1, \end{cases}, \quad \int_0^{2\pi} e^{i(m+1)y^1} dy^1 = \begin{cases} 2\pi, & m=-1, \\ 0, & m\neq -1, \end{cases}$$

we get that the sum (D.6) only makes sense if m = -1. Hence

$$Q_{+}\left(\frac{\widehat{h}_{11}(0)}{4\sqrt{\pi}}\begin{pmatrix}1\\1\end{pmatrix}e^{ix^{1}}\right) = -\frac{\widehat{h}_{11}(0)}{32\pi\sqrt{\pi}}e^{ix^{1}}\begin{pmatrix}1&-1\\-1&1\end{pmatrix}\begin{pmatrix}2\pi\\2\pi\end{pmatrix} = 0.$$

Putting the three above calculations together, we get that

$$Q_{+}((A^{(1)} - \lambda_{+}^{(1)})v^{(0)}) = -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \left( \frac{(m+1)\widehat{h}_{11}(m-1)e^{imx^{1}} + i(m-1)\overline{\widehat{h}_{21}(m+1)}e^{-imx^{1}} - (m-1)\overline{\widehat{h}_{31}(m+1)}e^{-imx^{1}}}{(m+1)\widehat{h}_{11}(m-1)e^{imx^{1}} - i(m-1)\overline{\widehat{h}_{21}(m+1)}e^{-imx^{1}} + (m-1)\overline{\widehat{h}_{31}(m+1)}e^{-imx^{1}}} \right).$$
(D.7)

Now we will calculate the part  $((W_{1/2}^{(1)} - \lambda_+^{(1)})Q_+((W_{1/2}^{(1)} - \lambda_+^{(1)})v^{(0)})$ . Since

$$\frac{d(Q_{+}((A^{(1)} - \lambda_{+}^{(1)})v^{(0)}))}{dx^{1}} = -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \left( \frac{im(m+1)\hat{h}_{11}(m-1)e^{imx^{1}} + m(m-1)\overline{\hat{h}_{21}(m+1)}e^{-imx^{1}} + im(m-1)\overline{\hat{h}_{31}(m+1)}e^{-imx^{1}}}{im(m+1)\hat{h}_{11}(m-1)e^{imx^{1}} - m(m-1)\overline{\hat{h}_{21}(m+1)}e^{-imx^{1}} - im(m-1)\overline{\hat{h}_{31}(m+1)}e^{-imx^{1}}} \right)$$
  
and

$$\begin{split} \frac{d}{dx^{1}} \left( \left( \begin{array}{cc} h_{31} & h_{11} - ih_{21} \\ h_{11} + ih_{21} & -h_{31} \end{array} \right) Q_{+} ((A^{(1)} - \lambda^{(1)}_{+})v^{(0)}) \right) \\ &= -\frac{1}{8\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \left[ (m+1)\widehat{h}_{11}(m-1) \frac{d}{dx^{1}} \left( \begin{array}{c} h_{11} - ih_{21} + h_{31} \\ h_{11} + ih_{21} - h_{31} \end{array} \right) e^{imx^{1}} \\ &+ i(m-1)\overline{\widehat{h}_{21}(m+1)} \frac{d}{dx^{1}} \left( \begin{array}{c} h_{11} - ih_{21} - h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ &- (m-1)\overline{\widehat{h}_{31}(m+1)} \frac{d}{dx^{1}} \left( \begin{array}{c} h_{11} - ih_{21} - h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ \end{split}$$

using equations (3.75), (D.1) and (D.7), we get that

$$\begin{split} (W_{1/2}^{(1)} - \lambda_{+}^{(1)})Q_{+}((W_{1/2}^{(1)} - \lambda_{+}^{(1)})v^{(0)}) \\ &= -\frac{i}{32\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left[ im(m+1)\widehat{h}_{11}(m-1)e^{imx^{1}} \left( \begin{array}{c} h_{11} - ih_{21} + h_{31} \\ h_{11} + ih_{21} - h_{31} \end{array} \right) \\ & + m(m-1)\overline{h}_{21}(m+1) \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ & + im(m-1)\overline{h}_{31}(m+1) \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ & - \frac{i}{32\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left[ (m+1)\widehat{h}_{11}(m-1) \frac{d}{dx^{1}} \left( \begin{array}{c} h_{11} - ih_{21} + h_{31} \\ h_{11} + ih_{21} - h_{31} \end{array} \right) e^{-imx^{1}} \\ & + i(m-1)\overline{h}_{21}(m+1) \frac{d}{dx^{1}} \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} - h_{31} \end{array} \right) e^{-imx^{1}} \\ & - (m-1)\overline{h}_{31}(m+1) \frac{d}{dx^{1}} \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ & - (m-1)\overline{h}_{31}(m+1) \frac{d}{dx^{1}} \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ & - (m-1)\overline{h}_{31}(m+1) \frac{d}{dx^{1}} \left( \begin{array}{c} -h_{11} + ih_{21} + h_{31} \\ h_{11} + ih_{21} + h_{31} \end{array} \right) e^{-imx^{1}} \\ & - \frac{h_{11}(0)}{16\sqrt{\pi}} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} \\ & \left( \begin{array}{c} (m+1)\widehat{h}_{11}(m-1)e^{imx^{1}} + i(m-1)\overline{h}_{21}(m+1)e^{-imx^{1}} - (m-1)\overline{h}_{31}(m+1)e^{-imx^{1}} \\ (m+1)\widehat{h}_{11}(m-1)e^{imx^{1}} - i(m-1)\overline{h}_{21}(m+1)e^{-imx^{1}} + (m-1)\overline{h}_{31}(m+1)e^{-imx^{1}} \end{array} \right). \end{split} \right] \end{split}$$

Finally, the second term  $\langle (W_{1/2}^{(1)} - \lambda_+^{(1)})Q_+(W_{1/2}^{(1)} - \lambda_+^{(1)})v^{(0)}, v^{(0)} \rangle$ , using (3.13), becomes

$$\langle (W_{1/2}^{(1)} - \lambda_{+}^{(1)}) Q_{+} ((W_{1/2}^{(1)} - \lambda_{+}^{(1)}) v^{(0)}), v^{(0)} \rangle$$

$$= \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} (m+1)^{2} \hat{h}_{11} (m-1) \overline{\hat{h}_{11} (m-1)}$$

$$+ \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \left( \hat{h}_{31} (m+1) + i \hat{h}_{21} (m+1) \right) \left( \overline{\hat{h}_{31} (m+1)} - i \overline{\hat{h}_{21} (m+1)} \right).$$
(D.8)

Combining equations (D.3) and (D.8) , we get the formula (3.53), for the coefficient  $\lambda_+^{(2)}$  is given by

$$\lambda_{+}^{(2)} = \frac{3}{8} \widehat{(h^2)}_{11}(0) - \frac{1}{8} \widehat{k}_{11}(0) - \frac{i}{16} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \widehat{h}_{\alpha\beta}(m) \widehat{h}_{\alpha\gamma}(m) - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} \frac{1}{m-1} (m+1)^2 \widehat{h}_{11}(m-1) \overline{\widehat{h}_{11}(m-1)} - \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{1\}} (m-1) \left( \widehat{h}_{31}(m+1) + i \widehat{h}_{21}(m+1) \right) \left( \overline{\widehat{h}_{31}(m+1)} - i \overline{\widehat{h}_{21}(m+1)} \right).$$

**Remark D.2.2.** Using the eigenvector (3.79) corresponding to the eigenvalue  $\lambda = -1$  of the massless Dirac operator and the pseudoinverse operator (3.80) of the operator  $W_{1/2} + I$ , analogously to the above calculations performed for the eigenvector (3.77) corresponding to the eigenvalue  $\lambda = 1$ , we get that the formula (3.53), for the coefficient  $\lambda_{-}^{(2)}$  is given by

$$\begin{split} \lambda_{-}^{(2)} &= -\frac{3}{8} \widehat{(h^2)}_{11}(0) + \frac{1}{8} \widehat{k}_{11}(0) - \frac{i}{16} \varepsilon_{\beta\gamma 1} \sum_{m \in \mathbb{Z} \setminus \{0\}} m \widehat{h}_{\alpha\beta}(m) \widehat{h}_{\alpha\gamma}(m) \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} \frac{1}{m+1} (m-1)^2 \, \widehat{h}_{11}(m+1) \overline{\widehat{h}_{11}(m+1)} \\ &- \frac{1}{16} \sum_{m \in \mathbb{Z} \setminus \{-1\}} (m+1) \, \left( \widehat{h}_{31}(m-1) + i \widehat{h}_{21}(m-1) \right) \left( \overline{\widehat{h}_{31}(m-1)} - i \overline{\widehat{h}_{21}(m-1)} \right). \end{split}$$

## Bibliography

- W. Adamowicz. Plane waves in gauge theories of gravitation. General Relativity and Gravitation, 12(9):677–691, 1980.
- [2] D. Alekseevsky. Holonomy groups and recurrent tensor fields in Lorentzian spaces. Problems of the Theory of Gravitation and Elementary Particles, 5:5–17, 1974.
- [3] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry. Bull. London Math. Soc., 5:229–234, 1973.
- [4] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry I. Mathematical Proceedings of the Cambridge Philosophical Society, 77(01):43–69, 1975.
- [5] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry II. *Mathematical Proceedings of the Cambridge Philosophical Society*, 78(03):405–432, 1975.
- [6] M. F. Atiyah, V. K. Patodi, and I. M. Singer. Spectral asymmetry and Riemannian geometry III. Mathematical Proceedings of the Cambridge Philosophical Society, 79(01):71–99, 1976.
- [7] J. Audretsch. Asymptotic behaviour of neutrino fields in curved spacetime. *Communications in Mathematical Physics*, 21(4):303–313, 1971.
- [8] O. Babourova, B. Frolov, and E. Klimova. Plane torsion waves in quadratic gravitational theories in Riemann-Cartan space. *Classical* and Quantum Gravity, 16(4):1149, 1999.
- [9] P. Baekler and F. W. Hehl. Beyond Einstein-Cartan gravity: quadratic torsion and curvature invariants with even and odd parity including all boundary terms. *Classical and Quantum Gravity*, 28(21):215017, 2011.
- [10] C. Bär. The Dirac operator on space forms of positive curvature. Journal of the Mathematical Society of Japan, 48(1):69–83, 1996.

- [11] A. Baykal. Gravitational wave solutions of quadratic curvature gravity using a null coframe formulation. *Physical Review D*, 89(6):064054, 2014.
- [12] A. Baykal. Pp-waves in modified gravity. Turk J Phys, 1:36, 2015.
- [13] M. Blagojević, F. W. Hehl, and T. Kibble. Gauge theories of gravitation: a reader with commentaries. Imperial College Press, 2013.
- [14] H. Brinkmann. On Riemann spaces conformal to Euclidean space. Proceedings of the National Academy of Sciences of the United States of America, pages 1–3, 1923.
- [15] R. L. Bryant. Pseudo-Riemannian metrics with parallel spinor fields and vanishing Ricci tensor. Sémin. Congr., 4:53–94, 2000.
- [16] J. Burnett and D. Vassiliev. Modelling the electron with Cosserat elasticity. *Mathematika*, 58(02):349–370, 2012.
- [17] O. Chervova. The massless Dirac equation from the continuum mechanics and microlocal analysis perspectives. PhD thesis, UCL (University College London), 2012.
- [18] O. Chervova, R. J. Downes, and D. Vassiliev. The spectral function of a first order elliptic system. *Journal of Spectral Theory*, 3(3):317–360, 2013.
- [19] O. Chervova, R. J. Downes, and D. Vassiliev. Spectral theoretic characterization of the massless Dirac operator. *Journal of the London Mathematical Society*, 89(1):301–320, 2014.
- [20] O. Chervova and D. Vassiliev. The stationary Weyl equation and Cosserat elasticity. *Journal of Physics A: Mathematical and Theoretical*, 43(33):335203, 2010.
- [21] C. Collinson and P. Morris. Space-times admitting neutrino fields with zero energy and momentum. *Journal of Physics A: Mathematical, Nuclear and General*, 6(7):915, 1973.
- [22] T. M. Davis and J. R. Ray. Ghost neutrinos in general relativity. *Physical Review D*, 9(2):331, 1974.
- [23] V. De Andrade, L. Guillen, and J. Pereira. Gravitational energymomentum density in teleparallel gravity. *Physical Review Letters*, 84(20):4533, 2000.

- [24] R. J. Downes, M. Levitin, and D. Vassiliev. Spectral asymmetry of the massless Dirac operator on a 3-torus. *Journal of Mathematical Physics*, 54(11):111503, 2013.
- [25] R. J. Downes and D. Vassiliev. Spectral theoretic characterization of the massless Dirac action. *Mathematika*, 62:701–718, 2016.
- [26] A. S. Eddington. The mathematical theory of relativity. Cambridge: University Press, 1963, 2nd ed., 1, 1963.
- [27] A. Einstein. Die Grundlage der allgemeinen Relativitätstheorie. Annalen der Physik, 354(7):769–822, 1916.
- [28] A. Einstein. The meaning of relativity: Including the relativistic theory of the non-symmetric field. Princeton University Press, 2014.
- [29] L. Erdős and J. P. Solovej. The kernel of Dirac operators on  $\mathbb{S}^3$  and  $\mathbb{R}^3$ . Reviews in Mathematical Physics, 13(10):1247–1280, 2001.
- [30] W. Esser. Exact solutions of the metric-affine gauge theory of gravity. University of Cologne: Diploma Thesis, 1996.
- [31] E. E. Fairchild Jr. Erratum: Gauge theory of gravitation. *Physical Review D*, 14(10):2833, 1976.
- [32] E. E. Fairchild Jr. Gauge theory of gravitation. *Physical Review D*, 14(2):384, 1976.
- [33] Y.-L. Fang and D. Vassiliev. Analysis as a source of geometry: a nongeometric representation of the Dirac equation. *Journal of Physics A: Mathematical and Theoretical*, 48(16):165203, 2015.
- [34] R. Ferraro and F. Fiorini. Modified teleparallel gravity: inflation without an inflaton. *Physical Review D*, 75(8):084031, 2007.
- [35] T. Friedrich. Zur Abhängigkeit des Dirac-Operators von der Spin-Struktur. Colloquium Mathematicae, 48(1):57–62, 1984.
- [36] T. Friedrich. *Dirac operators in Riemannian geometry*, volume 25. American Mathematical Society Providence, 2000.
- [37] G. Galilei. Dialogo sopra i due massimi sistemi del mondo. 1632.
- [38] G. Galilei. Discorsi e dimostrazioni matematiche intorno a due nuove scienze. 1638.

- [39] A. Garcia, A. Macias, D. Puetzfeld, and J. Socorro. Plane-fronted waves in metric-affine gravity. *Physical Review D*, 62(4):044021, 2000.
- [40] A. A. García, F. W. Hehl, C. Heinicke, and A. Macías. The Cotton tensor in Riemannian spacetimes. *Classical and Quantum Gravity*, 21(4):1099, 2004.
- [41] P. B. Gilkey. Asymptotic formulae in spectral geometry. CRC press, 2003.
- [42] J. Griffiths. Gravitational radiation and neutrinos. Communications in Mathematical Physics, 28(4):295–299, 1972.
- [43] J. Griffiths. Some physical properties of neutrino-gravitational fields. International Journal of Theoretical Physics, 5(3):141–150, 1972.
- [44] J. Griffiths. Neutrino fields in Einstein-Cartan theory. General Relativity and Gravitation, 13(3):227–237, 1981.
- [45] J. Griffiths and R. Newing. Tetrad equations for the two-component neutrino field in general relativity. *Journal of Physics A: General Physics*, 3(3):269, 1970.
- [46] J. Griffiths and R. Newing. The two-component neutrino field in general relativity. Journal of Physics A: General Physics, 3(2):136, 1970.
- [47] J. B. Griffiths. Colliding plane waves in general relativity. Clarendon Press Oxford, 1991.
- [48] J. B. Griffiths and J. Podolsky. Exact space-times in Einstein's general relativity. Cambridge University Press, 2009.
- [49] C. Grossmann, H.-G. Roos, and M. Stynes. Numerical treatment of partial differential equations. Springer, 2007.
- [50] F. W. Hehl and A. Macias. Metric-affine gauge theory of gravity II. Exact solutions. International Journal of Modern Physics D, 8(04):399– 416, 1999.
- [51] F. W. Hehl, J. D. McCrea, E. W. Mielke, and Y. Ne'eman. Metricaffine gauge theory of gravity: field equations, Noether identities, world spinors, and breaking of dilation invariance. *Physics Reports*, 258(1):1– 171, 1995.

- [52] P. W. Higgs. Quadratic Lagrangians and general relativity. Il Nuovo Cimento Series 10, 11(6):816–820, 1959.
- [53] A. D. King and D. Vassiliev. Torsion waves in metric-affine field theory. Classical and Quantum Gravity, 18(12):2317–2329, 2001.
- [54] R. C. Kirby. The topology of 4-manifolds. Springer, 1989.
- [55] D. Kramer and E. Schmutzer. Exact solutions of Einstein's field equations, volume 2003. Cambridge University Press Cambridge, 1980.
- [56] B. Kuchowicz and J. Żebrowski. The presence of torsion enables a metric to allow a gravitational field. *Physics Letters A*, 67(1):16–18, 1978.
- [57] C. Lanczos. A remarkable property of the Riemann-Christoffel tensor in four dimensions. Annals of Mathematics, pages 842–850, 1938.
- [58] C. Lanczos. Lagrangian multiplier and Riemannian spaces. Reviews of Modern Physics, 21(3):497, 1949.
- [59] L. Landau and E. Lifshitz. The classical theory of fields. Course of theoretical physics, 2:5, 1975.
- [60] H. B. Lawson and M.-L. Michelsohn. Spin geometry, volume 38. JS-TOR, 1989.
- [61] B. J. McCartin. Pseudoinverse formulation of Rayleigh-Schrödinger perturbation theory for the symmetric matrix eigenvalue problem. *Journal of Applied Mathematics*, 2003(9):459–485, 2003.
- [62] E. W. Mielke. On pseudoparticle solutions in Yang's theory of gravity. General Relativity and Gravitation, 13(2):175–187, 1981.
- [63] P. J. Mohr, B. N. Taylor, and D. B. Newell. CODATA recommended values of the fundamental physical constants: 2010. *Rev. Mod. Phys.*, 84:1527–1605, Nov 2012.
- [64] U. Muench, F. Gronwald, and F. W. Hehl. A brief guide to variations in teleparallel gauge theories of gravity and the Kaniel-Itin model. *General Relativity and Gravitation*, 30(6):933–961, 1998.
- [65] M. Nakahara. Geometry, topology and physics. IOP Publishing, Bristol, 2nd edition, 1998.

- [66] I. Newton. Philosophiae Naturalis Principia Mathematica. 1687.
- [67] Y. N. Obukhov. Generalized plane-fronted gravitational waves in any dimension. *Physical Review D*, 69(2):024013, 2004.
- [68] Y. N. Obukhov. Plane waves in metric-affine gravity. *Physical Review* D, 73(2):024025, 2006.
- [69] Y. N. Obukhov and J. Pereira. Metric-affine approach to teleparallel gravity. *Physical Review D*, 67(4):044016, 2003.
- [70] Y. N. Obukhov, E. Vlachynsky, W. Esser, R. Tresguerres, and F. Hehl. An exact solution of the metric-affine gauge theory with dilation, shear, and spin charges. *Physics Letters A*, 220(1):1–9, 1996.
- [71] P. Olesen. A relation between the Einstein and the Yang-Mills field equations. *Physics Letters B*, 71(1):189–190, 1977.
- [72] V. Pasic. New vacuum solutions for quadratic metric-affine gravity. PhD thesis, University of Bath, 2009.
- [73] V. Pasic. New vacuum solutions for quadratic metric-affine gravity–a metric affine model for the massless neutrino? *Mathematica Balkanica New Series*, 24(3-4):329–340, 2010.
- [74] V. Pasic and E. Barakovic. Pp-waves with torsion: a metric-affine model for the massless neutrino. *General Relativity and Gravitation*, 46(10):1–27, 2014.
- [75] V. Pasic and E. Barakovic. Torsion wave solutions in Yang-Mielke theory of gravity. Advances in High Energy Physics, Article ID 239076:7, 2015.
- [76] V. Pasic, E. Barakovic, and N. Okicic. A new representation of the field equations of quadratic metric-affine gravity. Adv. Math. Sci. J, 3(1):33–46, 2014.
- [77] V. Pasic and D. Vassiliev. Pp-waves with torsion and metric-affine gravity. Classical and Quantum Gravity, 22(19):3961, 2005.
- [78] W. Pauli. Zur Theorie der Garvitation und der Elektrizität von Hermann Weyl. Physikalische Zeitschrift, 20(20):457–467, 1919.
- [79] R. Pavelle. Unphysical solutions of Yang's gravitational-field equations. *Physical Review Letters*, 34(17):1114, 1975.

- [80] A. Peres. Some gravitational waves. Physical Review Letters, 3(12):571, 1959.
- [81] F. Pfäffle. The Dirac spectrum of Bieberbach manifolds. Journal of Geometry and Physics, 35(4):367–385, 2000.
- [82] D. Puetzfeld. Exact solutions in metric-affine gauge theory of gravity. University of Cologne: Diploma Thesis, 2000.
- [83] D. Puetzfeld. A plane-fronted wave solution in metric-affine gravity. In Exact Solutions and Scalar Fields in Gravity, pages 141–151. Springer, 2002.
- [84] F. Rellich. Perturbation theory of eigenvalue problems. CRC Press, 1969.
- [85] Y. Safarov and D. Vassilev. The asymptotic distribution of eigenvalues of partial differential operators, volume 155. American Mathematical Soc., 1997.
- [86] T. Sauer. Field equations in teleparallel space-time: Einstein's Fernparallelismus approach toward unified field theory. Historia mathematica, 33(4):399–439, 2006.
- [87] P. Singh. On axial vector torsion in vacuum quadratic Poincaré gauge field theory. *Physics Letters A*, 145(1):7–10, 1990.
- [88] P. Singh. On null tratorial torsion in vacuum quadratic Poincaré gauge field theory. Classical and Quantum Gravity, 7(11):2125, 1990.
- [89] P. Singh and J. Griffiths. A new class of exact solutions of the vacuum quadratic Poincaré gauge field theory. *General Relativity and Gravitation*, 22(8):947–956, 1990.
- [90] G. Stephenson. Quadratic Lagrangians and general relativity. Il Nuovo Cimento Series 10, 9(2):263–269, 1958.
- [91] A. Thompson. Yang's gravitational field equations. *Physical Review Letters*, 34(8):507, 1975.
- [92] A. H. Thompson. Geometrically degenerate solutions of the Kilmister-Yang equations. *Physical Review Letters*, 35(5):320, 1975.
- [93] A. Trautman. The Dirac operator on hypersurfaces. Acta Phys. Pol. B, 26(hep-th/9810018):1283–1310, 1995.

- [94] R. Tresguerres. Exact static vacuum solution of four-dimensional metric-affine gravity with nontrivial torsion. *Physics Letters A*, 200(6):405–410, 1995.
- [95] S. Ulhoa and R. Amorim. On teleparallel quantum gravity in Schwarzschild space-time. Advances in High Energy Physics, 2014, 2014.
- [96] A. Unzicker. Teleparallel space-time with defects yields geometrization of electrodynamics with quantized charges. arXiv preprint grqc/9612061, 1996.
- [97] A. Unzicker and T. Case. Translation of Einstein's attempt of a unified field theory with teleparallelism. arXiv preprint physics/0503046, 2005.
- [98] D. Vassiliev. A tensor interpretation of the 2D Dirac equation. arXiv:math-ph/0006019, 2000.
- [99] D. Vassiliev. Pseudoinstantons in metric-affine field theory. *General Relativity and Gravitation*, 34(8):1239–1265, 2002.
- [100] D. Vassiliev. Quadratic non-Riemannian gravity. Journal of Nonlinear Mathematical Physics, 11(sup1):204–216, 2004.
- [101] D. Vassiliev. Quadratic metric-affine gravity. Annalen der Physik, 14(4):231–252, 2005.
- [102] D. Vassiliev. Teleparallel model for the neutrino. *Physical Review D*, 75(2):025006, 2007.
- [103] H. Weyl. Eine neue Erweiterung der Relativitätstheorie. Ann. Phys., 59:101–133, 1919.
- [104] J. A. Wheeler. A journey into gravity and spacetime. Scientific American Library, New York, 1990.
- [105] F. Wilczek. Geometry and interactions of instantons. In Quark confinement and field theory. 1977.
- [106] C.-N. Yang. Integral formalism for gauge fields. Physical Review Letters, 33(7):445, 1974.